### BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

Order Instituting Rulemaking to Promote Policy and Program Coordination and Integration in Electric Utility Resource Planning

Rulemaking 04-04-003 (Filed April 1, 2004)

Order Instituting Rulemaking to Promote Consistency in Methodology and Input Assumptions in Commission Applications of Short-Run and Long-Run Avoided Costs, Including Pricing for Qualifying Facilities Rulemaking 04-04-025 (Filed April 22, 2004)

Application of Southern California Edison Company (U 338-E) for Applying the Market Index Formula and As-Available Capacity Prices Adopted in D.07-09-040 to Calculate Short-Run Avoided Cost for Payments to Qualifying Facilities Beginning July 2003 and Associated Relief Application 08-11-001 (Filed November 4, 2008)

Order Instituting Rulemaking into Implementation of Public Utilities Code Section 390 Rulemaking 99-11-022 (Filed November 18, 1999)

Order Instituting Rulemaking to Integrate Procurement Policies and Consider Long-Term Procurement Plans Rulemaking 06-02-013 (Filed February 16, 2006)

JOINT MOTION
FOR APPROVAL OF QUALIFYING FACILITY AND
COMBINED HEAT AND POWER PROGRAM SETTLEMENT AGREEMENT

Michael D. Montoya Carol A. Schmid-Frazee

Southern California Edison Company

2244 Walnut Grove Avenue Rosemead, CA 91770-3714 Telephone: (626) 302-1337 Facsimile: (626) 302-1935

E-Mail: Carol.SchmidFrazee@sce.com

### Attorneys for Southern California Edison Company

Georgetta J. Baker San Diego Gas & Electric Company 101 Ash Street, HQ12 San Diego, CA 92101-3017 Telephone: (619) 699-5064

Facsimile: (619) 699-5027 E-Mail: gbaker@sempra.com

#### Attorney for San Diego Gas & Electric Company

Jerry R. Bloom Winston & Strawn LLP 333 So. Grand Avenue Los Angeles, CA 90071-1504 Telephone: (213) 615-1700 Facsimile: (213) 615-1750 E-Mail: jbloom@winston.com

#### Attorney for California Cogeneration Council

Michael P. Alcantar Alcantar & Kahl 33 New Montgomery St., Suite 1850 San Francisco, CA 94105-4511 Telephone: (415) 421-4143 Facsimile: (415) 989-1263 E:mail: mpa@a-klaw.com

Attorney for the Cogeneration Association of California and The Energy Producers and Users Coalition

Mary A. Gandesbery Charles R. Middlekauff

Evelyn C. Lee

Pacific Gas and Electric Company

77 Beale Street

San Francisco, CA 94105-1814 Telephone: (415) 973-0675 Facsimile: (415) 973-5520 E-Mail: magq@pge.com

#### Attorney for Pacific Gas and Electric Company

Michel Peter Florio
The Utility Reform Network
115 Sansome Street, Suite 900
San Francisco, CA 94104-3624
Telephone: (415) 929-8776, ext. 302

Facsimile: (415) 929-1132 E-Mail: mflorio@turn.org

#### Attorney for The Utility Reform Network

Douglas K. Kerner Ellison, Schneider & Harris 2600 Capitol Avenue, Suite 400 Sacramento, CA 95816-5931 Telephone: (916) 447-2166 Facsimile: (916) 446-3512 E-Mail: dkk@eslawfirm.com

### Attorney for Independent Energy Producers Association

Lisa-Marie G. Salvacion Division of Ratepayer Advocates 505 Van Ness Avenue San Francisco, CA 94102-3214 Telephone: (415) 703-2069 Facsimile: (415) 703-2057 E-Mail: lms@cpuc.ca.gov

Attorney for California Public Utilities Commission, Division of Ratepayer Advocates

Dated: October 8, 2010

### TABLE OF CONTENTS

		P	age		
I.	FACTUAL AND PROCEDURAL BACKGROUND				
	A.	PURPA and The Commission's QF Program	4		
	B.	State Policy Favoring CHP	6		
	C.	CARB's Climate Change Scoping Plan	6		
	D.	Description Of the Settlement Process	7		
II.	SUMMARY OF THE PROPOSED SETTLEMENT AGREEMENT				
	A.	Section 1 – Goals and Objectives	8		
	B.	Section 2 – Settlement Periods	8		
	C.	Section 3 – Transition PPA	9		
	D.	Section 4 – CHP Procurement Process	9		
	E.	Section 5 – MW Targets	10		
	F.	Section 6 – GHG Emissions Reduction Targets	11		
	G.	Section 7 – GHG Emission Accounting Methodology	13		
	H.	Section 8 - Commission Jurisdictional Entities' Reporting Requirements	13		
	I.	Section 9 – CHP Auditor	14		
	J.	Section 10 – SRAC Energy Pricing Structure	14		
	K.	Section 11 – Legacy PPA Matters for Existing QFs	15		
	L.	Section 12 - CAISO Tariff Compliance			
	M.	Section 13 IOU Cost Recovery For CHP PPAs	16		
	N.	Section 14 Settlement Of Pending And Anticipated Litigation	17		
	O.	Section 15 – FERC 210(m) Application	18		
	P.	Section 16 - Conditions Precedent and Settlement Effective Date	19		
	Q.	Section 17 – Glossary	19		
	R.	Attachments	19		
III.		SETTLEMENT AGREEMENT IS REASONABLE AND IN THE PUBLIC	. 20		
	A.	The Settlement Agreement Is Reasonable And Consistent With Existing  Law	21		
		1. Consistent With State And Commission Policy, The Settlement Agreement Is Intended To Facilitate CHP Goals and Objectives	21		

### TABLE OF CONTENTS

(continued)

Page(s)

		2.	Consistent With State And Commission Policy, The Settlement Agreement Is Intended To Facilitate GHG Emissions Reductions From CHP Facilities	22	
		3.	The QF/CHP Program Procurement Process Is Consistent With The Commission's Preference For Competitive Procurement	23	
		4.	The Energy And Capacity Prices Are Reasonable And Consistent With Recent Commission Decisions	25	
		5.	The QF/CHP Targets Are Appropriate.	26	
		6.	The Semi-Annual Reports And CHP Auditor Process Are Consistent With Commission Policies Supporting Greater Public Information And Transparency	27	
		7.	The Pro Forma PPAs and Legacy QF PPA Amendment	27	
		8.	The Cost Recovery Proposal Is Reasonable And Consistent With California Law	32	
		9.	The Settlement Resolves Numerous Pending And Anticipated Disputes	34	
		10.	The Settlement Agreement Provides For Operationally Flexible Resources	35	
	B.	The Se	ettlement Agreement Is In The Public Interest	36	
IV.	THE JOINT PARTIES HAVE COMPLIED WITH THE REQUIREMENTS OF RULE 12.1(B)				
V.	HEAR	INGS A	ARE NOT REQUIRED	38	
VI.	TIMING FOR REVIEW OF THE SETTLEMENT AGREEMENT AND CONDITIONS PRECEDENT FOR THE SETTLEMENT AGREEMENT TO BECOME EFFECTIVE				
VII.			N.		
	55110		<u> </u>		

#### TABLE OF AUTHORITIES

Page	<b>:(s)</b>					
CASES						
Southern California Edison v. PUC, 101 Cal.App.4 <sup>th</sup> 982 (2002)	. 4					
FERC DECISIONS AND ORDERS						
New York State Electric & Gas Corporation and Rochester Gas and Electric Corporation, 130 FERC ¶ 61,216 (2010)	. 5					
Order No. 688 (Oct. 20, 2006)	. 5					
CALIFORNIA PUBLIC UTILITIES COMMISSION DECISIONS AND RESOLUTION	NS					
Re Pacific Gas & Electric Company, D.88-12-083, 30 CPUC 2d 189 (1988)	20					
D.03-12-062	24					
D.04-01-050	23					
D.04-12-048	33					
D.05-03-022	34					
D.06-07-029	32					
D.07-09-040	im					
D.07-12-052	35					
D.08-09-012	34					
D.08-10-037	22					
D.08-11-008	23					
D.09-10-017						
D.09-12-042						
D.10-04-055	32					
D.10-06-031						
Resolution E-4246	25					
STATUTES						
Public Utility Regulatory Policies Act, 16 U.S.C. § 796, et seqpass:	im					
§ 824a-3(m)						
18 CFR §292.205						
Cal. Health & Saf. Code, § 38501, et seq.						
Cal. Pub. Util. Code § 216.6						

## **TABLE OF AUTHORITIES** (continued)

	Page(s)
Cal. Pub. Util. Code § 365.1(c)(1)	32
Cal. Pub. Util. Code § 372(a)	6, 21
Cal. Pub. Util. Code § 8341	30
Assem. Bill No. 1613 (2007-2008 Reg. Sess.)	24, 32
Assem. Bill No. 32 (2005-2006 Reg. Sess.)	6, 13, 22
Sen. Bill No. 695 (2009-2010 Reg. Sess.)	
CPUC RULES OF PRACTICE AND PROCEI	
Rule 12.1(b)	
Rule 12.1(d)	
MISCELLANEOUS	
Energy Action Plan II (Oct. 2005)	21

### BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

Order Instituting Rulemaking to Promote Policy and Program Coordination and Integration in Electric Utility Resource Planning

Rulemaking 04-04-003 (Filed April 1, 2004)

Order Instituting Rulemaking to Promote Consistency in Methodology and Input Assumptions in Commission Applications of Short-Run and Long-Run Avoided Costs, Including Pricing for Qualifying Facilities Rulemaking 04-04-025 (Filed April 22, 2004)

Application of Southern California Edison Company (U 338-E) for Applying the Market Index Formula and As-Available Capacity Prices Adopted in D.07-09-040 to Calculate Short-Run Avoided Cost for Payments to Qualifying Facilities Beginning July 2003 and Associated Relief Application 08-11-001 (Filed November 4, 2008)

Order Instituting Rulemaking into Implementation of Public Utilities Code Section 390 Rulemaking 99-11-022 (Filed November 18, 1999)

Order Instituting Rulemaking to Integrate Procurement Policies and Consider Long-Term Procurement Plans Rulemaking 06-02-013 (Filed February 16, 2006)

# JOINT MOTION FOR APPROVAL OF QUALIFYING FACILITY AND COMBINED HEAT AND POWER PROGRAM SETTLEMENT AGREEMENT

The relationship among qualifying facilities ("QFs"), the investor-owned utilities ("IOUs") and ratepayer advocate groups has been contentious and litigious for most of the last thirty years. After more than a year and a half of intensive negotiations, QF representatives, the IOUs, and ratepayer advocate groups have developed a proposed combined heat and power ("CHP") settlement agreement ("Settlement Agreement") that resolves numerous outstanding

QF-related disputes and allows for a smooth transition from the California Public Utilities
Commission's ("Commission") existing QF Program to a new QF/CHP Program to preserve
resource diversity, fuel efficiency, greenhouse gas ("GHG") emissions reductions and other
benefits and contributions of CHP. In addition, the Settlement Agreement facilitates additional
CHP benefits and contributions by promoting new, lower GHG emitting CHP facilities and
encouraging the repowering, operational changes through utility pre-scheduling, or retirement of
existing, higher GHG emitting CHP facilities. Finally, the Settlement Agreement appropriately
allocates the costs of the QF/CHP Program to all customers in California who benefit from the
CHP portfolio. In short, the Settlement Agreement provides a reasonable, prudent and wellbalanced approach to the development of QFs and CHP facilities in California to ensure
customer benefits associated with CHP over the near- and long-term.

The parties to the proposed Settlement Agreement represent numerous different groups and interests. These parties include the three investor-owned utilities ("IOUs") -- Pacific Gas and Electric Company ("PG&E"), Southern California Edison Company ("SCE"), and San Diego Gas & Electric Company ("SDG&E"); cogeneration and combined heat and power qualifying facility ("CHP QF") representatives – the California Cogeneration Council ("CCC"), the Independent Energy Producers Association ("IEP"), the Cogeneration Association of California ("CAC"), and the Energy Producers and Users Coalition ("EPUC"); and statewide consumer and ratepayer groups – the Division of Ratepayer Advocates ("DRA") and The Utility Reform Network ("TURN") (the parties are referred to hereinafter individually as a "Party" and collectively as the "Joint Parties").

Pursuant to Rule 12.1 (a) of the California Public Utilities Commission's 
("Commission") Rules of Practice and Procedure, the Joint Parties respectfully file this Joint 
Motion for Approval of Qualifying Facility and Combined Heat and Power Program Settlement

Agreement ("Joint Motion") proposing adoption of the attached Settlement Agreement. While each of these groups have separate interests and concerns, the Joint Parties have worked together to develop a comprehensive framework for a QF/CHP Program in California that will encourage the development of efficient CHP, provide environmental benefits through reduced GHG emissions, resolve outstanding QF disputes and provide clear direction going forward on contentious QF issues including costs. During the settlement process, the Joint Parties were required to compromise and develop solutions. None of the Joint Parties received everything it wanted, and each of the Joint Parties was required to compromise in specific areas so that an overall settlement could be reached. The resulting Settlement Agreement represents a balance of the parties' interests. Consistent with Commission Rule 12.1, the Joint Parties are providing a statement of the factual and legal considerations that are addressed in the Settlement Agreement and demonstrate that the Settlement Agreement is reasonable in light of the whole record, consistent with law, and in the public interest. For these reasons, the Joint Parties respectfully request that the Settlement Agreement be approved by the Commission without modification.<sup>2</sup>

In addition to this Joint Motion, the Joint Parties are filing a Motion for Expedited

Consideration of Joint Motion For Approval of Qualifying Facility and Combined Heat and

Power Settlement Agreement ("Motion for Expedited Consideration"). In Section VI, below, the

Joint Parties provide a proposed schedule for Commission consideration of the Settlement

Agreement. The same schedule is included in the Motion for Expedited Consideration.

///

///

<sup>&</sup>lt;sup>1</sup> The Settlement Agreement is attached as Attachment A to this motion.

<sup>&</sup>lt;sup>2</sup> Each of the Joint Parties expressly reserves its rights to take positions contrary to the positions taken and arguments made in this motion if the Commission does not approve the Settlement Agreement without modification.

#### 1. FACTUAL AND PROCEDURAL BACKGROUND

#### A. PURPA and The Commission's QF Program

In 1978, Congress enacted the Public Utility Regulatory Policies Act ("PURPA"), which was part of a national effort to promote energy independence and efficiency. Under PURPA and the Federal Energy Regulatory Commission's ("FERC") subsequent regulations implementing PURPA, qualifying cogeneration and small power production facilities were provided certain benefits and exemptions. State regulatory agencies were delegated responsibility for developing QF programs and determining avoided-cost pricing. The Commission implemented PURPA in the early 1980s by adopting for the IOUs a number of standard form power purchase agreements ("PPAs") that were available to QFs and established energy and capacity prices to be paid under these PPAs. Many QFs signed these PPAs and built cogeneration and small power production facilities to provide energy and capacity to the IOUs.

Since the Commission implemented the QF program in the 1980s, there have been disputes between the QFs, IOUs and ratepayer advocates including: contract terms, Short-Run Avoided Cost ("SRAC") pricing, capacity payments, contract extensions and terminations, and the availability of new contracts. Many of these disputes are still pending at the Commission. Section 14 of the Settlement identifies disputes pending at the Commission regarding several proceedings, including: retroactive adjustments to SRAC pricing; disputes over pricing and ability to execute PPA extensions; motions for prospective QF PPA options; SRAC disputes dating back to the 2000-2001 energy crisis; disputes concerning administrative heat rates ("AHR") used to calculate SRAC; and applications for rehearing and petitions for modification

<sup>&</sup>lt;sup>2</sup> 16 U.S.C. § 796, et seq.; see also Southern California Edison v. PUC, 101 Cal.App.4<sup>th</sup> 982, 986-87 (2002) (describing PURPA).

of numerous QF decisions.<sup>4</sup> In addition to these disputes pending at the Commission, there are also disputes pending in the California Court of Appeal.<sup>5</sup>

Not only is the Commission faced with disputes regarding existing QF PPAs and the existing QF program, the Commission is also faced with challenges as to how to implement the QF program going forward. For example, in Decision ("D.") 07-09-040, the Commission recognized that it would need to address the impact of the California Independent System Operator's ("CAISO") Market Redesign and Technology Upgrade ("MRTU") on SRAC and the QF program.<sup>6</sup> The Commission also has before it disputes over the terms and conditions of the new QF Standard Offer Contract ("SOC")<sup>7</sup> and disputes over the amount of QF capacity to include in the Long-Term Procurement Process ("LTPP").<sup>8</sup>

On the federal level, recently there have been changes to the PURPA purchase obligation. In October 2006, FERC issued Order No. 688:

... revising its regulations governing utilities' obligations to purchase electric energy produced by QFs. Order No. 688 implements PURPA section 210(m), which provides for termination of the requirement that an electric utility enter into power purchase obligations or contracts to purchase electric energy from QFs, if the Commission finds the QFs have nondiscriminatory access to markets.<sup>2</sup>

<sup>&</sup>lt;sup>4</sup> See Term Sheet, §§ 14.1 – 14.2.

<sup>&</sup>lt;sup>5</sup> Id. at § 14.2.4.

<sup>&</sup>lt;sup>6</sup> D.07-09-040 at p. 68.

<sup>&</sup>lt;sup>2</sup> See e.g., Draft Resolution E-4242 and comments filed by parties concerning the draft resolution.

<sup>&</sup>lt;sup>8</sup> Joint Petition for Modification of D.07-12-052 by Southern California Edison Company (U 338-E), Pacific Gas & Electric Company (U 39-E), and San Diego Gas & Electric Company (U 902-E), filed December 17, 2008 in R.06-02-013.

 $<sup>^{9}</sup>$  New York State Electric & Gas Corporation and Rochester Gas and Electric Corporation, 130 FERC ¶ 61,216 (2010) at P. 3 (footnotes omitted).

Although the California IOUs have not yet sought from FERC a termination of their PURPA purchase obligation for QFs larger than 20 MW, the changes in PURPA further support a reexamination of California's existing QF program.

Given the numerous outstanding disputes, changes in PURPA, and challenges in determining a QF and CHP Program ("QF/CHP Program") going forward, the Joint Parties, California customers and the Commission will benefit from a Settlement that: (1) resolves the outstanding disputes; (2) sets out a clear path for the implementation of a cogeneration QF and CHP Program in California; and, (3) makes available additional PPA options for QFs under the QF/CHP Program ("CHP PPAs").

#### B. State Policy Favoring CHP

Public Utilities Code Section 372(a) and Energy Action Plan II both demonstrate that state policy supports the development of "efficient, environmentally beneficial" CHP. In the 2009 Integrated Energy Policy Report ("IEPR"), the California Energy Commission ("CEC") recommended the continued support and development of CHP as a means to meet state green house gas ("GHG") goals and other policy objectives. <sup>10</sup>

#### C. CARB's Climate Change Scoping Plan

On December 11, 2008, the California Air Resources Board ("CARB") adopted the Climate Change Scoping Plan for California pursuant to Assembly Bill ("AB") 32 (the "CARB Scoping Plan"). In the CARB Scoping Plan, CARB noted that,

[c]ombined heat and power (CHP), also referred to as cogeneration, produces electricity and useful thermal energy in an integrated system. The widespread development of efficient CHP systems would help displace the need to develop new, or expand existing, power plants. This measure sets a target of an additional 4,000 MW of installed CHP capacity by 2020, enough to displace

<sup>&</sup>lt;sup>10</sup> See, 2009 IEPR at pp. 8-9.

<sup>11</sup> See, http://www.arb.ca.gov/cc/scopingplan/document/adopted scoping plan.pdf.

approximately 30,000 GWh of demand from other power generation sources. 12

Although CARB has not yet issued final GHG regulations, the CARB Scoping Plan indicates support for the development of efficient CHP.

#### D. Description Of the Settlement Process

Recognizing the need to resolve outstanding disputes and to establish a new CHP program for California going forward, in May 2009, the Joint Parties and Commission representatives met to lay out a settlement framework. Since that time, the Joint Parties have conducted frequent and lengthy meetings and worked diligently to negotiate the Settlement Agreement now presented to the Commission. The Joint Parties had divergent interests, many of which had been escalated in proceedings at the Commission and before the appellate court, which had to be accommodated. As a result, the Settlement Agreement represents a compromise that should be evaluated as an integrated package. The Settlement Agreement is over 75 pages long and provides a detailed and comprehensive framework for a QF/CHP Program in California. In addition to the Settlement Agreement Term Sheet ("Term Sheet"), the Joint Parties also negotiated four *Pro Forma* PPAs and standard amendments for Legacy QF PPAs for each of the IOUs that will be used as a part of the QF/CHP Program.

Taken as a whole, the Settlement Agreement, including the *Pro Forma* PPAs and amendments described in more detail below, represent a reasonable and appropriate resolution of the many QF issues presently under consideration before the Commission and in other forum. Consequently, the Commission should adopt the Settlement Agreement in its entirety and without change.

Consistent with Rule 12.1(b), the Joint Parties, on September 24, 2010, provided notice to the service lists in these proceedings of a formal settlement conference. <sup>13</sup> The conference was

<sup>&</sup>lt;sup>12</sup> CARB Scoping Plan, at pp. 42-43 (footnotes omitted).

conducted on October 7, 2010. An overview of the proposed Settlement Agreement was presented, participants were able to ask questions and provide comments. Those that were interested in joining to support the Settlement Agreement were invited to do so. After the settlement conference was completed and participants were given an opportunity to review and comment on the Settlement Agreement, this Joint Motion was filed.

#### II. SUMMARY OF THE PROPOSED SETTLEMENT AGREEMENT

This section includes a summary of the key terms of each section of the Term Sheet, as well as the *Pro Forma* PPAs and the *Pro Forma* PPA amendments included with the Settlement Agreement. Given the length of the Settlement Agreement, this section is only intended to be a summary of key terms. Any inconsistencies between this summary and the Term Sheet should be governed by the Term Sheet.

#### A. Section 1 – Goals and Objectives

This section outlines the goals and objectives of the Settlement Agreement.

#### B. Section 2 – Settlement Periods

This section describes the three periods covered by the Settlement Agreement — the Transition Period, the Initial Program Period, and the Second Program Period. The Transition Period is designed to facilitate the transition from the existing QF Program to the new QF/CHP Program. During the Initial Program Period, which overlaps with the Transition Period, the IOUs have specific Megawatt ("MW") Targets ("MW Targets") for entering into new PPAs with

<sup>&</sup>lt;sup>13</sup> Because of widespread interest in matters at issue in these proceedings, notice of potential settlement was also provided to the service lists in R.03-10-003, R.07-05-025, and R.08-06-024.

The fact that a specific provision in the Settlement Agreement is not discussed here does not explicitly or implicitly imply that any provision or term of the Settlement Agreement is more or less important. Moreover, if there is any unintended ambiguity created by the summary below as compared to specific Settlement Agreement terms, the specific provisions in the Settlement Agreement or applicable PPAs and amendments are controlling. The Settlement Agreement is an integrated package and each provision and term was carefully negotiated as a part of that integrated package.

CHP and other facilities. In the Second Program Period, the IOUs procure any portion of the MW Targets that they did not procure during the Initial Program Period and additional CHP capacity to meet GHG Emissions Reduction Targets ("GHG Targets") or other CHP procurement targets established by the Commission. SDG&E has a target to procure an additional 51 MW during the Second Program Period.

#### C. Section 3 – Transition PPA

This section describes the eligibility requirements for QF and CHP facilities for a PPA during the Transition Period and the pricing for Transition Period PPAs. The "Transition Standard Contract for Existing Qualifying Cogeneration Facilities" ("Transition PPA") is included as an exhibit to the Term Sheet and is an attachment to the Settlement Agreement.

#### D. Section 4 – CHP Procurement Process

This section describes the various aspects of the CHP procurement process under the new QF/CHP Program. First, Section 4.2 describes the new CHP Request for Offers ("CHP RFO") process under which the IOUs will procure generation from CHP facilities to meet MW Targets and GHG Targets specified in the Settlement Agreement. Section 4.2 includes eligibility requirements for CHP participating in the RFOs (Section 4.2.2), the delivery terms of PPAs resulting from the RFOs (Section 4.2.3), pricing (Section 4.2.4), and RFO evaluation and selection criteria (Section 4.2.5). In addition, the Joint Parties developed a Pro Forma power purchase agreement for CHP RFOs ("CHP RFO PPA") that will be attached as an exhibit to the Term Sheet.

Section 4 also describes the procurement processes for CHP other than through CHP RFOs that will count towards meeting MW and GHG Targets. Specifically, Sections 4.3 - 4.6

 $<sup>\</sup>frac{15}{1}$  Term Sheet, §§ 3.1 – 3.2.

<sup>16</sup> The MW Targets and GHG Targets are described in Sections 5 and 6 of the Term Sheet, respectively.

describe bilaterally negotiated CHP PPAs, PPAs under the AB 1613 feed-in tariff, PPAs for QFs of 20 MW or less under PURPA, and Optional As-Available PPAs for certain large CHP facilities that have significant on-site load and specific operating characteristics. Section 4.7 addresses utility-owned CHP and limits the contribution of utility-owned facilities to ten percent (10%) of each IOU's GHG Target. IOU-owned facilities will not count toward the MW Targets in the Initial Program Period. Section 4.8 describes "utility prescheduled facilities" which are existing QF facilities that convert to IOU-dispatchable generating facilities. Finally, Section 4.9 addresses new behind-the-meter CHP facilities as one of the procurement options under the QF/CHP Program.

Section 4.10 specifies the Commission approval process required for new PPAs arising from the procurement options in the QF/CHP Program. This includes Tier 2 advice letter filings for existing CHP facilities that execute the CHP RFO PPA without material modification, and a Tier 3 advice letter process for all other CHP PPAs. CHP PPAs that are less than five years in duration do not require Commission pre-approval but will be reported in the IOUs' Quarterly Compliance Reports and CHP Program Semi-Annual Report.

Section 4.11 specifies information that CHP facilities must provide to the IOUs on an annual basis for monitoring purposes and Section 4.12 specifies the timing for commencement of deliveries from a CHP facility that has entered into a new CHP PPA.

#### E. Section 5 – MW Targets

Section 5 establishes a total MW Target for the IOUs of 2,949 MW during the Initial Program Period and a total MW Target of 3,000 MW for the entire QF/CHP Program. Section 5.1.2 includes a chart allocating this MW Target to three target periods for each of the IOUs. For example, the first MW Target for SCE, PG&E, and SDG&E are 630 MW, 630 MW, and 60

<sup>17</sup> This provision in the Settlement Agreement is described in more detail in Section III.A.9, below.

MW, respectively. SDG&E has a specified MW Target during the Second Program Period. If the IOUs have not fulfilled the MW Targets assigned to them for the Initial Program Period they will also need to procure MWs during the latter period to fulfill those targets.

Section 5.1.4 provides that the IOUs are required to conduct three CHP RFOs during the Initial Program Period to seek CHP PPAs to meet the MW Targets. The number of CHP RFOs during the Second Program Period will be established in the LTPP proceedings. <sup>18</sup>

Section 5.2 includes detailed counting rules as to how CHP PPAs executed during the Initial Program Period, whether through a CHP RFO or another procurement process, count toward the MW Targets. Section 5.3 clarifies the appropriate use of the MW counting procedure.

Section 5.4 addresses justifications for an IOU's failure to meet its MW Target. These justifications include lack of sufficient offers in the RFOs, the efficiency of CHP participating in the procurement programs, excessive offer prices 19, and the amount of GHG reductions.

#### F. Section 6 – GHG Emissions Reduction Targets

One of the key benefits of the Settlement Agreement is the implementation of a CHP Program designed to reduce GHG, consistent with the CARB Scoping Plan. Section 6.1 describes the Settlement Agreement strategy for reducing GHG, including maintaining existing, efficient CHP facilities, adding new, efficient CHP resources and achieving the GHG Targets by December 31, 2020. Section 6.2 addresses maintaining the GHG emissions reductions from existing CHP and establishing new targets for GHG reductions from new facilities. In particular, the Settlement Agreement establishes a GHG Emissions Reduction Target or "GHG Target" of 4.3 million-metric tons ("MMT") for the IOUs and 0.5 MMT for Energy Service Providers

<sup>18</sup> Term Sheet, § 5.1.4.7.

<sup>&</sup>lt;sup>19</sup> An IOU claiming that RFO offer prices are excessive must support its claim with information from independent or publicly available sources. *Id.*, § 5.4.1.

("ESPs") and Community Choice Aggregators ("CCA").<sup>20</sup> These targets are based on the 6.7 MMT GHG reduction attributable to CHP in the CARB Scoping Plan.<sup>21</sup> Based on the current percentage of retail sales in California, the 6.7 MMT would be allocated as follows: (1) 4.3 MMT to the IOUs; (2) 0.5 MMT to ESPs and CCAs; and (3) 1.9 MMT to publicly-owned utilities ("POUs").<sup>22</sup> The Commission does not have jurisdiction over POUs, but can set GHG Emissions Reduction Targets for the IOUs, ESPs and CCAs.

Section 6.2.2.3.3 provides for the adjustment of the allocation of the GHG Targets based on changes in retail sales during the term of the Settlement Agreement.<sup>23</sup> Thus, for example, if customers depart utility service for ESPs or CCAs, the GHG Targets for the IOUs will decrease and the targets for the ESPs and CCAs will increase. The GHG Targets can also be adjusted among the IOUs.

Section 6.3 identifies the GHG Target allocated to ESPs and CCAs and indicates that it is the preference of the Joint Parties that these non-IOU load-serving entities ("LSEs") achieve these targets by entering into CHP PPAs. However, if these non-IOU LSEs are not required to enter into CHP PPAs, the IOUs will procure the appropriate amount of CHP for these LSEs to meet their GHG Target and the costs of this procurement by the IOUs will then be allocated to the customers of non-IOU LSEs. The allocation of CHP PPA costs is addressed in Section 13 of the Settlement Agreement. Section 6.4 describes the methodology for establishing the GHG Targets for each of the IOUs. Section 6.5 requires each IOU to report its progress toward meeting its GHG Target in its semi-annual CHP Program Reports that are submitted to the

<sup>&</sup>lt;sup>20</sup> *Id.*, § 6.3.1.

 $<sup>\</sup>frac{21}{2}$  Id., §6.2.2.1.

 $<sup>\</sup>frac{22}{2}$  Id., §6.2.2.3.

 $<sup>\</sup>frac{23}{2}$  See also id., § 6.3.3.

Commission. Section 6.6 states that the GHG Targets for the Second Program Period are subject to review and revision in the LTPP process.

Section 6.7 provides for revisions to the GHG Targets if CARB modifies its CHP reduction goals and provides for GHG Targets to be adjusted in the LTPP if AB 32 compliance is suspended or delayed. In Section 6.8, the Joint Parties agree to advocate at CARB in support of the Settlement Agreement, subject to certain conditions.

Finally, Section 6.9 sets out the justifications for failing to meet the GHG Targets, including the efficiency of CHP facilities participating in the IOUs' procurement programs, excessive offer prices and a lack of need for CHP resources.

#### G. Section 7 - GHG Emission Accounting Methodology

Section 7 establishes the accounting principles for determining the IOUs' progress toward meeting their GHG Targets. This section adopts a Double Benchmark methodology for determining GHG reductions and provides detailed accounting procedures for new, repowered, and existing CHP facilities to determine the amount of GHG emissions reductions that are attributable to these different types of facilities.

#### H. Section 8 - Commission Jurisdictional Entities' Reporting Requirements

Section 8 establishes reporting requirements for Commission-jurisdictional LSEs (*i.e.*, the IOUs, ESPs and CCAs). Each LSE must prepare a semi-annual report detailing progress toward meeting its MW Targets and GHG Targets.<sup>24</sup> Sections 8.2 – 8.5 describe the contents of the semi-annual reports, and specify report content for different categories of CHP generation (*e.g.*, new, legacy, terminated).

13

<sup>&</sup>lt;sup>24</sup> Id., § 8.1.1.

#### I. Section 9 - CHP Auditor

Section 9 provides for a CHP auditor ("CHP Auditor") who is to act as an advocate for CHP interests regarding the implementation of the QF/CHP Program. The CHP Auditor is used in situations where an IOU provides notice that it does not anticipate meeting the MW Targets during a particular RFO or the GHG Targets. The CHP party or parties requesting a CHP Auditor bear the costs<sup>26</sup> and the CHP Auditor is provided with an opportunity to receive and review confidential IOU information regarding the relevant QF/CHP RFO. Section 9 includes provisions for execution of a non-disclosure agreement by the CHP Auditor (Section 9.1.4), when an IOU notice triggers an audit (Section 9.2), the time period for an audit review (Section 9.3), receipt and review of confidential information (Section 9.4), and the number of CHP Auditors, as well as rules regarding any potential conflicts of interest (Section 9.5).

#### J. Section 10 – SRAC Energy Pricing Structure

Section 10 establishes methodologies and formulas for SRAC to be used in Transition PPAs, Legacy PPAs, other existing QF PPAs and Optional As-Available PPAs.<sup>27</sup> Section 10.2 includes a methodology for transitioning, by January 1, 2015, SRAC pricing from a formula that is based in part on administratively-determined heat rates to a formula that uses solely market heat rates. Section 10.4 includes a process for addressing market disruptions that may impact the market heat rate to be used in SRAC. Section 10.2 also includes IOU-specific time-of-use ("TOU") factors to be applied to energy prices to encourage energy deliveries during the times when the energy is most needed by customers. The SRAC formula also includes a locational adjustment based on CAISO nodal prices. Section 10.2 also includes pricing options based on

<sup>25</sup> Id., § 9.1.2

 $<sup>\</sup>frac{26}{10}$  Id., § 9.1.3.

<sup>&</sup>lt;sup>27</sup> Prices for RFO PPAs are based on competitive bids in the RFO process and bilateral PPA prices are based on negotiated prices between the IOU and the CHP party.

whether a cap-and-trade program or other form of GHG regulation is developed in California or nationally.

When such a cap-and-trade program is initially developed that applies to California,

Section 10.2 establishes a floor test which compares an energy price developed with a marketbased heat rate to an energy price developed with either a negotiated heat rate, or a heat rate from
a period prior to the start of a cap and trade program, plus the market price of GHG allowances.

The higher of the two energy prices is the one chosen as SRAC.

Section 10.3 requires the Seller under a CHP PPA to provide certain information to the 10U regarding GHG information that it has reported to CARB or another governmental authority, and information concerning the operation of its facility. Finally, Section 10.5 addresses the responsibility for GHG-related costs.

#### K. Section 11 – Legacy PPA Matters for Existing QFs

Under Section 11.1, QFs with existing standard offers or other PPAs ("QF PPAs") at the time of the Settlement Effective Date<sup>28</sup> will be paid for energy based on the SRAC formula specified in Section 10 (unless the QF PPA specifies a different price) or may elect to amend their standard offer QF PPA to choose one of the energy price options described in the Legacy QF Amendments, are attached as an exhibit to the Settlement Agreement. Unless otherwise specified in the QF PPA, capacity payments for QF PPAs will be based on the capacity price established by the Commission in D. 07-09-040. Section 11.2 provides for the transition from a QF PPA to a new CHP PPA and ensures that delivery from an existing CHP facility continues uninterrupted during that period. The amendments are not available to QFs participating in the Renewable Portfolio Standard program.

-

<sup>28</sup> The Settlement Effective Date is described in Section 16 of the Term Sheet.

Section 11.3 provides that the Seller under an existing QF PPA shall make a good faith effort to provide forecasting information to the IOU so that the IOU can more accurately schedule QF generation in the CAISO markets. This section provides specific forecasting submittal procedures.

#### L. Section 12 - CAISO Tariff Compliance

Section 12 provides that all CHP facilities subject to the CAISO Tariff shall comply with CAISO requirements when the facility begins deliveries under a CHP PPA. Section 12 also includes requirements for the installation of metering and telemetry equipment at existing CHP facilities within six (6) months of the execution of a CHP PPA. The Joint Parties also acknowledge that the CAISO may condition, waive or modify certain requirements for QF and CHP facilities.

#### M. Section 13 -- IOU Cost Recovery For CHP PPAs

Section 13 addresses cost allocation if the Commission determines that IOUs should purchase CHP generation on behalf of ESPs and CCAs.<sup>29</sup> In this circumstance, the IOUs are authorized to recover "net capacity costs" from all bundled, direct access ("DA") and CCA customers on a non-by-passable basis. Net capacity costs are the total costs paid by the IOU under the QF/CHP Program less the value of the energy and ancillary services supplied to the IOU under the program.

Section 13.1.1 recognizes that PPAs under the QF/CHP Program may be greater than ten (10) years and requires that the Commission: (1) affirmatively supersede the ten (10)-year limitation for stranded cost recovery established in D. 04-12-048 and D. 08-09-012 and (2) determine that all above-market or net capacity costs associated with the QF/CHP Program can be recovered for the entire duration of any CHP PPA.

<sup>&</sup>lt;sup>29</sup> Term Sheet, § 13.1.2.2.

Section 13.1.2.1 provides that if the Commission determines that ESPs and CCAs are responsible for procuring CHP generation for their customers, any above-market costs associated with the QF/CHP Program can be allocated to future departing load customers who depart for DA or CCA service.

In Sections 13.1.3 and 13.1.4, the Joint Parties agree that they will not advocate the imposition of QF/CHP Program costs on CHP customer generation departing load, and in Section 13.1.5 the Joint Parties agree to advocate that CHP PPAs entered into as a result of the QF/CHP Program not be included in the existing Competition Transition Charge.

Finally, Section 13.2 provides that all payments made by the IOUs under the QF/CHP Program can be recovered in the IOUs' respective Energy Resources Recovery Account.

#### N. Section 14 -- Settlement Of Pending And Anticipated Litigation

Section 14 addresses the settlement of pending, as well as anticipated, claims and litigation. In Section 14.1, the IOUs agree under certain conditions to withdraw with prejudice all SRAC retroactive price adjustment claims. The Joint Parties mutually agree not to raise any new SRAC retroactive adjustment claims as long as the PURPA purchase obligation remains suspended (as described in more detail in Section 15).

In Section 14.2, the Joint Parties agree to release or withdraw a number of pending claims, rehearing applications, or motions including claims and motions at the Commission (Sections 14.2.1 - 14.2.3, 14.2.5 - 14.2.12) and pending appeals at the Court of Appeal (Section 14.2.4). Section 14 does not affect the Joint Parties' rights to advocate their respective position regarding the confidentiality of IOU procurement information.  $\frac{30}{1000}$ 

<sup>30</sup> Id., § 14.3.2.

#### O. Section 15 – FERC 210(m) Application

Under Section 15, after the Commission approves the Settlement Agreement, the IOUs will submit an application to FERC requesting termination of the IOUs' PURPA purchase requirement from QFs with net capacity in excess of 20 MW, consistent with Section 210(m) of PURPA. Section 15.1 establishes a process for the CHP representatives to review the IOUs' FERC application and provides that these parties can intervene and comment on, but not protest, the IOUs' application. Under Section 15.1.10, the CHP representatives can file at FERC for reinstatement of the PURPA purchase obligation if an IOU "breaches its obligations under the Settlement [Agreement] or the CHP Program adopted in the Settlement [Agreement] is not successfully implemented, based upon the IOU's failure to meet the targets established by the CPUC pursuant to the Settlement [Agreement], without justification as provided for in the Settlement [Agreement]."

Section 15.2 addresses a circumstance where FERC reinstates the PURPA purchase obligation. In this case, SRAC pricing established under the Settlement Agreement stays in place until changed by the Commission (Section 15.2.1.1), although Joint Parties may advocate for a change to SRAC (Section 15.2.1.3). Joint Parties may also advocate for retroactive adjustments to SRAC pricing (Section 15.2.1.4). If the PURPA purchase obligation is reinstated, the IOUs' obligations to conduct CHP RFOs or to engage in alternative procurement processes and the MW Targets and GHG Targets are suspended "provided that the CPUC may on grounds other than the Settlement [Agreement] direct the procurement of CHP resources." (Section 15.2.1.7) Any procurement target to be established by the Commission in the LTPP remains in place unless and until modified by the Commission in a subsequent proceeding. The Joint Parties also agree in Section 15.2.1.8 that for purposes of Section 210(m), designated CHP PPAs constitute "legally enforceable obligations."

#### P. Section 16 - Conditions Precedent and Settlement Effective Date

Section 16.2 specifies that the Settlement Agreement becomes effective upon satisfaction of the following conditions precedent: (1) a final and non-appealable FERC order approving the IOUs' application to terminate their PURPA purchase obligation (Section 16.2.1); (2) a final and non-appealable Commission decision approving the Settlement, including a determination that the Settlement supersedes certain portions of existing Commission decisions (Sections 16.2.2 and 16.2.4 – 16.2.6); and (3) CARB support, in written form, for the Settlement (Section 16.2.3).

Section 16.3 provides that after the Settlement Agreement becomes effective, if CARB adopts regulations directly imposing a MW Target or GHG Emissions Target that differs from the Settlement Agreement for the Second Program Period, the IOUs' obligations to purchase from CHP to meet these targets will remain in place until such time as the Commission is able to consider such change in an LTPP or other pertinent proceeding.

#### Q. Section 17 – Glossary

The section includes a glossary of the defined terms used in the Settlement.

#### R. Attachments

The Settlement Agreement attaches the Term Sheet and Exhibits 1-11 below:

- 1. Amendment to Legacy QF PPA for PG&E
- 2. Amendment to Legacy QF PPA for SCE
- 3. Amendment to Legacy QF PPA for SDG&E
- 4. Transition PPA for existing Qualifying Cogeneration Facilities
- 5. CHP RFO Pro Forma PPA for CHP Facilities Participating in Solicitations
- 6. QF PPA for QFs 20 MW or Less;
- 7. Optional As-Available PPA for eligible As-Available Facilities;
- 8. Non-Disclosure Agreement ("NDA") for CHP Auditor;

- 9. List of Members of CAC
- 10. List of Members of CCC
- 11. List of Members of EPUC

Exhibits 1-7 containing the PPAs are described in more detail below in Section III.A.6. An additional attachment to this Joint Motion, offered for the Commission's information, is the Letter Agreement between the CAISO and the three utilities describing their understanding concerning the utilities' responsibilities concerning CHP/QF compliance with CAISO Tariffs and Protocols under the PPAs attached in Exhibits 4-7.

In addition, included as Attachment B to this Joint Motion is a letter agreement between the CAISO and the IOUs regarding implementation of the Settlement Agreement.

### III. THE SETTLEMENT AGREEMENT IS REASONABLE AND IN THE PUBLIC INTEREST

The Commission will approve a settlement if it finds the settlement "reasonable in light of the whole record, consistent with law, and in the public interest." The proposed Settlement Agreement readily meets these criteria. The Joint Parties negotiated in good faith, bargained aggressively, compromised, and agreed to the Settlement Agreement as an interrelated package; the resolution of any one issue cannot be assessed discreetly. Due to the divergence of the interests of the Joint Parties that had to be accommodated, the Settlement Agreement with regard to each issue represents compromises by various Parties. The Commission, in evaluating the Settlement Agreement, should evaluate it as a package. Each element of the Settlement Agreement is related to all others, any change to the resolution of any one issue may offset the balance that the entire package strikes and represents.

20

<sup>31</sup> Rule 12.1(d); see also D.09-10-017 (applying Rule 12.1(d) criteria).

Factors that the Commission has considered in reviewing settlements include: (1) the risk, expense, complexity and likely duration of further litigation, (2) whether the settlement negotiations were at arms-length, (3) whether major issues were addressed, and (4) whether the parties were adequately represented. In this case, the Settlement Agreement resolves complex and contentious litigation on QF and SRAC pricing matters presently before the Commission and the Court of Appeal. The lengthy settlement negotiations were at arms-length and addressed the major issues regarding the development and operation of CHP in California historically and going forward.

#### A. The Settlement Agreement Is Reasonable And Consistent With Existing Law

1. Consistent With State And Commission Policy, The Settlement Agreement Is Intended To Facilitate CHP Goals and Objectives.

As set forth in the Settlement Agreement, the policy objectives addressed by the Settlement Agreement include requirements under:

- Section 372(a) of the California Public Utilities Code which states: "it is the policy of the state to encourage and support the development of cogeneration technology as an efficient, environmentally beneficial, competitive energy resource that will enhance the reliability of local generation supply, and promote local business growth."
- The Energy Action Plan II which states: "The loading order identifies energy efficiency and demand response as the State's preferred means of meeting growing energy needs. After cost effective efficiency and demand response, we rely on renewable sources of power and distributed generation, such as combined heat and power applications. To the extent efficiency, demand response, renewable resources, and distributed generation are unable to satisfy increasing energy and capacity needs, we support clean and efficient fossil-fired generation."

According to the Settlement Agreement;

"The purpose of the State CHP Program is to encourage the continued operation of the State's Existing CHP Facilities, and the development, installation, and interconnection of new, clean and efficient CHP Facilities, in order to increase the diversity, reliability, and environmental benefits of the energy resources available to the State's electricity

<sup>&</sup>lt;sup>32</sup> Re Pacific Gas & Electric Company, 30 CPUC 2d 189, 222.

consumers."

"These policies and purposes will be achieved by a State CHP Program that procures CHP as set forth in this Settlement, retains existing efficient CHP, supports the change in operations of inefficient CHP to provide greater benefits to the State, and replaces CHP that will no longer be under contract with the IOUs with new, efficient CHP."

2. Consistent With State And Commission Policy, The Settlement Agreement Is Intended To Facilitate GHG Emissions Reductions From CHP Facilities.

When it enacted AB 32, the California Legislature declared that global warming caused by GHG emissions posed a serious threat to California.<sup>33</sup> AB 32 was designed to reduce California's GHG emissions. Since AB 32 was enacted, the Commission has repeatedly indicated that the reduction in GHG emissions is a key policy objective for the utility industry.<sup>34</sup> The Commission, CARB and the CEC have all recognized that efficient and clean CHP can reduce GHG emissions.<sup>35</sup> Indeed, CARB has made CHP one element in its Scoping Plan to implement AB 32 and reduce GHG emissions in California.

As stated in the Settlement Agreement: "In addition, this State CHP Program will secure additional Greenhouse Gas (GHG) emissions reduction benefits, consistent with the reduction targets of Assembly Bill (AB) 32, by adding new, efficient CHP." Consistent with state law and these policy objectives, the Settlement is intended to facilitate the reduction in GHG emissions in a number of ways.

First, under the Settlement, GHG Targets are set for all Commission-jurisdictional LSEs, including the IOUs, ESPs and CCAs. These targets are intended to facilitate the LSEs meeting

<sup>33</sup> Cal. Health and Safety Code, § 38501, et seq.

<sup>&</sup>lt;sup>34</sup> See e.g., D.07-12-052 at pp. 2-5, 243; D.08-10-037 at pp. 2-3 (providing general overview of utility industry role in GHG reduction).

<sup>35</sup> D.08-10-037 at pp. 237-238 (Commission discussion of CHP); CARB Scoping Plan at pp. 43-44; 2009 IPER at pp. 97-98.

<sup>&</sup>lt;sup>36</sup> Term Sheet, § 6.

CARB's CHP goals by December 31, 2020.<sup>37</sup> To the extent CARB modifies its CHP goals, the Settlement provides flexibility to incorporate any modification in the CARB goals.<sup>38</sup>

Second, the Settlement creates incentives for upgrading existing, inefficient CHP facilities, or, alternatively, for facilities that cannot participate or are unsuccessful in the CHP Program, the Settlement Agreement provides an orderly exit strategy. All CHP facilities will be able to participate in the CHP RFOs, and some will be able to participate in other procurement processes and obtain contracts that facilitate the financing, construction and operation of upgraded and/or new facilities. The CHP RFO PPA will explicitly include efficiency performance obligations. The Settlement Agreement recognizes as one of the QF/CHP Program goals upgrading inefficient existing CHP facilities, or allowing them to retire, and encouraging the development of new, clean and efficient CHP.

Third, the Settlement Agreement includes a requirement for all Commission-jurisdictional LSEs to file semi-annual compliance reports that include GHG emissions information. This will allow the Commission and other interested parties to monitor the GHG emissions resulting from the QF/CHP Program and to determine if LSEs are obtaining the GHG benefits expected, and to address any shortfalls in expected GHG emission reduction benefits in a timely manner.

<sup>&</sup>lt;sup>37</sup> *Id.*, § 6.1.

<sup>38</sup> Id., § 6.7

<sup>&</sup>lt;sup>39</sup> *Id.*, § 4.2.9.

<sup>&</sup>lt;sup>40</sup> Id., §§ 1.2.2.3 – 1.2.2.5; see also § 7.3 (GHG accounting methodology which takes into account GHG benefits from new facilities and retirement of inefficient existing CHP facilities).

 $<sup>\</sup>frac{4!}{}$  Id., § 8.

## 3. The QF/CHP Program Procurement Process Is Consistent With The Commission's Preference For Competitive Procurement.

The Commission has repeatedly stated a policy preference for competitive wholesale energy markets and competitive solicitations to procure new resources in those markets. <sup>42</sup>
Currently, CHP QF contracting is not conducted through a competitive solicitation process. The Commission's early QF Program involved the issuance of standard offer contracts that a QF of any technology could sign. In recent years, the CHP QF Program has primarily been sustained by extensions of existing contracts and the availability of short-term contracting options. In D.07-09-040, however, the Commission ordered the IOUs to offer QFs five (5) year as-available and ten (10) year firm PPAs. Despite considerable efforts, those contracts have never been finalized or made available to QFs.

Under the Settlement Agreement, a new, competitive procurement process will be adopted in lieu of the Commission ordered contracts. In particular, the Settlement Agreement creates a CHP RFO process that allows the IOUs to run competitive, transparent RFOs for CHP resources. This is a significant change in CHP procurement and puts CHP resources into a process similar to the one currently used for conventional and Renewable Portfolio Standard ("RPS") procurement. This process will result in competitive prices that are ultimately subject to Commission approval.

In addition, the Commission has also provided for other methods for utility procurement, such as bilateral contracting.<sup>44</sup> The Settlement Agreement provides similar additional flexibility to the IOUs in the CHP procurement process by including not only RFOs, but also other processes such as bilateral contracting, AB 1613 feed-in tariffs, a PURPA Program for QFs

D.04-01-050 at p. 63 (discussing competitive solicitations); D.07-12-052 at p. 205 (discussing development of functional competitive energy market); D.08-11-008 at p. 20 (same).

<sup>43</sup> Term Sheet, § 4.2.

<sup>&</sup>lt;sup>44</sup> See e.g. D.03-12-062 at pp. 38-40 (approving bilateral contracting under certain conditions).

under 20 MW, utility-ownership, and other procurement options. The Settlement Agreement also includes a regulatory approval process for CHP PPAs that result from these procurement options. In short, the Settlement Agreement adopts a procurement process for QF and CHP resources that is competitive, flexible, and allows for sufficient regulatory oversight to ensure that the IOUs are able to minimize costs and select appropriate resources for California customers.

### 4. The Energy And Capacity Prices Are Reasonable And Consistent With Recent Commission Decisions.

There are several different pricing and contracting options in the Settlement. First, CHP PPA prices will be set on a contract-specific basis through a competitive RFO process subject to Commission approval. Allowing CHP developers to bid into the RFO will allow them to propose prices that are sufficient to finance and develop their facilities, while at the same time allowing the IOUs to pick the best offers based on a number of criteria, including price. An RFO procurement process, similar to the processes currently used for conventional and Renewable Portfolio Standard ("RPS") contracts, will result in competitive prices that are ultimately subject to Commission approval. In addition, to the extent that RFO prices are excessive, the Settlement Agreement expressly provides that an IOU may use excessive prices as a justification for failing to meet the MW Targets and GHG Targets.

Second, the Settlement Agreement establishes SRAC prices for the Transition PPAs, Legacy PPAs, QF contracts that are still available under PURPA for facilities less than 20 MW,

 $<sup>\</sup>frac{45}{1}$  Term Sheet, §§ 4.3 – 4.9.

<sup>46</sup> *Id.*, § 4.10.

 $<sup>\</sup>frac{47}{2}$  Bilaterally negotiated PPAs will set contract-specific prices subject to Commission regulatory approval.

<sup>&</sup>lt;sup>48</sup> Term Sheet, § 5.4 and § 5.4.1 (addressing failure to meet the MW Target); § 6.9 (addressing failure to meet the GHG Targets).

and the Optional As-Available PPAs. <sup>49</sup> The SRAC included in the Settlement Agreement is based on the current Commission-approved SRAC pricing formula <sup>50</sup> and achieves the Commission goal of ultimately transitioning to a market heat rate to determine SRAC by January 1, 2015. <sup>51</sup> There is a long history of setting SRAC prices through settlements. The Settlement Agreement resolves this very contentious issue through an arms-length negotiation among adverse parties. As a result, the established SRAC prices are reasonable and in the public interest.

Finally, the Settlement Agreement includes capacity prices that have already been approved by the Commission in D.07-09-040 or are already incorporated in existing contracts. 52

#### 5. The QF/CHP Targets Are Appropriate.

The Settlement establishes MW Targets for each IOU. 53 These MW Targets are the result of heated and protracted negotiations among parties with divergent interests. The CPUC has recognized that a settlement of contested issues among parties with divergent interests is reasonable and in the public interest. In addition, the Settlement Agreement establishes a GHG Target for all Commission-jurisdictional LSEs. These targets are consistent with the CHP targets included in CARB's Scoping Plan, but can also be adjusted to reflect changes by CARB in CHP targets for GHG emissions reductions and if there is a lack of need is asserted by an IOU and determined by the Commission. 54

<sup>49</sup> *Id.*, § 10.1.

<sup>50</sup> D.07-09-040 at p. 67; Resolution E-4246 (issued July 10, 2009) (adopting Market Index Formula).

<sup>51</sup> D.07-09-040 at p. 68 (indicating intent to transition from administrative heat rates to market heat rates).

<sup>&</sup>lt;sup>52</sup> See e.g., Term Sheet, § 3.2.1 (capacity pricing for Transition PPAs); § 4.6.2.2 (capacity pricing for Optional As-Available PPA); and § 11.1 (capacity prices for existing Legacy PPAs).

<sup>53</sup> Term Sheet, § 5.

<sup>&</sup>lt;sup>54</sup> Id., §§ 6.7 (addressing changes to CARB CHP targets); 6.9.3 (lack of need as a justification for not meeting the GHG Targets).

# 6. The Semi-Annual Reports And CHP Auditor Process Are Consistent With Commission Policies Supporting Greater Public Information And Transparency.

The Commission has encouraged transparency in RFO and procurement processes. The Settlement Agreement includes several provisions that promote transparency. First, Commission-jurisdictional LSEs are required to submit semi-annual reports concerning their progress toward achieving the MW Targets and GHG Targets. The Settlement Agreement contains detailed requirements for the type of information to be included in the semi-annual reports. This will provide the Commission and interested parties with information concerning the progress of the QF/CHP Program, and will provide this information with sufficient frequency that the Commission will have an opportunity to address issues and concerns as they arise, rather than waiting until the end of the program to address these issues.

Second, the Settlement Agreement provides for a CHP Auditor to be used for the CHP RFOs if an IOU does not or anticipates that it will not meet its MW Targets or GHG Targets. The CHP Auditor provisions provide the auditor with access to confidential IOU information, to review the CHP RFO process, while including appropriate safeguards to prevent the disclosure of confidential information. The CHP Auditor can review the results of the IOU CHP RFOs, and raise any concerns about the RFOs to the Commission or the Energy Division. This provides an additional level of transparency in the implementation of the OF/CHP Program.

#### 7. The Pro Forma PPAs and Legacy QF PPA Amendment.

The Commission has previously approved the use of *Pro Forma* PPAs for QFs, as well as for use in RFOs for conventional and RPS resources. The Settlement Agreement includes the

<sup>&</sup>lt;sup>55</sup> See e.g. D.07-12-052 at pp. 148-151 (discussing transparency in RFO process).

<sup>&</sup>lt;sup>56</sup> Term Sheet, § 8.

<sup>57</sup> Id., § 9.

following four *Pro Forma* PPAs that were developed for specific circumstances and a *Pro Forma* Legacy QF PPA Amendment for each IOU:

- Legacy QF PPA Amendment -- These *Pro Forma* Amendments offer QFs under unexpired Legacy QF PPAs as of the Settlement Effective Date ("Legacy QFs") the option of amending the energy payment terms of their QF PPAs by selecting one of several payment options and executing the Legacy Amendment within 180 days of the Settlement Effective Date.
- Transition PPA This *Pro Forma* PPA offers an existing CHP facility whose existing QF PPA or extension thereof is scheduled to expire prior to 2015 the option to continue existing deliveries until July 1, 2015.
- CHP RFO PPA This Pro Forma PPA will be issued in the CHP RFOs to
  procure deliveries from CHP and other eligible generators larger than five (5)
  MW.
- Optional As-Available CHP PPA This *Pro Forma* PPA offers gas-fired CHP facilities with nameplates greater than 20 MW, but annual average deliveries less than 131,400 MWh, the option to make as-available deliveries to meet criteria specified in the Settlement Agreement.
- PPA for QFs of 20 MW or Less This *Pro Forma* PPA offers QFs of 20 MW or less, including small power producers and renewable energy resources, the option to make firm or as-available sales to the lOUs.

#### a. Legacy OF PPA Amendments.

The Legacy PPA Amendments allow a QF under a currently effective PPA, excluding those executed in the Renewable Portfolio Standard ("RPS") program, to amend the energy price formula by selecting one of the defined energy pricing options within 180 days of the effective date of the Settlement Agreement. Each of the energy price options is generally based on the SRAC energy pricing structure established by the Settlement ("Settlement SRAC"), as described in Section II.J, above. The energy pricing options differ in terms of the negotiated heat rates and the risk assumed by Seller for the recovery of GHG costs:

• Option A: Option A is identical to the Settlement SRAC pricing structure described in Section II.J., above.

- Option B: Option B employs the same formula for calculating the energy price as used for Option A. However, the negotiated heat rate is higher than Option A until it becomes market-based in 2015 and GHG compliance costs are the responsibility of the Seller.
- Option C1: The Seller's selection of Option C1 triggers a 90-day negotiating period, following the Amendment Effective Date, where parties may agree to a tolling agreement pursuant to which Seller will cause the generating facility to be dispatchable, and Buyer will purchase dispatchable electricity. If Option C1 is selected, the Seller must check a fallback option which shall apply in the absence of a Tolling Agreement.
- Option C2: In addition to making energy payments to the Seller based on a
  negotiated heat rate that is 265 Btu/kWh lower than in Option B, in the event of a
  cap-and-trade GHG control program is established, the Buyer will make payments
  of \$20 per metric ton ("MT") to Seller based on a fixed emission rate for GHG
  compliance costs. In exchange, the Seller is solely responsible for all GHG
  compliance costs.
- Option C3: The energy pricing terms of C3 are identical to those of C2, except that GHG costs are based on facility-specific emissions, capped at Base Year emissions, and an allowance price capped at \$12.50/MT. Annual heat rates are identical to those in Option C2.

The availability of the Legacy QF PPA Amendments is subject to the Commission Approval of the Settlement Agreement and FERC approval of the California IOUs' request to waive the PURPA must-take procurement obligation. This *Pro Forma* amendment incorporates the Joint Parties' settlement of the SRAC pricing issues and offers QFs flexibility to manage the risk of GHG compliance cost.

#### b. Transition PPA.

The Transition PPA is available to CHP facilities currently selling to an IOU under a Legacy PPA or an extension thereof that is due to expire during the Transition Period. A CHP facility may only enter into a Transition PPA with the same IOU that it currently delivers electricity to under a Legacy PPA or an extension thereof. The term of the Transition PPA begins upon the expiration of the CHP facility's existing PPA and may be terminated upon 180 days' notice when a CHP facility has executed a PPA resulting from either a solicitation or

bilateral negotiation. The Seller may provide firm, as-available, or both forms of capacity. The Transition PPA provides firm capacity payment at the rate of \$91.97/kW-yr and as-available capacity payment at \$41.22/kW-yr escalating annually. Energy is priced at the Settlement SRAC.

The Transition PPA requires a delivery schedule, the installation of a CAISO-approved meter within 180 days of contract execution, and agreements to curtail power production upon notification of CAISO or transmission owner instruction.

Although deliveries are generally limited to historic levels under the Legacy PPA, both capacity and energy levels may be modified, provided that any CHP facility modification does not increase the Buyer's GHG costs. Certain CHP facilities with unique operational constraints may negotiate an amendment to the Transition PPA to deliver a standard additional capacity product that meets Commission and /or CAISO requirements for resource adequacy and CAISO protocols.

#### c. CHP RFO PPA.

The CHP RFO PPA is used to solicit competitive offers from certain CHP generators. Within 90 days of the Settlement Effective Date, each IOU will initiate a CHP RFO and issue this CHP RFO PPA to establish the terms and conditions by which existing, new or expanded CHP facilities located within California may offer to sell firm or as-available capacity to the IOU.<sup>58</sup> To be eligible to participate in the CHP RFO, the CHP facility must, among other things, be larger than five (5) MW, must meet the definition of "cogeneration" under Cal. Pub. Util. Code §216.6, must satisfy the Emissions Performance Standard established by Cal. Pub. Util. Code §8341, and must satisfy the definition of "cogeneration facility" under 18 CFR §292.205. Utility Prescheduled Facilities are also eligible to bid into the CHP RFO

<sup>28</sup> The same CHP RFO PPA will be used in subsequent CHP solicitations as well.

Under the CHP RFO PPA, the delivery term for existing facilities and expanded facilities that elect not to satisfy the credit and collateral requirements of the RFO is up to seven (7) years; for new, repowered or expanded facilities that elect to meet the credit and collateral requirements in the RFO, the term is up to 12 years. Terms in the CHP RFO PPA may be modified on a bilateral basis during negotiations for a particular CHP PPA. If the Seller's offer is accepted, the offer will establish the terms of the PPA.

#### d. Optional As-Available CHP PPA.

The As-Available CHP PPA is one of several commercial alternatives available to new, existing, or repowered gas-fired CHP facilities with nameplates greater than 20 MW that meet certain requirements, including the following: the CHP facility's average annual deliveries may not exceed 131,400 MWh; the project host(s) must consume at least 75% of the total electricity generated by a Topping Cycle CHP Facility or at least 25% of the total electricity generated by a Bottoming Cycle CHP Facility; and for Topping Cycle or Bottoming Cycle with supplemental firing, the facility must meet a 60% efficiency standard. <sup>59</sup>

Seller will be paid an as-available capacity price set forth in Exhibit D, Section 3, and a time of delivery ("TOD") energy price set forth in Exhibit D, Section 2. If the generating facility is a new CHP facility, it must maintain Development Security and Performance Assurance in accordance with scheduled amounts or as negotiated between Seller and Buyer. Seller may terminate the Agreement if Seller's facility is selected in a competitive solicitation.

As-available capacity payments will be paid for deliveries of up to 20 MW in any hour. The Seller is required to schedule all deliveries with the IOU on a day-ahead basis sufficiently in advance to allow the IOU to schedule energy into the CAISO day-ahead market. Energy scheduled on a day-ahead basis and delivered up to 20 MW per hour will be priced at Settlement

<sup>&</sup>lt;sup>59</sup> There is no efficiency requirement for a Bottoming Cycle CHP Facility with no supplemental firing.

SRAC. Energy scheduled on a day-ahead basis and delivered at a rate in excess of 20 MW per hour will be priced at the MRTU Day-Ahead market PNode energy price. Unscheduled energy incremental to scheduled energy will be purchased by the IOU at the MRTU real time PNode price, while the Seller will bear CAISO charges and receive all CAISO revenues for such deliveries. The Seller may designate a delivery term of up to seven (7) years.

#### e. PPA for QFs of 20 MW or Less.

The PPA for QFs of 20 MW or Less will be available to QFs with firm or as-available capacity of 20 MW or less under the Commission's continuing PURPA program, regardless of whether the QF has submitted an offer in the CHP RFO or seeks alternative contracting options. The PPA for QFs of 20 MW or Less contains standard terms and conditions and incorporates the capacity prices established in D. 07-09-040, and employs the Settlement SRAC price for energy. There are few terms subject to negotiation. New or repowered facilities must post project development security and performance assurance.

# 8. The Cost Recovery Proposal Is Reasonable And Consistent With California Law.

The Commission has repeatedly determined that where DA and CCA customers benefit from procurement, these customers should pay their share of the procurement costs. For example, the Commission authorized the allocation of new generation resource costs to DA and CCA customers because these customers benefitted from the system reliability provided by the new generation resources. The Commission also allocated GHG compliance costs and certain locational costs associated with CHP facilities developed under AB 1613 to DA and CCA customers because these customers benefitted from the AB 1613 program.

<sup>&</sup>lt;sup>60</sup> D.06-07-029 at p. 7.

<sup>61</sup> D.09-12-042 at pp. 21-25, aff'd, D.10-04-055 at pp. 11-18.

Here, one of the purposes of the Settlement Agreement is to develop a QF/CHP Program that can facilitate meeting CARB's CHP goal as specified in its Scoping Plan. The CARB CHP goal is not limited to the IOUs, but applies to all LSEs in California. Section 365.1(c)(1) of the Public Utilities Code, enacted as part of Senate Bill 695 (2009), requires this Commission to "ensure" that ESPs and CCAs "are subject to the same requirements that are applicable to the state's three largest electrical corporations under any programs or rules adopted by the commission to implement . . . the requirements for the electricity sector adopted by the State Air Resources Board." Under the Settlement Agreement, the CARB CHP goal is equitably allocated among Commission-jurisdictional LSEs based on their respective percentage of total retail sales. This allocation is used to establish GHG Targets for all LSEs, including the IOUs, ESPs and CCA.

As part of its decision on this Settlement Agreement, and based upon input from the parties, including ESPs and CCAs, the Commission will decide whether these entities will be required to meet their portion of the GHG Target by procuring CHP resources, which is the approach the Joint Parties prefer. However, if the Commission determines that ESPs or CCAs are unable or unwilling to meet their portion of the GHG Targets by contracting with CHP facilities, the IOUs have agreed under the terms of the Settlement Agreement to procure CHP resources on behalf of these entities. In this case, however, ESP and CCA customers will be responsible for the costs of CHP resources procured on their behalf by the IOUs. This is consistent with the Commission's recent decisions on cost allocation when ESP and CCA customers benefit from IOU procurement on their behalf.

 $<sup>\</sup>frac{62}{1}$  Term Sheet, §§ 6.2 – 6.3.

<sup>63</sup> *Id.*, § 6.3.2.

<sup>&</sup>lt;sup>64</sup> *Id.*, § 13.1.2.2.

As an alternative to the allocation of costs for CHP resources procured on behalf of ESP and CCA customers, if these entities are required to procure their own CHP resources, then the Settlement Agreement provides for the allocation of any stranded CHP costs to future DA and CCA departing load customers. This allocation of costs is consistent with the Commission's recent departing load cost allocation decisions. However, because PPAs under the Settlement Agreement can have up to a 12-year duration, a condition precedent of the Settlement Agreement becoming effective is that the Commission affirmatively supersede the 10-year limitation in D.08-09-012<sup>67</sup> and determine that PPA above-market costs can be recovered from departing load customers for the entire 12-year term. 88

# 9. The Settlement Resolves Numerous Pending And Anticipated Disputes.

The Commission has a long-standing policy of supporting settlements. <sup>69</sup> "The Commission favors settlements because they generally support worthwhile goals, including reducing the expense of litigation, conserving scarce Commission resources, and allowing parties to reduce the risk that litigation will produce unacceptable results." <sup>70</sup> In this case, rather than resolving a single dispute, the Settlement Agreement resolves numerous disputes pending at both the Commission and in the California Court of Appeal. <sup>71</sup> These disputes involve QF pricing, QF SOC terms and conditions, the amount of QF/CHP capacity included in long-term planning, retroactive SRAC price adjustments dating back to 2000, and numerous other disputes

<sup>65</sup> *Id.*, § 13.1.1.

 $<sup>^{66}</sup>$  See e.g. D.04-12-048 at pp. 56-58; D.08-09-012 at p. 37 (allocating new QF contract costs to DA and CCA departing load customers).

 $<sup>^{62}</sup>$  D.08-09-012 at pp. 52-55 (discussing 10-year limitation).

<sup>68</sup> Term Sheet, § 16.2.5.

 $<sup>\</sup>frac{69}{}$  D.05-03-022 at pp. 7-8; D.10-06-031 at p. 12.

 $<sup>\</sup>frac{70}{2}$  D.10-06-031 at p. 12.

<sup>&</sup>lt;sup>71</sup> Term Sheet, § 14.

Agreement effectively resolves pending disputes by requiring the Joint Parties to either withdraw pending motions and applications, or release certain claims. In addition, the Settlement Agreement precludes the Joint Parties from raising new retroactive SRAC adjustment claims as long as certain conditions are met.<sup>72</sup> Thus, the Settlement not only resolves past disputes, but it also limits potential future disputes regarding SRAC energy prices.

The Settlement Agreement also resolves future potential disputes at FERC concerning an application by the IOUs for waiver of the PURPA purchase obligation by clearly defining what type of application the IOUs will file, and the type of disputes or filings that can be made by the CHP Representatives. But for the Settlement Agreement, the Joint Parties would likely have expended considerable time and resources litigating at FERC the waiver of the PURPA purchase obligation.

# 10. The Settlement Agreement Provides For Operationally Flexible Resources.

Recognizing the amount of intermittent, renewable resources that will be added in California as a result of the RPS requirements, the Commission has recently encouraged the development of operationally flexible conventional resources to assist with renewables integration. One of the challenges for CHP facilities is that these facilities are often operated as baseload facilities and/or need to operate consistent with the needs of a thermal host such that these facilities lack significant operational flexibility. Under the Settlement Agreement, the lOUs can contract with a limited group of existing CHP facilities that convert from a QF facility

<sup>&</sup>lt;sup>72</sup> *Id.*, § 14.1.1.

<sup>&</sup>lt;sup>73</sup> *Id.*, § 15.1.

<sup>&</sup>lt;sup>74</sup> See e.g. D.07-12-052 at pp. 106, 111-112, 115.

to a dispatchable generation facility. The dispatchable generating facility is referred to in the Settlement Agreement as a "Utility Prescheduled Facility." This aspect of the Settlement Agreement has several benefits.

First, if an existing CHP facility converts to a dispatchable facility, it gives the IOU the ability to dispatch the resource when it is needed, rather than the facility providing baseload generation or operating based on a thermal host's needs. This is similar to the contracts the IOUs have with peaking and other existing conventional generation facilities.

Second, conversion to a dispatchable facility may ultimately result in GHG emission reductions. If an existing CHP facility operates as a baseload facility, and is not efficient, its GHG emissions may be higher than a new conventional facility or other resource options. By giving the IOU the flexibility to dispatch a facility, the utility can optimize its GHG emissions reductions by choosing to operate facilities with the lowest total GHG emissions.

#### B. The Settlement Agreement Is In The Public Interest

The Settlement Agreement is clearly in the public interest for a number of reasons. First, the Settlement Agreement resolves numerous pending disputes, motions and applications and will likely limit disputes in the future. As explained above, settlements of disputes benefit the public by reducing the costs and expense of litigation and conserving Commission resources. In addition, because there are pending disputes at the California Court of Appeal and likely will be disputes at FERC, the Settlement Agreement also preserves the resources of the courts and FERC.

Second, the Settlement Agreement creates a framework for a QF/CHP Program going forward that is much more closely aligned with other Commission-approved procurement processes. For example, under the Settlement Agreement, the IOUs will initiate a CHP RFO

<sup>&</sup>lt;sup>25</sup> Term Sheet, § 4.8.

process, which is similar to how conventional and RPS resources are now procured. The Settlement Agreement also includes *Pro Forma* PPAs, which will allow CHP developers and the IOUs to reduce transaction costs and resources, which they would otherwise be expended in the time-consuming process of negotiating individual PPAs.

Third, the Settlement Agreement will encourage the continued operation of the State's existing CHP facilities, and the development, installation and interconnection of new, clean and efficient CHP facilities in order to increase the diversity, reliability and environmental benefits of the CHP energy resources available to the State's electricity consumers.

Fourth, the Settlement Agreement creates a framework for achieving CARB's current CHP goals for the reduction of GHG emissions. GHG emissions pose a serious threat to the California economy, environment and the health and welfare of California's citizens. By providing a framework for the implementation of one aspect of the CARB Scoping Plan, the Settlement Agreement will facilitate efforts for California to meet its ambitious AB 32 goals. The Settlement Agreement encourages the retirement of existing, inefficient CHP facilities or repowering existing CHP facilities to make them more clean and efficient, and the development of new, clean and efficient CHP.

Fifth, the Settlement Agreement adopts a methodology for determining SRAC energy prices that is consistent with Commission decisions. The Settlement Agreement also provides for CHP PPA energy prices that are determined as a part of a competitive process, so that the prices accurately reflect a market price. Customers will benefit from clearly established SRAC prices, or prices determined through a competitive process. In addition, the capacity prices adopted in the Settlement Agreement have already been approved by the Commission.

Sixth, the Settlement Agreement creates a transparent procurement process. The Commission, interested parties and the public all benefit from a transparent procurement process with appropriate protections for confidential IOU information.

Seventh, the Settlement Agreement establishes clear rules for pricing and treatment of existing QF PPAs. For example, under the Settlement, QFs with existing PPAs are encouraged to provide forecasting information to the IOUs so that the IOUs can more accurately forecast QF generation. QFs also have greater certainty as the SRAC formula is clearly established rather than being subject to continued and ongoing disputes.

Finally, the Settlement Agreement provides for the equitable allocation of costs associated with the QF/CHP program to all Commission-jurisdictional LSEs.

# IV. THE JOINT PARTIES HAVE COMPLIED WITH THE REQUIREMENTS OF RULE 12.1(B)

Commission Rule 12.1(b) requires parties to provide a notice of a settlement conference at least seven (7) days before a settlement is signed. On September 24, 2010, the Joint Parties notified all of the parties on the service list in these proceedings of a settlement conference and subsequently convened the settlement conference on October 7, 2010 to describe and discuss the terms of the proposed Settlement Agreement. Representatives of the Joint Parties participated in the settlement conference. After the settlement conference was concluded, the Settlement Agreement was finalized and executed on October 8, 2010.

#### V. HEARINGS ARE NOT REQUIRED

The Joint Parties respectfully request that the Commission approve the Settlement

Agreement without evidentiary hearings as there are no disputed issues of material fact related to
the Settlement Agreement that require hearings. In addition, hearings would prevent the
expeditious approval of the Settlement Agreement. Should evidentiary hearings be deemed

necessary, the Joint Parties request that such hearings be held at the earliest opportunity, and concluded in a speedy and efficient manner.

# VI. TIMING FOR REVIEW OF THE SETTLEMENT AGREEMENT AND CONDITIONS PRECEDENT FOR THE SETTLEMENT AGREEMENT TO BECOME EFFECTIVE

In a separate Motion for Expedited Consideration, which is being filed concurrently with this Motion, the Joint Parties have requested the Commission expeditiously review and approve the Settlement Agreement. Expeditious review and approval of the Settlement Agreement will allow the IOUs to proceed with filing of the FERC application described in Section 15 of the Settlement Agreement and to obtain written support from CARB. FERC approval of an application for termination of the PURPA purchase obligation and CARB written support are conditions precedent to the Settlement Agreement becoming effective. However, because the IOUs cannot file an application at FERC until after the Commission approves the Settlement Agreement, <sup>76</sup> expeditious review of the Settlement Agreement by the Commission is a necessary first step in satisfying all of the conditions precedent.

One of the conditions precedent for the Settlement Agreement to become effective is a Commission decision approving the Settlement Agreement "as submitted for approval without revisions unacceptable to any Party or in an alternative form that is acceptable to all Parties."

The Joint Parties strongly urge the Commission to adopt the Settlement Agreement as is, without modification, and to select one of the two identified options for participation by ESPs and CCAs and their customers. If a Commission decision proposes modifications to the Settlement Agreement, the Joint Parties will then need to review and agree to the modifications before the condition precedent of Commission approval is satisfied. Given that it has taken the Joint Parties

<sup>&</sup>lt;sup>76</sup> Term Sheet, § 15.1.6.

<sup>&</sup>lt;sup>27</sup> Id. at § 16.2.2.

more than a year and a half to negotiate the Settlement, and the Settlement Agreement involves a complex series of compromises and agreements, a Commission modification of the Settlement Agreement is likely to result in months of additional delay and may ultimately result in the Joint Parties being unable to agree to the modifications and the Settlement Agreement terminating. In light of the substantial benefits of the Settlement Agreement, the Commission should approve the Settlement Agreement as is, without modification, to avoid further delay negotiating the modifications or, potentially, termination of the Settlement Agreement as a result of the proposed modifications being unacceptable to the Joint Parties.

The Joint Parties are proposing in their Motion for Expedited Consideration the following schedule for consideration of the Settlement Agreement:

Event	Dates Per The Commission's Rules	Proposed Dates
Motion for Approval of Settlement Agreement		Filed October 8, 2010
Comments on Motion for Approval of Settlement Agreement (Rule 12.2.)	November 8, 2010	October 25, 2010
Reply Comments on Motion for Approval of Settlement Agreement (Rule 12.2.)	November 23, 2010	November 1, 2010
ALJ's Proposed Decision (Rule 14.2.)	-	November 16, 2010
Comments on Proposed Decision (Rule 14.3(a).)	20 days after Proposed Decision	December 6, 2010
Reply comments on Proposed Decision (Rule 14.3(d).)	5 days after opening comments on Proposed Decision	December 13, 2010
Commission vote on Proposed Decision		December 16, 2010

#### VII. CONCLUSION

As demonstrated above, the Settlement Agreement is reasonable in light of the whole record, consistent with law, and in the public interest. Thus, the Joint Parties respectfully request that the Commission: (1) approve the Settlement Agreement without modification; (2) approve the *Pro Forma* PPAs attached to the Settlement Agreement without modification; and (3) determine that the decision approving the Settlement Agreement supersedes certain existing Commission decisions identified in Sections 16.2.4, 16.2.5 and 16.2.6 of the Settlement Agreement.

///

///

///

Respectfully submitted,

Michael D. Montoya

Carol A. Schmid-Frazee

Southern California Edison Company

2244 Walnut Grove Avenue Rosemead, CA 91770-3714 Telephone: (626) 302-1337 Facsimile: (626) 302-1935

E-Mail: Carol.SchmidFrazee@sce.com

# Attorneys for Southern California Edison Company

Georgetta J. Baker

San Diego Gas & Electric Company

101 Ash Street, HQ12 San Diego, CA 92101-3017 Telephone: (619) 699-5064 Facsimile: (619) 699-5027 E-Mail: gbaker@sempra.com

Attorney for San Diego Gas & Electric Company

Jerry R-Bloom

Winston & Strawn LLP 333 So. Grand Avenue

Los Angeles, CA 90071-1504 Telephone: (213) 615-1700 Facsimile: (213) 615-1750 E-Mail: jbloom@winston.com

Attorney for California Cogeneration Council

Mary A. Gandesbery

Charles R. Middlekauff

Evelyn C. Lee

Pacific Gas and Electric Company

77 Beale Street

San Francisco, CA 94105-1814 Telephone: (415) 973-0675 Facsimile: (415) 973-5520 E-Mail: magq@pge.com

Attorney for Pacific Gas and Electric Company

Michel Peter Florio

The Utility Reform Network 115 Sansome Street, Suite 900 San Francisco, CA 94104-3624 Telephone: (415) 929-8776, ext. 302

Facsimile: (415) 929-1132 E-Mail: mflorio@turn.org

Attorney for The Utility Reform Network

our Ktert/a

Douglas K. Kerner

Ellison, Schneider & Harris

2600 Capitol Avenue, Suite 400

Sacramento, CA 95816-5931 Telephone: (916) 447-2166 Facsimile: (916) 446-3512

E-Mail: dkk@eslawfirm.com

Attorney for Independent Energy Producers Association

Michael P. Alcantar Alcantar & Kahl

33 New Montgomery St., Suite 1850

San Francisco, CA 94105-4511 Telephone: (415) 421-4143

Facsimile: (415) 989-1263 E:mail: mpa@a-klaw.com

Attorney for the Cogeneration Association of California and The Energy Producers and Users Association

Lisa-Marie G. Salvacion

Division of Ratepayer Advocates

505 Van Ness Avenue

San Francisco, CA 94102-3214

Telephone: (415) 703-2069 Facsimile: (415) 703-20**6**7 E-Mail: lms@cpuc.ca.gov

Attorney for California Public Utilities Commission, Division of Ratepayer Advocates

October 8, 2010

# ATTACHMENT A SETTLEMENT AGREEMENT

# BEFORE THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA

Application of Southern California Edison Company (U 338-E) for Applying the Market Index Formula and As-Available Capacity Prices Adopted in D.07-09-040 to Calculate Short-Run Avoided Cost for Payments to Qualifying Facilities Beginning July 2003 and Associated Relief

Order Instituting Rulemaking to Integrate Procurement Policies and Consider Long-Term Procurement Plans

Order Instituting Rulemaking to Promote Policy and Program Coordination and Integration in Electric Utility Resource Planning

Order Instituting Rulemaking to Promote Consistency in Methodology and Input Assumptions in Commission Applications of Short-Run and Long-Run Avoided Costs, Including Pricing for Qualifying Facilities

Order Instituting Rulemaking into Implementation of Public Utilities Code Section 390

Application 08-11-001 (Filed November 4, 2008)

Rulemaking 06-02-013 (Filed February 16, 2006)

Rulemaking 04-04-003 (Filed April 1, 2004)

Rulemaking 04-04-025 (Filed April 22, 2004)

Rulemaking 99-11-022 (Filed November 18, 1999)

#### CHP PROGRAM SETTLEMENT AGREEMENT

The parties to this CHP Program Settlement Agreement<sup>1</sup> are Pacific Gas and Electric Company ("PG&E"), a California corporation, Southern California Edison Company ("SCE"), a California corporation, and San Diego Gas & Electric Company ("SDG&E"), a California corporation (hereinafter collectively referred to as "Investor Owned Utilities" or "IOUs"), The Utility Reform Network ("TURN") and the Division of Ratepayer Advocates ("DRA") (hereinafter collectively referred to as "Consumer Interest Groups"), and California Cogeneration Council and all of its members ("CCC"),

<sup>&</sup>lt;sup>1</sup> The term Settlement Agreement as used herein includes the CHP Program Settlement Term Sheet and Exhibits 1 through 11 listed below.

Cogeneration Association of California and all of its members ("CAC"), Energy Producers and Users Coalition and all of its members ("EPUC"), and the Independent Energy Producers Association ("IEP") (hereinafter collectively referred to as qualifying facilities ("QF") parties ("QF Parties")). All of these entities are hereinafter referred to collectively as the "Parties" or individually as a "Party."

#### **RECITALS**

The IOUs are all investor-owned public utilities in the State of California and all are subject to the jurisdiction of the California Public Utilities Commission ("Commission") with respect to providing electric service to their customers.

The Consumer Interest Groups consist of DRA and TURN. DRA is an independent division of the Commission that advocates solely on behalf of utility ratepayers. TURN is an independent, non-profit consumer advocacy organization that represents the interests of residential and small commercial utility customers.

The QF Parties consist of CCC, CAC, EPUC, and IEP. The QF Parties are organizations that represent, *inter alia*, the interests of cogeneration Qualifying Facility operations and combined heat and power ("CHP") facilities in the State of California. On September 20, 2007, the Commission issued D.07-09-040, in R.04-04-025/R.04-04-003, that established Short-Run Avoided Cost ("SRAC") energy and as-available and firm capacity pricing for QFs and ordered the IOUs and QF Parties to work together to develop a Standard QF Contract through negotiations in which the Commission's Energy Division took an active role. On August 1, 2009, the SRAC energy and as-available capacity pricing established in D.07-09-040 was implemented pursuant to Resolution No. E-4246, dated July 9, 2009.

On October 25, 2007, the IOUs and TURN filed an Application for Rehearing of D.07-09-040, CAC and EPUC filed an Application for Rehearing of D.07-09-040, and CCC filed an Application for Rehearing of D.07-09-040. In response to these

Applications for Rehearing, the Commission issued D.08-07-048, modifying D.07-09-040. In response to D.08-07-048, SCE and TURN filed a Petition for Writ of Review with the California Court of Appeals.

D.08-07-048 allowed the IOUs to seek retroactive application of the energy and as-available capacity pricing for QFs adopted in D.07-09-040. On October 3, 2008, the QF Parties filed a Petition for Modification requesting that the Commission modify D.08-07-048 to eliminate the opportunity for the IOUs to seek retroactive application of the energy and as-available capacity pricing adopted in D.07-09-040 as modified by D.08-07-048. D.08-07-048 established a deadline of November 4, 2008 for the IOUs to file applications for retroactive application of the SRAC energy and as-available capacity pricing.

On November 4, 2008, SCE filed its SRAC Update Application (A.08-11-001) which, pursuant to requests of the QF Parties, was ultimately held in abeyance until after issuance of a decision on the QF Parties' October 3 Petition for Modification of D.08-07-048. On that same date, November 4, 2008, PG&E and SDG&E filed notices of reservations of rights to later file claims to recover amounts exceeding the SRAC energy and as-available capacity pricing adopted on September 20, 2007, but indicating that they would not request recovery of any such amounts paid prior to September 20, 2007.

In December 2008, pursuant to Assembly Bill ("AB") 32, the California Air Resources Board ("CARB") adopted the Climate Change Scoping Plan. In part, that plan embraced a statewide target to reduce 6.7 million metric tons ("MMT") of Greenhouse Gas ("GHG") from the incremental development of Combined Heat and Power ("CHP") facilities.

On January 26, 2009, Administrative Law Judge ("ALJ") DeBerry issued a ruling suggesting proposed principles as a way of resolving the dispute over the reasonableness of PG&E's and SCE's payments to QFs during the period December 2000 through March 2001 ("Proposed Principles"). This dispute was remanded to the Commission by the

California Court of Appeal in Southern California Edison Co. v. Public Utility

Commission, 101 Cal. App 4th 982 (2002) (Remand Dispute). On May 1, 2009, Opening

Comments on the Proposed Principles were filed with the Commission, and, on May 15,

2009, Reply Comments on the Opening Comments were filed with the Commission.

On April 16, 2009, the Commission issued D.09-04-032 which modified D.08-07-048 concerning the showing that the IOUs must make in support of any requests for retroactive application of changes to SRAC. D.09-04-032 also allowed SCE to amend its SRAC Update Application by May 7, 2009 which SCE did, and gave PG&E and SDG&E 45 days to submit their applications. On May 7, 2009, SCE filed its amended SRAC Update Application. PG&E and SDG&E have requested and been granted by the Commission's Executive Director multiple extensions for the filing of their applications to accommodate settlement negotiations.

The lOUs plan to submit an application to the Federal Energy Regulatory Commission ("FERC") pursuant to Section 210(m) of the Public Utility Regulatory Policies Act ("PURPA") to terminate the IOUs' QF purchase obligation.

On May 18, 2009, the Parties commenced settlement negotiations of outstanding QF and CHP issues before the Commission and FERC, and resulting from the CARB AB 32 Scoping Plan. These negotiations continued for over a year. This Settlement Agreement is the result of those negotiations.

#### **TERMS OF AGREEMENT**

In consideration of the mutual obligations, promises, covenants, and conditions contained herein, the Parties agree to the terms and conditions of this Settlement Agreement and agree to support its approval by the Commission.

Each Party shall review any Commission orders regarding this Settlement

Agreement to determine if the Commission has changed, modified, or severed any
portion of the Settlement Agreement, deleted a term, or imposed a new term. If a Party is

unwilling to accept such change, modification, severance, deletion, or addition of a new term of the Settlement Agreement, that Party shall so notify the other Parties within ten (10) business days of issuance of any such Commission order regarding this Settlement Agreement. The Parties shall thereafter promptly discuss each change, modification, severance, deletion or new term found unacceptable and negotiate in good faith to achieve a resolution acceptable to all Parties and promptly seek Commission approval of the resolution so achieved. Failure to resolve such change, modification, severance, deletion, or new term to this Settlement Agreement to the satisfaction of all Parties within ninety (90) calendar days of notification, and to obtain Commission Approval of such resolution promptly thereafter, shall cause this Settlement Agreement to terminate.

This Settlement Agreement is the result of extended negotiations. It represents a compromise of disputed claims between the Parties, which are identified in Section 14 of the Term Sheet. The Parties have reached this Settlement Agreement after taking into account the possibility that each Party may or may not prevail on any given issue. This Settlement Agreement is reasonable in light of the whole record, consistent with law, and in the public interest.

As provided under Rule 12.5 of the Commission's Rules of Practice and Procedure, the Parties request that the Commission expressly find the Settlement Agreement Term Sheet is precedential.

This Settlement Agreement includes the CHP Program Settlement Agreement

Term Sheet ("Term Sheet") and Exhibits 1-11, each of which is attached and incorporated

by reference into this document. Exhibits 1-11 are as follows:

- Amendment to Legacy QF Power Purchase Agreement ("PPA") for PG&E;
- 2. Amendment to Legacy QF PPA for SCE;
- 3. Amendment to Legacy QF PPA for SDG&E;
- 4. Transition PPA;

- 5. CHP Request for Offer ("RFO") Pro Forma PPA;
- 6. QF PPA for QFs 20 MW or Less;
- 7. Optional As-Available PPA;
- 8. Non-Disclosure Agreement ("NDA") for CHP Auditor;
- 9. List of Members of CAC;
- 10. List of Members of CCC; and
- 11. List of Members of EPUC.

The Settlement Agreement contains the entire agreement and understanding between the Parties as to the subject matter of this agreement, and supersedes all prior agreements, commitments, representations, and discussions between the Parties. In the event there is any conflict between the terms and scope of the Settlement Agreement and the terms and scope of the accompanying *Joint Motion for Approval of the Qualifying Facility and Combined Heat and Power Settlement Agreement*, the Settlement Agreement shall govern. In the event there is any conflict between the terms and scope of the Term Sheet on contract issues and the attached PPAs (Exhibits 1-7), the PPAs shall govern.

None of the provisions of this Settlement Agreement shall be considered waived by any Party unless such waiver is given in writing. The failure of a Party to insist in any one or more instances upon strict performance of any of the provisions of this Settlement Agreement or to take advantage of any of their rights hereunder shall not be construed as a waiver of any such provisions or the relinquishment of any such rights for the future, but the same shall continue and remain in full force and effect.

This Settlement Agreement shall be interpreted, governed and construed under the laws of the State of California, including Commission decisions, orders and rulings as if executed and performed wholly within the State of California.

PACIFIC GAS and ELECTRIC	SAN DIEGO GAS & ELECTRIC
COMPANY	COMPANY
A California Corporation	A California Corporation
0010010	
By Mustofky Potus	By
Title President	Title
Date 10/8/10	Date
SOUTHERN CALIFORNIA EDISON	THE UTILITY REFORM NETWORK
COMPANY	
A California Corporation	By
	Title
By Wan 1	Date
Title Nice President Renewable & Alternation	•
Date 10/8/2010 Power	
INDEPENDENT ENERGY	DIVISION OF RATEPAYER
PRODUCERS ASSOCIATION	ADVOCATES
D.,	D.
By	By
Title	Title
Date	Date

PACIFIC GAS and ELECTRIC	SAN DIEGO GAS & ELECTRIC
COMPANY	COMPANY
A California Corporation	A California Corporation
By	By Man Bushawa Title VP-Elect & Fuel Procurement Date 10/8/10
SOUTHERN CALIFORNIA EDISON COMPANY	THE UTILITY REFORM NETWORK
A California Corporation	<u>By</u>
	Title
By	Date
Title	
Date	
INDEPENDENT ENERGY PRODUCERS ASSOCIATION	DIVISION OF RATEPAYER ADVOCATES
Ву	Ву
Title	Title
Date	-Date

PACIFIC GAS and ELECTRIC COMPANY A California Corporation  By Title Date  SOUTHERN CALIFORNIA EDISON COMPANY A California Corporation  By Title Date	SAN DIEGO GAS & ELECTRIC COMPANY A California Corporation  By Title Date  THE UTILITY REFORM NETWORK  By TitleSenior Attorney Date DateOctober 8, 2010
INDEPENDENT ENERGY PRODUCERS ASSOCIATION	DIVISION OF RATEPAYER ADVOCATES
ByTitle	ByTitle
Date	Date

7

PACIFIC GAS and ELECTRIC COMPANY A California Corporation	SAN DIEGO GAS & ELECTRIC  COMPANY  A California Corporation
By	By Title Date
SOUTHERN CALIFORNIA EDISON COMPANY A California Corporation  By Title Date	ByTitleDate
INDEPENDENT ENERGY PRODUCERS ASSOCIATION  By Title Date	By Action Director Division of Respect

10/08/2010 09:00

This Settlement Agreement is executed in nine counterparts, each of which shall be deemed an original. The undersigned represent that they are authorized to sign on behalf of the Party represented.

PACIFIC GAS and ELECTRIC	SAN DIEGO GAS & ELECTRIC
COMPANY	COMPANY
A California Corporation	A California Corporation
By	By
Title	Title
Date	Date
)	
SOUTHERN CALIFORNIA EDISON	THE UTILITY REFORM NETWORK
COMPANY	
A California Corporation	By_
``	Title
By	Date
Title	
Date	
INDEPENDENT ENERGY	DIVISION OF RATEPAYER
PRODUCERS ASSOCIATION	ADVOCATES
By Bull	By
Title Altono tor	Title
By Delan for Date 10 / 08/10	Date

COGENERATION ASSOCIATION OF	CALIFORNIA COGENERATION
CALIFORNIA / //	COUNCIL
my (bolk the	
By Michael & Raylo	By M. Bloom
Title Coonsel	Title Coursel
Date October 7,2010	Date october 7, 2010
	,
	ENERGY PRODUCERS AND USERS
	COALITION OF
	By Window Hearton
	Title CoorteL
	Date October 7, 2010

# **CHP Program Settlement Agreement Term Sheet**

### October 8, 2010

СНР	CHP Program Settlement Agreement		
1	Goals and Objectives	5	
1.1	Settlement Goals and Objectives	5	
1.2	State CHP Program	5	
2	Settlement Periods	7	
2.1	Transition Period	7	
2.2	Initial Program Period and Targets	8	
2.3	Second Program Period and Targets	8	
3	Transition PPA Matters	8	
3.1	Eligibility for a Transition PPA	8	
3.2	Pricing of Transition PPAs	9	
3.3	Form of Transition PPA	9	
3.4	Modifications	9	
4	CHP Procurement Processes	12	
4.1	Overview	12	
4.2	CHP RFOs	12	
4.3	Bilaterally Negotiated PPAs.	20	
4.4	AB 1613 Feed-In Tariff	21	
4.5	PURPA Program for QFs 20 MW or Less (QF PPAs)	21	
4.6	As-Available Procurement Alternatives and Optional As-Available PPA	21	
4.7	10U-Owned CHP	25	
4.8	Utility Prescheduled Facilities	25	
4.9	New Behind the Meter CHP Facilities	25	

4.10	Approval of PPAs	25
<b>4</b> .11	Seller Reporting Requirements for PPAs	26
4.12	Timing of Commencement of Delivery	26
5	MW Targets	26
5.1	IOUs' MW Targets	26
5.2	MW Counting Rules	28
5.3	CHP Program MW Counting Precedent or Use	29
5.4	Justification for Failure to Meet MW Targets	29
6	GHG Emissions Reduction Targets	30
6.1	Strategy	30
6.2	IOUs' GHG Emissions Reduction Targets	30
6.3	ESP and Community Choice Aggregator (CCA) Portion of the CARB CHP RRM	31
6.4	Method to Determine Each IOU's GHG Emissions Reduction Target	32
6.5	Each IOU's GHG Emissions Reduction Target for the Second Program Period will be calculated and submitted to the CPUC by the IOUs in their Semi-Annual CHP Program Reports.	33
6.6	The IOUs' GHG Emissions Reduction Target for the Second Program Period is subject to review and revision in the LTTP process.	33
6.7	Changes to the CARB CHP RRM	33
6.8	Advocating at CARB and in other Forums	33
6.9	Justification for Failure to Meet GHG Emissions Reduction Targets	34
7	GHG Emission Accounting Methodology	34
7.1	GHG Accounting Principles.	34
7.2	The Double Benchmark, which may be later modified pursuant to this Settlement, is as follows:	35
7.3	Detailed GHG Accounting Methodology to Measure Progress Toward the IOUs' GHG Emissions Reduction Targets	35
7.4	Effective Date for Accounting of Changes in GHG Credits and GHG Debits	37
Я	CPUC Jurisdictional Entities' Semi-Annual CHP Program Reports	37

8.1	General Description	37
8.2	Overview of CHP Program Report Content	38
8.3	Report Content for New PPAs	38
8.4	Report Content for Legacy PPA C1 Amendments, Utility Prescheduled Facilities or Other Changes in Operations	40
8.5	Report Content for Terminated PPAs and Facilities that Cease Operations	40
9	CHP Auditor	41
9.1	Description of CHP Auditor	41
9.2	IOU Notice Triggering Audit	43
9.3	Time Period for Audit Review	43
9.4	Receipt and Review of Confidential Information and Other Relevant Data	43
9.5	Designation, Notice, Conflict of Interest Review and Number of CHP Auditors	44
10	SRAC Energy Pricing Structure	45
10.1	Applicability	45
10.2	Methodologies and Formulae	45
10.3	Reporting Requirements	50
10.4	Market Disruption Event	51
10.5	Seller's Responsibility	51
11	Legacy PPA Matters for All Existing QFs	52
11.1	Energy and Capacity Pricing	52
11.2	Other Matters	52
11.3	Non-Binding Forecasting Requirements for Legacy PPAs	53
12	CAISO Tariff Compliance	55
12.1	CAISO Tariff Compliance for New PPAs	55
13	IOU Cost Recovery for CHP Program PPAs	55
13.1	Cost Allocation and Departing Load Charges	55
13.2	IOUs' Energy Resources Recovery Accounts	5 <del>€</del>

14	Settlement of Pending and Anticipated Litigation	57
14.1	Retroactive Adjustment of SRAC Prices	57
14.2	Released Claims	57
14.3	Confidentiality	58
15	Federal Energy Regulatory Commission 210(m) Application	58
15.1	PURPA §210(m) Application at the FERC	58
15.2	FERC Reinstatement of PURPA Obligation	60
16	Conditions Precedent and Subsequent to Settlement Effective Date	61
16.1	Approval of Settlement by CPUC	61
16.2	Conditions Precedent to Effectiveness of the Settlement	61
16.3	Conditions Subsequent to Effectiveness of Settlement	61
17	Glossary of Defined Terms	63

#### CHP Program Settlement Agreement: Term Sheet

#### 1 Goals and Objectives

- 1.1 Settlement Goals and Objectives
  - 1.1.1 Develop a State combined heat and power (CHP) program (CHP Program).
  - 1.1.2 Create a smooth transition from the existing QF CHP PURPA Program to a State-administered CHP Program.
  - 1.1.3 Settle all CHP/QF litigation referenced in Section 14.
- 1.2 State CHP Program
  - 1.2.1 Policy Objectives
    - 1.2.1.1 Section 372(a) of the California Public Utilities Code states: "it is the policy of the state to encourage and support the development of eogeneration technology as an efficient, environmentally beneficial, competitive energy resource that will enhance the reliability of local generation supply, and promote local business growth."
    - 1.2.1.2 The Energy Action Plan II states: "The loading order identifies energy efficiency and demand response as the State's preferred means of meeting growing energy needs. After cost effective efficiency and demand response, we rely on renewable sources of power and distributed generation, such as combined heat and power applications. To the extent efficiency, demand response, renewable resources, and distributed generation are unable to satisfy increasing energy and capacity needs, we support clean and efficient fossil-fired generation."
    - 1.2.1.3 The purpose of the State CHP Program is to encourage the continued operation of the State's Existing CHP Facilities, and the development, installation, and interconnection of new, clean and efficient CHP Facilities, in order to increase the diversity, reliability, and environmental henefits of the energy resources available to the State's electricity consumers.
    - 1.2.1.4 These policies and purposes will be achieved by a State CHP Program that procures CHP as set forth in this Settlement, retains existing efficient CHP, supports the change in operations of inefficient CHP to provide greater benefits to the State, and replaces CHP that will no longer be under contract with the IOUs with new, efficient CHP.
    - 1.2.1.5 In addition, this State CHP Program will secure additional Greenhouse Gas (GHG) emissions reduction benefits, consistent with the reduction targets of Assembly Bill (AB) 32, by adding new, efficient CHP.

#### 1.2.2 CHP Program Objectives

- 1.2.2.1 Greater regulatory and market certainty for CHP Facilities;
- 1.2.2.2 Retains existing CHP GHG emissions reduction benefits;
- 1.2.2.3 Encourages the upgrade of the inefficient CHP Facilities in the IOUs' electric portfolios into efficient CHP through repowering or change of operations;
- 1.2.2.4 Provides an orderly exit strategy for CHP Facilities that cannot participate, or are unsuccessful, in the new CHP Program;
- 1.2.2.5 Encourages the development of new, clean and efficient CHP;
- 1.2.2.6 Recognizes the distinct products provided by CHP in order to procure costeffective and efficient CHP power;
- 1.2.2.7 To the extent available, sustains and enhances reductions in GHG emissions from existing facilities and incrementally adds new CHP with the goal of reducing GHG emissions consistent with AB 32;
- 1.2.2.8 Achieves other benefits for California and electricity consumers in the IOUs' Service Territories;
- 1.2.2.9 Establishes a platform for a State CHP Program with identified features through 2020, and sets a framework for a sustained State CHP Program beyond 2020; and
- 1.2.2.10 Provides Power Purchase Agreement (PPA) options for CHP Facilities.
- 1.2.3 Facility Owner Goals and Objectives
  - 1.2.3.1 On-site reliability;
  - 1.2.3.2 Cost control;
  - 1.2.3.3 Improvement in business and regulatory certainty;
  - 1.2.3.4 Fuel efficiency through the use of a single fuel to produce two energy products, specifically thermal and electrical energy;
  - 1.2.3.5 Compliance with GHG regulations; and
  - 1.2.3.6 Maintaining CHP Program certainty.
- 1.2.4 Societal Goals and Objectives
  - 1.2.4.1 Fuel efficiency through the use of a single fuel to produce two energy products, specifically thermal and electrical energy;
  - 1.2.4.2 Reduction in GHG emissions and criteria pollutant emissions from avoided combustion of fossil fuel:

- 1.2.4.3 Grid reliability and/or local area reliability, including self-service of power supply for facilities' loads in resource-constrained areas;
- 1.2.4.4 Reduction in transmission and distribution energy losses;
- 1.2.4.5 Securing CHP generation resources as part of a diversified portfolio of resources for California generation supply; and
- 1.2.4.6 Support for California's manufacturing, industrial and commercial base without cross subsidization by electric customers.
- 1.2.5 Bundled IOU Customer Goals and Objectives
  - 1.2.5.1 Securing cost-effective GHG reductions;
  - 1.2.5.2 Providing system, grid, and local area reliability;
  - 1.2.5.3 Reducing transmission and distribution losses and investment;
  - 1.2.5.4 Moving to viable, market-based compensation for QFs with Legacy PPAs and CHP resources to sustain California CHP operations at fair prices; and
  - 1.2.5.5 Complementing other existing State policy programs that reduce GHG emissions.
- 1.2.6 GHG Emissions Reduction Goals and Objectives
  - 1.2.6.1 Maintain existing GHG benefits in the IOUs' portfolios attributable to CHP; and
  - 1.2.6.2 Achieve additional GHG emissions reductions beyond those benefits already provided by the existing CHP fleet towards the California Air Resources Board (CARB) Scoping Plan statewide target goal of 6.7 million metric tons (MMT) from efficient CHP. In the Settlement this is referred to as the CARB Combined Heat and Power Recommended Reduction Measure (CARB CHP RRM).<sup>1</sup>

#### 2 Settlement Periods

#### 2.1 Transition Period

2.1.1 The Transition Period is a period in which a CHP Facility will either obtain a new PPA as per Section 4, sell into the wholesale market, shut down, or cease to export to the grid.

As referenced on pages 43-44 of the "Approved Scoping Plan" adopted by the CARB at its December 11, 2008 Board meeting. <a href="http://www.arb.ca.gov/cc/scopingplan/document/adopted\_scoping\_plan.pdf">http://www.arb.ca.gov/cc/scopingplan/document/adopted\_scoping\_plan.pdf</a>

- 2.1.2 The Transition Period will begin on the Settlement Effective Date and shall not extend beyond July 1, 2015.
- 2.2 Initial Program Period and Targets
  - 2.2.1 The Initial Program Period shall commence on the Settlement Effective Date, and shall conclude 48 months thereafter.
  - 2.2.2 Each IOU shall enter into new PPAs with CHP Facilities and Utility Prescheduled Facilities as further described in Section 5 as follows:
    - 2.2.2.1 SCE: 1,402 MW
    - 2.2.2.2 PG&E: 1,387 MW
    - 2.2.2.3 SDG&E: 160 MW.
- 2.3 Second Program Period and Targets
  - 2.3.1 The Second Program Period shall commence at the end of the Initial Program Period and end December 31, 2020.
  - 2.3.2 For CHP procurement in the Second Program Period, the IOUs will procure the following:
    - 2.3.2.1 Any portion of the IOUs' MW Targets that was not attained in the Initial Program Period.
    - 2.3.2.2 SDG&E shall procure an additional 51 MW.
    - 2.3.2.3 Additional CHP capacity to meet the IOUs' GHG Emissions Reduction Targets as established by the CPUC in the Long Term Procurement Planning (LTPP) proceeding, taking into account the progress toward the MW Targets in the Initial Program Period.

#### 3 Transition PPA Matters

- 3.1 Eligibility for a Transition PPA
  - 3.1.1 A CHP Facility currently selling to an IOU under a Legacy PPA or an extension thereof that is expiring during the Transition Period, may sign a Transition PPA with the same IOU-Buyer.
  - 3.1.2 The Transition PPA begins upon the expiration of the Legacy PPA or extensions of a Legacy PPA and ends at the election of the Seller but no later than the last day of the Transition Period.
  - 3.1.3 The capacity and energy that the CHP Facility may sell to the IOU under the Transition PPA are limited to an amount consistent with the QF's historical deliveries under its Legacy PPA, but energy delivery may be lower upon the election of the Seller.

3.1.4 The Transition PPA is unavailable to CHP Facilities whose owners elect under the CHP Program to sell to the lOUs outside of the competitive solicitation process, including CHP Facilities selling under a bilateral PPA, the Optional As-Available PPA, and CHP Facilities eligible to sell to the lOUs pursuant to the AB 1613 Feed-In Tariff. During the Settlement Term, QFs who elect to sign a Transition PPA waive their rights to sign a CHP PPA that is not obtained through competitive procurement, bilateral negotiations or the under 20 MW nameplate PURPA must-take obligation.

#### 3.2 Pricing of Transition PPAs

- 3.2.1 Capacity prices shall be paid as established in Decision 07-09-040.
- 3.2.2 Energy pricing will be Short Run Avoided Cost (SRAC) as defined in Section 10 below.

#### 3.3 Form of Transition PPA

3.3.1 The Transition PPA is the QF Standard Offer Contract (SOC) modified for the Transition Period. The Standard Form Transition PPA is attached hereto as Exhibit 4.

#### 3.4 Modifications

- 3.4.1 Modifications of the SOC for the Transition PPA are described in part as follows: (The terms and conditions of the attached Transition PPA govern)
  - 3.4.1.1 Double penalties under the Transition PPA and CAISO Tariff for failure to deliver capacity for Transition PPA and the CHP RFO Pro Forma PPA.

Under both the applicable CAISO Tariff and the PPAs there are penalties for failure to deliver capacity or scheduled energy. Parties agree that only a single, and not double, penalty shall apply to such deliveries and schedules.

Accordingly, the CAISO Tariff penalties pursuant to the Standard Capacity Product shall be treated as follows:

If Buyer is the Scheduling Coordinator, and if the Generating Facility is subject to the terms of the Availability Standards, Non-Availability Charges, and Availability Incentive Payments as contemplated under Section 40.9 of the CAISO Tariff, then any Availability Incentive Payments will be for the benefit of Buyer and for Buyer's account and any Non-Availability Charges will be the responsibility of Buyer and for Buyer's account.

If Buyer is not the Scheduling Coordinator, and if the Generating Facility is subject to the terms of the Availability Standards, Non-Availability Charges, and Availability Incentive Payments as contemplated under Section 40.9 of the CAISO Tariff, then any Availability Incentive Payments will be for the benefit of Seller and for Seller's account and any Non-Availability Charges will be the responsibility of Seller and for Seller's account.

3.4.1.2 Sale of Additional Dispatchable Capacity beyond the Transition PPA Capacity Product

This option for Additional Dispatchable Capacity is viewed as being limited to a few CHP facilities, each with unique operational constraints. A specific amendment to the Transition PPA is required to accommodate Additional Dispatchable Capacity.

The Transition PPA shall provide that a Seller may elect to deliver a standard capacity product (including associated energy and RA with such capacity) at the Transition PPA firm capacity and energy prices or as-available capacity and energy prices. In addition to these standard products, a Seller may elect to sell to Buyer under a Transition PPA Additional Dispatchable Capacity above the standard contract capacity set forth in the Transition PPA (Additional Dispatchable Capacity). Buyer must negotiate in good faith for 120 days to amend the Transition PPA to incorporate a competitive market price for the Additional Dispatchable Capacity. If negotiations are unsuccessful, Buyer and Seller will mediate the terms of the amendment using the mediation procedures set forth in Section 10.02 of the Transition PPA Within ninety (90) days after the Transition PPA is executed by Buyer and Seller, Seller shall designate the initial Additional Dispatchable Capacity offered to Buyer for the term of the PPA. In addition, such Additional Dispatchable Capacity will be offered with an associated fixed heat rate, or fixed heat rate curve, established by the Seller. The fixed heat rate or fixed heat rate curve will be used as the applicable heat rate in the energy price formula calculated pursuant to Section 10.2.1.1. Additional Dispatchable Capacity must meet CPUC/CAISO requirements for RA and such capacity must be offered and provided in a manner consistent with CAISO Tariff and Protocols. Parties will work in good faith to get clarification from CAISO to implement this section according to the CAISO Tariff and Protocols on a case-by-case basis in advance of any deliveries of electric power pursuant to this provision.

In order to be eligible to provide Additional Dispatchable Capacity pursuant to this section, the Seller must have twenty-five (25) MWs or more capacity available as Additional Dispatchable Capacity; provided, however, that the Seller must schedule such capacity in quantities of at least 10 MW for scheduled delivery.

After the initial designation of Additional Dispatchable Capacity a Seller providing such capacity to Buyer on a firm basis shall offer such capacity no later than thirty (30) days before the annual Resource Adequacy year-ahead showing each year during the term of the Transition PPA. Such Additional Dispatchable Capacity will include a fixed heat rate, or fixed heat rate curve, established by the Seller.

If the Buyer elects not to accept Seller's offer of Additional Dispatchable Capacity for the term of the Transition PPA, then the Buyer, as the Scheduling Coordinator, will facilitate an alternative sale and delivery of the Dispatchable Capacity to the CAISO market, as long as such capacity meets the CAISO determined requirements for compliance with the CAISO Tariff and Protocols. Such alternative sale and delivery shall be consistent with the applicable CAISO Tariff and Protocols for such sales. The Seller shall have financial responsibility for any applicable CAISO charges or penalties for deviations in the delivery of the Additional Dispatchable Capacity, so long as such charges

or deviations are due to actions of the Seller. Seller shall receive CAISO revenues and charges associated with the delivery of any Additional Dispatchable Capacity into the CAISO markets.

When the IOU is not the Scheduling Coordinator and elects not to accept Seller's offer of Additional Dispatchable Capacity for the term of the Transition PPA, the IOU will facilitate the scheduling and delivery of the Transition PPA Power Product in cooperation with the Seller's Scheduling Coordinator. Seller may undertake any necessary actions with the CAISO or any other jurisdictional entity to facilitate a delivery or sale of Additional Dispatchable Capacity pursuant to this Section 3.4.1.2.

Any resulting amendment to the Transitional PPA must contain terms that allow proper operation consistent with CAISO requirements and appropriate settlement terms consistent with the risk sharing and optional product purchase described above.

- 3.4.2 Section 3.07: The amount of capacity and energy may be modified pursuant to Section 3.07 in the Transition PPA, provided that any CHP Facility modification under Section 3.07 shall not increase the GHG costs of the IOU as a result of a change in the CHP Facility pursuant to Section 3.07. The Seller shall render the Buyer indifferent to GHG cost increases resulting from Section 3.07.
- 3.4.3 CHP Facilities Operating under Transitional SO1s
- 3.4.4 QFs that previously delivered firm capacity under a Legacy PPA or other CHP PPA, and petitioned the CPUC for this firm contract option before January 1, 2010, can elect to sign a firm capacity Transition PPA and shall be paid under such PPA as if deliveries commenced on January 1, 2010. Eligible petitioners under this section are: (1) Mid-Set Cogeneration Company, QF ID #25C123E02; (2) Coalinga Cogeneration Company, QF ID #25c124E01; (3) Sargent Canyon Cogeneration Company, QF ID #18C052E01; (4) Salinas River Cogeneration Company, QF ID #18C053E01; (5) Berry Petroleum Company Taft/Kern, QF ID #25C151EO1; (6) Graphic Packing Int'l (Blue Grass) Santa Clara, QF ID #08C023EO1; (7) Berry Petroleum Company Taft/Kern, QF ID#25C099EO1; (8) Berry Petroleum, QF ID #2224 Newhall II; and, (9) US Borax, QF ID #2019 U.S. Borax and Chemical Corp.

#### 3.4.5 Curtailment

- 3.4.5.1 Seller shall promptly curtail the production of the Power Product (as defined in the PPAs attached to this Settlement) upon receipt of a notice or instruction from the CAISO, which may be communicated by Buyer if Buyer is the Scheduling Coordinator. Such notice or instruction shall only be provided when the CAISO orders curtailment and the Scheduling Coordinator implements such curtailment in compliance with the CAISO Tariff or applicable orders to avoid or address a declared System Emergency.
- 3.4.5.2 Seller shall promptly curtail the production of the Power Product upon receipt of a notice or instruction from the Transmission Provider, which may be communicated by Buyer if Buyer is the Scheduling Coordinator. Such notice

or instruction shall only be provided when curtailment of the Power Product is required to comply with:

- 3.4.5.2.1 A CAISO curtailment declared pursuant to Section 3.15 (a) or Transmission Provider declared Emergency Condition, subject to the interconnection agreement between the Seller and the Transmission Provider; or
- 3.4.5.2.2 Transmission Provider's maintenance requirements, subject to the interconnection agreement between the Seller and the Transmission Provider.
- 3.4.5.3 Notwithstanding the above, except as may be required in order to respond to any Emergency Condition or System Emergency, Buyer shall, consistent with FERC Order 888 and the interconnection agreement between the Seller and the Transmission Provider and with the applicable provisions of the CAISO Tariff:
  - 3.4.5.3.1 Use reasonable good faith efforts to coordinate Transmission Provider's curtailment needs with Seller to the extent it can influence such needs; or
  - 3.4.5.3.2 Request the Transmission Provider and CAISO limit the curtailment duration.
- 3.4.5.4 If Seller has entered into a Participating Generator Agreement (PGA) or Qualifying Facility Participating Generator Agreement (QF PGA) with the CAISO, or an interconnection agreement, the terms of the applicable PGA or QF PGA and the applicable interconnection agreement with respect to CAISO or Transmission Provider curtailments, shall govern the rights and obligations of Buyer and Seller to the extent any provision of this Section 3.4.5 is inconsistent with such applicable PGA or QF PGA, and interconnection agreement.
- 3.4.5.5 In the event Seller interconnects with an individual or entity other than the CAISO, the Seller shall adhere to any reliability curtailment order by such individual or entity pursuant to the applicable tariff provisions of such individual or entity.

# 4 CHP Procurement Processes

### 4.1 Overview

4.1.1 This section outlines various procurement methods encompassed in this CHP Program to meet the MW Targets and GHG Emissions Reduction Targets.

#### 4.2 CHP RFOs

4.2.1 Each IOU shall conduct RFOs exclusively for CHP resources (CHP RFOs) as a means of achieving its MW Target and GHG Emissions Reduction Targets, consistent with the terms of this Settlement.

### 4.2.2 Eligibility

- 4.2.2.1 Any CHP Facility with a nameplate larger than 5 MW may bid into the CHP RFO, including CHP Facilities seeking firm and as-available capacity PPAs, provided that the CHP Facility meets the definition of cogeneration under California Public Utilities Code §216.6 and the Emissions Performance Standard established by Public Utilities Code §8341 (Senate Bill 1368). A CHP Facility must meet the federal definition of a qualifying cogeneration facility under 18 CFR §292.205 implementing PURPA.
- 4.2.2.2 CHP Facilities converting to Utility Prescheduled Facilities. A CHP Facility that met the PURPA efficiency requirements (18 C.F.R. §292.205) as of September 2007 and converts to a Utility Prescheduled Facility is also eligible to participate in the CHP RFOs. After the Existing CHP Facility converts to a Utility Prescheduled Facility, it may be either a Qualifying Facility or an Exempt Wholesale Generator if the facility otherwise meets the criteria in this Section 4.2.2.2.
- 4.2.3 Term. The maximum term for PPAs resulting from the CHP RFO shall be:
  - 4.2.3.1 Up to seven (7) years for Existing CHP Facilities, both firm capacity and asavailable capacity or Expanded CHP Facilities, if they do not provide credit and collateral as set forth in Section 4.2.8. An Existing CHP Facility is one that was operational before the Settlement Effective Date.
  - 4.2.3.2 Up to twelve (12) years for New CHP Facilities, Repowered CHP Facilities, or Expanded CHP Facilities if they provide credit and collateral as set forth in Section 4.2.8. A New CHP Facility is one that becomes operational after the Settlement Effective Date and a Repowered CHP Facility is one that repowers after the Settlement Effective Date.

### 4.2.4 Pricing

- 4.2.4.1 Defined according to the executed PPA.
- 4.2.4.2 Buyers and Sellers will not assert that the prices in the PPAs executed as a result of the CHP RFOs or bilateral negotiations are inconsistent with the Buyer's avoided costs as defined by PURPA.
- 4.2.5 CHP RFO Scope, Evaluation and Selection Criteria
  - 4.2.5.1 A CHP Facility may also elect to participate in an all-source RFO or a renewable energy solicitation provided it meets the eligibility requirements for the solicitation.
  - 4.2.5.2 The CHP RFO will recognize that CHP has unique attributes.
  - 4.2.5.3 CHP offers shall be compared only to other CHP offers within the CHP RFO process.

- 4.2.5.4 The IOU shall conduct an evaluation process, including an analysis of market value, in its CHP RFO process.
- 4.2.5.5 When evaluating an offer from an Existing CHP Facility, the IOU should evaluate the energy that is being delivered to the grid from that CHP Facility.
- 4.2.5.6 CHP offers shall be evaluated on all of the CHP Program goal characteristics, including GHG emissions.
- 4.2.5.7 Each IOU shall use an Independent Evaluator (IE) similar to that used in other IOU RFO processes. It is preferable that the IE have CHP expertise and financial modeling experience.
- 4.2.5.8 The PRG advises and the IE shall review the entire CHP RFO process.
- 4.2.6 The CHP RFO Pro-Forma PPA is attached as Exhibit 5. The CHP Pro-Forma PPA may be modified on a bilateral basis during negotiations for a particular CHP PPA or Utility Prescheduled Facility PPA. As set forth in Section 4.2.12 below, the IOUs may also offer other contract options in the CHP RFO.
- 4.2.7 Modifications of the SOC for the CHP RFO Pro-Forma PPA are described in part as follows: (The terms and conditions of the attached CHP RFO Pro-Forma PPA govern)
  - 4.2.7.1 Remove all standard-offer pricing terms.
  - 4.2.7.2 GHG Compliance Cost: Seller must offer both options below:
    - 4.2.7.2.1 Seller assumes GHG Compliance Cost, and
    - 4.2.7.2.2 Seller elects to pass-through GHG Compliance Costs to Buyer.
  - 4.2.7.3 Seller and Buyer may elect a hybrid approach for GHG cost recovery. For example, Buyer covers GHG costs up to a certain Heat Rate and Seller assumes additional costs above that Heat Rate. This hybrid approach may create an efficiency incentive for the Seller to provide additional reductions of GHG emissions.
- 4.2.8 Credit and Collateral provisions for New, Repowered, or Expanded CHP Facilities
  - 4.2.8.1 Credit and collateral provisions shall apply only to PPAs for New CHP Facilities, Repowered or Expanded Facilities CHP Facilities
    - 4.2.8.1.1 An IOU may request additional offers for different credit and collateral terms from New CHP Facilities, Repowered CHP Facilities or Expanded Facilities in RFO Bid solicitations; and such options will be considered according to the terms and conditions of such options.
  - 4.2.8.2 Credit and collateral provisions for an Existing CHP Facility will not be required in any CHP PPA but may be requested by an IOU in CHP RFOs or bilateral negotiations and will be evaluated by IOUs and Sellers accordingly.

- 4.2.8.3 Performance Assurance for New or Repowered CHP Facilities shall be established equal to the value from one of the following options, and the selection of the option is at the election of the Seller:
  - 4.2.8.3.1 Twelve (12) months of capacity payments
  - 4.2.8.3.2 Twelve (12) months of revenues
  - 4.2.8.3.3 Five percent (5%) of anticipated revenues projected over the term of the PPA (e.g., 5% of revenues over the course of a 12 year PPA would be approximately seven (7) months of anticipated revenues);
  - 4.2.8.3.4 Negotiated performance assurance value and conditions for providing security for such Performance Assurance.

#### 4.2.8.4 Cross-default Provision

- 4.2.8.4.1 The Seller may provide a Letter of Credit to the Buyer. In the event of cross default event implicating the Letter of Credit, the Seller must provide a substitute Letter of Credit within five (5) to ten (10) Business Days.
- 4.2.8.4.2 Any cross-default provision in a CHP PPA shall apply specifically to the Seller and the Seller's guarantor under the CHP Pro-Forma PPA (and not a parent of the Seller, a subsidiary of the Seller or an affiliate of the Seller, unless they are the guarantor).
- 4.2.8.4.3 In addition to the default provisions contained in the attached applicable PPAs, the following two scenarios of potential default conditions are acknowledged by the Parties to be addressed by PPA credit and collateral terms and conditions:
  - 4.2.8.4.3.1 Scenario 1: If Seller breaches a CHP PPA with the Buyer, and Buyer has another PPA with Seller, then cross default could be triggered; provided that another PPA means a PPA between the specific Seller, as signatory and as referenced in Section 4.2.8.4.2, and the specific IOU Buyer, as signatory to such PPA. Another PPA is a contract between the Buyer and Seller for electric power, and not a contract for other goods or services.

Scenario 2: If the guaranter becomes uncreditworthy for the guarantee, or its credit rating is downgraded to below investment grade by either Moody's or Standard & Poor's or Fitch then default could be triggered; provided that the Seller may prevent such trigger by providing another form of credit within ten (10) Business Days.

4.2.9 Efficiency Performance Obligations and Compliance

The Efficiency Performance Obligation shall apply as incorporated into the final CHP RFO PPA, and the 60% efficiency in the Optional As-Available PPA.

Failure to meet the Efficiency requirement in the CHP PPA throughout the Term shall be, at the Buyer's election, an Event of Default under the PPA. The failure to meet the annual Efficiency requirement may be determined by the Seller's report to CARB (CARB GHG Emissions Reduction Report), the Seller's report to the IOUs pursuant to the CHP QF Compliance Monitoring Program, or other information available to the IOU, such as the loss of a thermal host. If the Seller is out of compliance with the Efficiency Performance Obligation, Buyer shall issue to Seller a written notice of an Efficiency Performance Deficiency. Within three (3) months of its receipt of the notice, Seller will provide Buyer with a plan to cure the Efficiency Performance Deficiency. Buyer shall accept or reject the plan within 30 days of receipt. From the date of written notice from the Buyer of acceptance of the Seller's cure plan, Seller shall have six (6) months to execute the plan and cure the Efficiency Performance Deficiency. If Buyer rejects the cure plan, then Buyer and Seller shall confer to resolve the issues. A Seller meeting these cure provisions shall not incur a default under the PPA.

If a Seller fails to provide an adequate showing that it cured the Efficiency Performance Deficiency within the six (6) month cure period, and does not secure a waiver from the IOU, that Seller is subject to the otherwise applicable default provisions in the PPA.

Seller may have up to two cure periods during the term of the applicable PPA for no more than two Efficiency Performance Deficiencies. A second cure period is available to Seller only if Seller is able to cure the first Efficiency Performance Deficiency. A third Efficiency Performance Deficiency is, at the election of Buyer, an Event of Default pursuant to section 6.2 of the applicable PPA and there is no applicable cure period.

#### 4.2.10 Curtailment: Reliability

- 4.2.10.1 Seller shall promptly curtail the production of the Power Product (as defined in the PPAs attached to this Settlement) upon receipt of a notice or instruction from the CAISO, which may be communicated by Buyer if Buyer is the Scheduling Coordinator. Such notice or instruction shall only be provided when the CAISO orders curtailment and the Scheduling Coordinator implements such curtailment in compliance with CAISO Tariff or applicable orders to avoid or address a declared System Emergency.
- 4.2.10.2 Seller shall promptly curtail the production of the Power Product upon receipt of a notice or instruction from the Transmission Provider, which may be communicated by Buyer if Buyer is the Scheduling Coordinator. Such notice or instruction shall only be provided when curtailment of the Power Product is required to comply with:
  - 4.2.10.2.1 A CAISO curtailment declared pursuant to Section 3.15 (a) or
    Transmission Provider declared Emergency Condition, subject to
    the interconnection agreement between the Seller and the
    Transmission Provider; or

- 4.2.10.2.2 Transmission Provider's maintenance requirements, subject to the interconnection agreement between the Seller and the Transmission Provider.
- 4.2.10.3 Notwithstanding the above, except as may be required in order to respond to any Emergency Condition or System Emergency, Buyer shall, consistent with FERC Order 888 and the interconnection agreement between the Seller and the Transmission Provider and with the applicable provisions of the CAISO Tariff:
  - 4.2.10.3.1 Use reasonable good faith efforts to coordinate Transmission Provider's curtailment needs with Seller to the extent it can influence such needs: or
  - 4.2.10.3.2 Request the Transmission Provider and CAISO limit the curtailment duration.
- 4.2.10.4 If Seller has entered into a PGA or QF PGA with the CAISO, or an interconnection agreement, the terms of the applicable PGA or QF PGA and the applicable interconnection agreement with respect to CAISO or Transmission Provider curtailments, shall govern the rights and obligations of Buyer and Seller to the extent any provision of this Section 4.2.10 is inconsistent with such applicable PGA or QF PGA, and interconnection agreement.
- 4.2.10.5 In the event Seller interconnects with an individual or entity other than the CAISO, the Seller shall adhere to any reliability curtailment order by such individual or entity pursuant to the applicable tariff provisions of such individual or entity.

### 4.2.11 Curtailment; Economic

- 4.2.11.1 The CHP Pro-Forma RFO PPA will contain an option, specifically, an Economic Curtailment Option (ECO), that may be selected by the CHP RFO participant and the participant may bid zero under this option.
- 4.2.11.2 The Buyer ean only instruct the Seller to curtail production in those hours when the CAISO published Day-Ahead Integrated Forward Market (IFM) results indicate there is a negative EZ-Gen Hub Locational Marginal Price (LMP)<sup>2</sup> or negative System Marginal Energy Cost (SMEC) as defined and set forth in the CAISO Tariff.
- 4.2.11.3 The ECO offered by the CHP RFO participant must contain the fixed MWhs shown in the table entitled Curtailment Period Cap below, for each calendar quarter and on-peak and off-peak time period. The differences between the Seller's Day-Ahead Forecast ("i.e., the schedule Seller provides to Buyer by 5:00 PPT on the day before the Trading Day") and the Economic Curtailment Limit (ECL) schedules submitted over the applicable period must be equal to the Curtailment Period Cap shown in the CHP RFO offer.

NP15, SP15 or ZP26, based on the location of the CHP Facility.

# Curtailment Period Cap

Calendar	On-Peak (MWhs) <sup>3</sup>	Off-Peak (MWhs)4	Total (MWhs)
Quarter			,
Jan – Mar	TBD by Seller	TBD by Seller	TBD by Seller
Apr – June	TBD by Seller	TBD by Seller	TBD by Seller
July -Sept	TBD by Seller	TBD by Seller	TBD by Seller
Oct – Dec	TBD by Seller	TBD by Seller	TBD by Seller
Total	TBD by Seller	TBD by Seller	TBD by Seller

Once a Curtailment Period Cap is reached for any quarterly period (either onpeak or off-peak), no additional economic curtailment is available to the Buyer in that quarterly on-peak or off-peak period. Any MWhs not called by the Buyer in any period cannot be rolled over to another period.

- 4.2.11.4 Those Sellers electing the ECO will provide two Day-Ahead power delivery schedules to the Buyer: a Seller's Day-Ahead Forecast in accordance with the existing terms of the CHP Pro-Forma RFO PPA, and an ECL Schedule indicating the amount of MWhs that Seller offers to be curtailed in each hour for the next day.
- 4.2.11.5 By 2 pm or one hour after receiving published IFM results, whichever is later, Buyer may instruct the Seller to curtail deliveries down to the ECL Schedule, if the IFM indicates a negative EZ-Gen Hub LMP or negative SMEC for such hour. Within one hour of the Buyer's economic curtailment order, Seller shall provide a binding notice of Seller's intention to either implement the Buyer's economic curtailment order (curtailment order) or maintain its Seller's Day-Ahead Forecast for the negative EZ-Gen Hub LMP or negative SMEC hour(s). If the Seller chooses to maintain its Seller's Day-Ahead Forecast, it will contemporaneously notify the Buyer of its pricing election pursuant to Section 4.2.11.7.
- 4.2.11.6 If Buyer orders an economic curtailment for a schedule hour and Seller curtails to its ECL Schedule, then the energy payment pursuant to the RFO PPA will be made by Buyer to Seller for Seller's Day-Ahead Forecast.
- 4.2.11.7 If Buyer orders an economic curtailment for a schedule hour and Seller elects not to curtail to its ECL Schedule, then energy payments for that schedule hour will be made pursuant to the Seller's election as provided under Section 4.2.11.5 for all deliveries above the ECL Schedule at:
  - 4.2.11.7.1 IFM Pricing: Seller will pay the Buyer either the negative SMEC or the negative EZ-Gen Hub LMP, provided if both prices are

On-Peak TOU period includes PG&E's Peak and Partial Peak, SCE's On-Peak and Mid-Peak, and SDG&E's On-Peak and Semi-Peak.

Off Peak TOU period includes Off-Peak and Super Off-Peak for all three IOUs.

- negative Seller will pay the average of the two negative prices for that scheduled hour, or
- 4.2.11.7.2 Real Time Collar Pricing: Seller will be paid for energy deliveries above the ECL Schedule at the Real Time (RT) SMEC subject to the designated floor of negative \$50.00 per MWh and ceiling value of the PPA energy price (collar values). The collar values would be, incorporated into the RFO PPA and could not be changed over the term of the PPA. If the hour has a negative RT SMEC, the negative SMEC would be paid by the Seller to the Buyer down to the floor value. If the RT SMEC is positive, the Buyer would pay the Seller the RT SMEC for that hour up to the ceiling value.
- 4.2.11.8 If, at the end of a calendar quarter in the table above, Seller has not offered the Curtailment Period Cap, then Seller will pay Buyer the shortfall at \$50 per MWh for the applicable period.
- 4.2.11.9 For those hours where the Buyer has ordered an economic curtailment, the difference between the Seller's Day-Ahead Forecast and the ECL Schedule will count toward the Curtailment Period Cap irrespective of whether or not the Seller chooses to curtail its operation.
- 4.2.11.10 If Seller curtails, any economic curtailment hour will not be included in the penalty assessment for failure to meet Seller's scheduled amounts and all CAISO charges and penalties, if any, for schedule deviations associated with curtailment will be borne by the Buyer.
- 4.2.11.11 For capacity payment calculation purposes, the Seller's Day-Ahead Forecast will be used for any hour affected by the curtailment so as to not impact the capacity payment. This is necessary due to the physical limitations of the CHP Facilities, such as ramp rates, to respond to the curtailment order.

#### **EXAMPLES**

For a given hour, assume
Contract energy Price = \$30/MWh
DA EZ-Gen Hub LMP = -\$9
DA SMEC = -\$11
Seller's Day-Ahead Forecast Energy = 100 MWs
Curtailment Period Cap = 1,000 MWhs
RT SMEC Collar = + \$30/MWh/- \$50/MWh

<u>Case 1 (4.2.11.6)</u>. Assume Seller's ECL Schedule is 90 MWs and Seller agrees to curtail deliveries in response to the order. Seller delivers 90 MWs but is paid for all 100 MWs of the Seller's Day-Ahead Forecast at \$30/MWh. Seller gets credit for 10 MWhs toward fulfilling the Curtailment Period Cap.

<u>Case 2 (4.2.11.6)</u>. Assume Seller's ECL Schedule is 80 MWs and Seller agrees to curtail deliveries in response to the order. Seller delivers 80 MWs but is paid for all 100 MWs of the Seller's Day-Ahead Forecast at \$30/MWh. Seller gets credit for 20 MWhs toward fulfilling the Curtailment Period Cap.

Case 3 (4.2.11.7.1). Seller notifies Buyer that it will deliver the Seller's Day-Ahead Forecast quantity (100 MWs) and accept IFM pricing (Average = -\$10/MWh) for deliveries in excess of the ECL Schedule (90 MWs). Seller is paid the contract price of \$30/MWh for 90 MWs and owes Buyer \$10 per MWh for the 10 MWs of energy that was not curtailed. Seller receives credit for 10 MWhs toward fulfilling the Curtailment Period Cap.

<u>Case 4 (4.2.11.7.2)</u>. Seller notifies Buyer that it will deliver the Seller's Day-Ahead Forecast quantity (100 MWs) and accept the RT collar pricing for deliveries in excess of the ECL Schedule (90 MWs). Seller will be paid for 90 MWs at the contract price of \$30/MWh.

- a. If the RT SMEC for that hour is \$-20/MWh, Seller will pay Buyer for the 10 MWs at a price of \$20/MWh.
- b. If the RT SMEC for that hour is -\$75/MWh, Seller will pay Buyer for the 10 MWs at a price of \$50/MWh as limited by the price exposure collar.
- c. If the RT SMEC for that hour is \$250/MWh, Seller will be paid for the 10 MWs at a price of \$30/MWh as limited by the price exposure collar.

In each case (4a, 4b and 4c) Seller receives credit for 10 MWhs toward fulfilling the Curtailment Period Cap.

## 4.2.12 PPA Options in CHP RFOs

As part of the bid package for each CHP-Only RFO, each IOU may request offers with specific (1) credit and collateral, (2) voluntary curtailment, and (3) dispatchability terms that differ from the CHP RFO Pro Forma PPA. As part of the bid package, IOUs may also offer the all source RFO in addition to the CHP-Only RFO and may also sign a hybrid contract of the two.

In IOU evaluations of final offers from CHP bidders, the IOUs will give preference to Pro Forma offers with no options, relative to non-Pro Forma offers, to the extent that such Pro Forma offers are competitive with the non-Pro Forma offers.

For example, assume there are 10 RFO bids, 6 Pro Forma bidders and 4 non-Pro Forma bidders. If the 5 of the 6 Pro Forma bidders (those offering the product contemplated by the program) are competitive (based upon the standards in the Settlement and are otherwise commensurate with the product solicited) the IOU would choose those 5 bids and cannot choose the 4 non-Pro Forma bidders in lieu of Pro-Forma bidders; this does not preclude the IOU from selecting the 4 non-Pro Forma bidders. Non-Pro Forma bids cannot replace or substitute for the selection of Pro Forma bids.

## 4.3 Bilaterally Negotiated PPAs

4.3.1 Bilaterally negotiated and executed CHP PPAs or Utility Prescheduled Facilities PPAs are part of the procurement options in this CHP Program.

- 4.3.2 Use of an IE shall be required for any negotiations between an IOU and its affiliate and may be used, at the election of either the Buyer or the Seller, in other negotiations.
- 4.3.3 Pricing, terms, and conditions will be according to the executed and approved PPA.

#### 4.4 AB 1613 Feed-In Tariff

- 4.4.1 Feed-In Tariff for equal to or less than 20 MW (nameplate) CHP Facilities that are New CHP Facilities/Repowered CHP Facilities (Assembly Bill 1613 (codified in California Public Utilities Code §2840 et seq., which establishes the Waste Heat and Carbon Emissions Reductions Act)(R.08-06-024)) are part of the procurement options in this CHP Program.
- 4.5 PURPA Program for QFs 20 MW or Less (QF PPAs)
  - 4.5.1 The CPUC will continue to administer a QF Program under PURPA for QFs 20 MW or less. QFs must continue to meet the QF requirements established by PURPA and any FERC rules as amended from time to time (18C.F.R. Part 292, §292.203 et seq.) implementing PURPA.
  - 4.5.2 QFs that are equal to or less than 20 MW nameplate will be provided a QF PPA containing standard terms and conditions for eligible Qualifying Facilities.
  - 4.5.3 Eligible QFs may continue to operate under the PURPA must-take obligation program. Any CHP Facility meeting the CHP RFO eligibility requirements may also submit an offer in the CHP RFO or seek alternative contracting options; however, the lOUs must continue to offer QF PPAs to these QF CHP Facilities.
  - 4.5.4 Pricing
    - 4.5.4.1 Energy prices will be SRAC as set forth in Section 10.
    - 4.5.4.2 Capacity prices will be as specified in D.07-09-040.
  - 4.5.5 The QF PPA for QFs equal to or less than 20 MW is based on the SOC and is attached as Exhibit 6.
- 4.6 As-Available Procurement Alternatives and Optional As-Available PPA
  - As-Available CHP Facilities are eligible for several procurement alternatives under the CHP Program, including the CHP RFOs, bilaterally negotiated PPAs, the AB 1613 Feed-In Tariff and the PURPA Program for QFs 20 MWs or under. In addition to these alternatives for the procurement from as-available CHP Facilities, there is also an Optional As-Available PPA.
  - 4.6.2 Optional As-Available PPA is subject to the following terms and conditions:
    - 4.6.2.1 Eligibility
      - 4.6.2.1.1 Gas-fired CHP Facilities with nameplates greater than 20 MW, but average annual deliveries less than 131,400 MWh (as specified in D.07-09-040 and D.08-09-024).

- 4.6.2.1.2 The as-available project host(s) must consume, consistent with Public Utilities Code §218(b), at least 75% of the total electricity generated by a Topping Cycle CHP Facility or at least 25% of the total electricity generated by a Bottoming Cycle CHP Facility. For purposes of this section, "CHP Facility" includes all CHP units serving the host's thermal or electric loads consistent with Public Utilities Code §218(b).
- 4.6.2.1.3 For Topping Cycle or Bottoming Cycle with supplemental firing, the as-available CHP Facility must meet sixty percent (60%) efficiency calculated by dividing the total annual useful thermal and electrical output by the total annual fuel use, based on Higher Heating Value. There will be no efficiency requirement for a Bottoming Cycle CHP Facility with no supplemental firing.
- 4.6.2.2 Capacity Pricing: The capacity price shall be set consistent with the asavailable capacity price in D.07-09-040, subject to escalation as provided in that decision, and shall be applied up to a maximum of 20 MW of deliveries measured on an Integrated Hour. For purposes of this section, an Integrated Hour is the sum of all measured meter intervals for the applicable hour.
- 4.6.2.3 Energy Pricing: Energy scheduled on a day-ahead basis and delivered up to 20 MW per hour in a given hour based on the lesser of Day-Ahead scheduled energy or metered deliveries will he priced at SRAC.
  - 4.6.2.3.1 Energy scheduled on a Day-Ahead basis and delivered above 20 MW per hour in a given hour will be priced at the MRTU Day-Ahead Market PNode energy price.
  - 4.6.2.3.2. Seller shall schedule all deliveries with the IOU on a Day-Ahead basis in advance of timing required for Buyer to schedule energy into the CAISO Day-Ahead market (8 hours in the CAISO day-ahead market).
  - 4.6.2.3.3 Unscheduled energy incremental to scheduled energy in a given hour shall be priced at the MRTU real time PNode price for such energy, thus Seller shall pay any applicable CAISO Charges and receive all CAISO Revenues attributable to unscheduled deviations between Seller's scheduled and metered deliveries for such incremental energy. Applicable CAISO Charges for deviations shall be the responsibility of the Seller.
  - 4.6.2.3.4 The Performance Tolerance Band under Scheduling and Delivery Deviation (SDD) Energy Adjustment and the SC Trade Tolerance Band shall be set at the greater of (a) three (3) percent of the Seller's Final Energy Forecast divided by the number of Settlement Intervals in such hour or (b) three (3) MWh divided by the number of Settlement Intervals in such hour.
- 4.6.2.4 The term of an Optional As-Available PPA shall be up to seven (7) years at the discretion of the Seller. If the Seller chooses a PPA term of five (5) years or

- greater, the Seller will provide on a confidential basis to the IOU sufficient information for the IOU to confirm that the CHP Facilities comply with the Emissions Performance Standard, if such standard is applicable to the CHP as an as-available facility.
- 4.6.2.5 The SOC is the basis for the development of the Optional As-Available PPA, and is attached here as Exhibit 7.
- 4.6.2.6 (Intentionally omitted)
- 4.6.2.7 Seller shall comply with the applicable provisions of the CAISO Tariff as determined by the CAISO.
- 4.6.2.8 Buyer, at Seller's election, shall be Seller's Scheduling Coordinator. At Seller's option, Seller may establish and pay for a Scheduling Coordinator (SC) ID for the CHP Facility.
- 4.6.2.9 GHG emissions reductions associated with New, Repowered, or Expanded CHP under an Optional As-Available PPA shall be the amount of GHG emissions reductions from the entire CHP Facility as, as set forth in Section 7.
- 4.6.2.10 Average MW (AMW) Energy Delivery Cap for the Optional As-Available PPAs
  - 4.6.2.10.1 There is no specific MW capacity target for the Optional As-Available Program.
  - 4.6.2.10.2 There is a cap on energy deliveries from all of the Optional As-Available PPAs measured each year in AMW for each IOU (AMW Cap) until the AMW Cap is reached. The AMW Cap shall be: PG&E: 75 AMW, SCE: 75 AMW, SDG&E: 10 AMW. An Optional As-Available PPA will remain available to eligible CHP Facilities in an IOU Service Territory until the IOU's AMW Cap has been reached. The IOU shall report annually on the determination of reaching the AMW Cap until the cap is reached. The IOU may choose to offer a PPA to a CHP Facility although the IOU has exceeded its AMW Cap, in its sole discretion.
  - 4.6.2.10.3 To determine progress toward an IOU's AMW Cap, AMW will be calculated by dividing MWh of deliveries under an Optional As-Available PPA by the total hours in the year.
- 4.6.2.11 Counting Rules for Optional As-Available PPA Enrollment and MW Targets
  - 4.6.2.11.1 Each gas-fired CHP Facility eligible pursuant to Section 4.6.2.1 (including bottoming-cycle CHP Facilities) with an existing CHP PPA or which had a CHP PPA within twenty-four (24) months before the CHP Facility provides notice of its intention to execute an Optional As Available PPA may upon expiration or termination of its CHP PPA execute an Optional As-Available PPA provided the IOU for the Service Territory where the CHP Facility is located

has not reached its AMW Cap. AMW deliveries from these CHP Facilities may be counted toward the AMW Cap using deliveries as reported in the IOU's most recent public information such as FERC Form 1 or CEC Annual Power Disclosure Report and towards the IOU MW Target. If the CHP Facility re-contracts with an IOU upon expiration or termination of its PPA, the procuring IOU may count the full project capacity toward its CHP MW Target. If there are multiple IOUs procuring power from the same CHP Facility, then the procured contract capacity for each IOU shall be counted respectively;

- 4.6.2.11.2 An as-available Seller eligible pursuant to Section 4.6.2.1 (including Bottoming Cycle CHP Facilities) that has not had a PPA with an IOU at any time, or whose CHP PPA terminated more than twenty-four (24) months before the CHP Facility provides notice to secure an Optional As-Available PPA, may execute an Optional As-Available PPA provided the IOU in the Service Territory where the CHP Facility is located has not reached its AMW Cap. AMW of deliveries from these CHP Facilities may be counted toward the AMW Cap using, if available, deliveries as reported in publicly available reports such as FERC Electric Quarterly Reports or the most recent FERC Form 1. If such reports are not available, the counting for the AMW Cap for such CHP Facilities shall rely upon either the Seller's forecast of AMW delivery from the CHP Facility, or actual deliveries to the IOU, if available; and thereafter from available public information to establish AMW deliveries, e.g., FERC Form 1 data. AMW deliveries from these CHP Facilities may be counted toward the MW Targets. If there are multiple IOUs procuring power from the same CHP Facility then the AMW deliveries for each IOU shall be counted respectively;
  - 4.6.2.11.2.1 AMW of deliveries from a New CHP Facility or Repowered CHP Facility eligible for an Optional As-Available PPA will count toward an IOU's AMW Cap. The capacity of such a CHP Facility may be counted toward the MW Targets. Capacity for a New CHP Facility or Repowered CHP Facility shall be established using a one-time Capacity Demonstration Test, net of auxiliary or station power.
  - 4.6.2.11.2.2 A CHP Facility currently operating under an evergreen Legacy PPA may not terminate its evergreen Legacy PPA to obtain a new Optional As-Available PPA. Neither the MW nor the AMW of deliveries under these CHP PPAs may be counted toward the MW Targets or the AMW Cap. If, however, the CHP Facility adds new capacity, it may receive an Optional As-Available PPA for any deliveries that are determined to be associated with the new capacity, and the MW of new capacity shall be counted toward the MW Targets.

#### 4.7 IOU-Owned CHP

4.7.1 IOU-owned CHP counts towards the IOUs' GHG Emissions Reduction Targets for the Second Program Period, but not for the 3,000 MW Target. The counting from these resources is capped at 10% of the IOUs' GHG Emissions Reduction Target.

## 4.8 Utility Prescheduled Facilities

## 4.8.1 Eligibility

- 4.8.1.1 A CHP Facility that met the PURPA efficiency requirements as of September 20, 2007 and converts to a Utility Prescheduled Facility is eligible to participate in a CHP RFO or to obtain a PPA through bilateral negotiations or amend an existing Legacy PPA through bilateral negotiations.
- 4.8.1.2 New PPAs with Utility Prescheduled Facilities (not Legacy PPA Amendments) count towards the MW Targets if the existing QF PPA expires before the end of the Transition Period. For SDG&E, any procurement of a Utility Prescheduled Facility (regardless of the date of expiration of the CHP PPA) counts toward SDG&E's MW Target.
- 4.8.1.3 Amendments to Legacy PPAs to convert to a Utility Prescheduled Facility count towards each IOU's GHG Emissions Reduction Targets.

#### 4.9 New Behind the Meter CHP Facilities

4.9.1 New CHP Facilities that are behind the meter including the Self-Generation Incentive Program (SGIP) and New CHP Facilities that do not export to the grid are part of the procurement options in this CHP Program.

## 4.10 Approval of PPAs

- 4.10.1 IOUs will utilize a Tier 2 Advice Letter for Existing CHP Facilities that execute the CHP RFO Pro Forma PPA without material modification.
- 4.10.2 lOUs will utilize a Tier 3 Advice Letter for all other PPAs (new, repowering or existing PPAs that contain any material modification of the PPAs approved in this Settlement).
- 4.10.3 PPAs of less than five (5) years do not require advance CPUC approval according to existing CPUC policy. Such CHP PPAs should be included in the IOU Quarterly Compliance Reports (QCRs) and CHP Program Semi-Annual Report specified in Section 8.1.1, below.
- 4.10.4 Emissions Performance Standard (EPS) Approval for PPAs
  - 4.10.4.1 For those PPAs equal to or greater than five (5) years in length that are submitted to the CPUC in a Tier 2 or Tier 3 Advice Letter, the CPUC must make a specific finding that the PPA with the Generating Facility is in compliance with the Emissions Performance Standard.

- 4.10.4.2 The Sellers must provide sufficient information, on a confidential basis as necessary, to the lOUs to allow the lOUs to make a sufficient showing of compliance with the EPS for all PPAs with terms five (5) years or greater.
- 4.10.4.3 For those PPAs equal to or greater than five (5) years in length that are not submitted to the CPUC in a Tier 2 or Tier 3 Advice Letter, the Buyer shall use reasonable efforts to obtain a letter from the CPUC Energy Division acknowledging that the Generating Facility is in compliance with the EPS (EPS Compliance Letter). The Buyer will not execute the applicable PPA until the CPUC Energy Division provides the EPS Compliance Letter.

### 4.11 Seller Reporting Requirements for PPAs

Subject to CPUC confidentiality rules applicable to the information being reported, on an annual basis, Seller will provide a report to the IOU showing the annual used thermal output, total electricity generation, and fuel consumed. Such report shall be provided within five (5) Business Days after the CARB Mandatory GHG Emissions Annual Reports are due to CARB or a successor regulatory agency.

CHP QFs will continue annual reporting to the lOUs pursuant to the CPUC-approved CHP QF Monitoring Program for CHP QFs. If the CHP Facility is not a QF, it will provide the same data to the lOUs and CPUC as is contained in the CHP QF Compliance Monitoring Program report, and on the same schedule.

# 4.12 Timing of Commencement of Delivery

An IOU may record in the CHP Program Reports as progress towards obtaining its MW Targets and GHG Emissions Reduction Targets the MWs, GHG Credits or GHG Debits associated with the PPA at the time of execution. If the CPUC subsequently disapproves such PPA, or if an executed PPA terminates before beginning deliveries, the IOU shall remove any MW, GHG Credits or GHG Debits from such disapproved or terminated PPA from its CHP Program Reports.

Subject to force majeure conditions, Sellers under PPAs for Existing CHP Facilities will be required to commence deliveries on the earlier of the following two dates: (a) two (2) years after PPA execution; or (b) the Term Start Date specified in the PPA. Subject to force majeure conditions, Sellers under PPAs for New CHP Facilities or Repowered CHP Facilities will be required to commence deliveries on the earlier of the following two events: (a) five (5) years after CPUC approval of the PPA, or (b) the Term Start Date in the PPA. Subject to force majeure conditions, Sellers under PPAs for Expanded CHP Facilities will be required to commence deliveries on the earlier of the following two events: (a) three (3) years after CPUC approval of the PPA, or (b) the Term State Date in the PPA. The IOU's procedures and administration of the PPAs, including milestone provisions should target these timeframes for the start of the term. For good cause the Buyer and Seller may agree to delay the commencement of the term.

# 5 MW Targets

## 5.1 IOUs' MW Targets

5.1.1 The IOUs' combined CHP MW Targets will be 3,000 MW of capacity. The MW Targets may be met through any of the CHP Procurement Processes described in Section 4.

5.1.2 The allocation of the 3,000 MW for the Initial Program Period shall be as follows:

Utility	MW Target A	MW Target B	MW Target C	IOU Total MW Target
SCE	630 MW	378 MW	394 MW	1,402 MW
PG&E	630 MW	376 MW	381 MW	1,387 MW
SDG&E	60 MW	50 MW	50 MW	160 MW
Total Targets	1320 MW	804 MW	825 MW	2,949 MW*

<sup>\*</sup>To meet the total of 3,000 MW, SDG&E will have an additional 51 MW Target by 2018 in the Second Program Period.

5.1.3 Each IOU shall enter into PPAs to meet the MW Targets and GHG Emissions Reduction Targets consistent with the CHP Procurement Processes listed in Section 4.

#### 5.1.4 CHP RFOs

- 5.1.4.1 Each IOU shall conduct three RFOs during the Initial Program Period to seek PPAs for the portion of the MW Targets set forth in Section 5.1.2., not otherwise procured by other procurement processes described in Section 4 (Net MW Target(s)).
- 5.1.4.2 The CHP RFOs during the Initial Program Period shall be scheduled at regular intervals, provided the first CHP RFO is initiated within ninety (90) days of the Settlement Effective Date.
- 5.1.4.3 The amount of CHP sought in each CHP RFO during the Initial Program Period shall not be less than the Net MW Target for each IOU. An IOU may procure MWs in excess of the MW Targets in Section 5.1.2 above relying upon an RFO or any other CHP Procurement Processes listed in Section 4.
- 5.1.4.4 If an IOU does not meet its MW Targets identified in Section 5.1.2 at the end of any particular RFO, the portion of the unmet MW Target shall be rolled over into the subsequent MW Target for procurement.
- 5.1.4.5 Any MW shortfall that occurs in the Initial Program Period shall be rolled over into the Second Program Period to reach the 3,000 MW Target; however, such shortfall may also be addressed by other actions deemed appropriate by the CPUC.

- 5.1.4.6 The number of CHP RFOs in the Second Program Period to meet MW Targets, if any, shall be established in the LTPP proceedings. Except for SDG&E there are no established MW Targets in the Second Program Period. There are GHG Emissions Reduction Targets in the Second Program Period as defined in Section 6.
- 5.1.4.7 The number of CHP RFOs in the Second Program Period shall be established in the LTPP proceedings.
- 5.1.4.8 The MWs required in the Second Program Period may be limited or expanded, as determined by the CPUC in the LTPP process, but shall be no less than the MW Targets for the Second Program Period specified in Section 5.1.2 above.

### 5.2 MW Counting Rules

- 5.2.1 The timing for procuring the MW Targets shall be as provided in Section 5.1.2.
- 5.2.2 PPAs executed during the period between September 1, 2009 and the Settlement Effective Date count towards the lOUs' MW Targets and GHG Emissions Reduction Targets.
- 5.2.3 Progress towards the MW Targets for Existing CHP Facilities shall be determined as follows:
  - 5.2.3.1 For the purposes of Section 5.2 regarding MW counting, Existing CHP Facilities are gas-fired Topping Cycle CHP Facilities that exported and delivered electric power to an IOU listed by QF ID number in each IOU's July 2010 Cogeneration and Small Power Production Report (July 2010 Semi-Annual Report) "Contract Nameplate," as amended, if necessary. The MWs counted for New PPAs executed with Existing CHP Facilities will be the published Contract Nameplate value, unless otherwise noted in this Settlement.
  - 5.2.3.2 CHP PPAs executed with QFs who formerly sold to the lOUs and are not listed in the July 2010 Semi-Annual Reports will count towards the MW Targets based on the Contract Nameplate (or for PG&E the "operating size") in the most recent Cogeneration and Small Power Production Semi-Annual Report, if any, in which the CHP was listed. If not listed as contract nameplate, the contract nameplate in the most recent QF or CHP PPA will be counted.
- 5.2.4 Exceptions Related to Counting for Existing CHP Facilities
  - 5.2.4.1 To the extent the full net output is offered by an Existing CHP Facility but is not procured by the IOU, only the capacity procured by an IOU shall count towards any IOU's MW Target.
    - 5.2.4.1.1 For example if a 50 MW capacity CHP Facility from the July 2010 Scmi-Annual Report offers 40 MW for procurement, and the IOU procures less than 40 MW, only the capacity procured will count towards the IOU's MW Target. If the CHP Facility sells the full 40 MW offered, then the IOU may count the full 50 MW of capacity reported in the July 2010 Semi-Annual Report.

- 5.2.4.2 Coal-fired, wood waste, and renewable QFs in an IOU's portfolio as of July 2010 will not count towards the IOU's MW Targets.
- 5.2.4.3 Any Existing CHP Facility that does not export power to the grid will not count toward the IOU's MW Target, including existing SGIP facilities.
- 5.2.4.4 If an IOU contracts with an Existing CHP Facility that is currently operating in the State but has never sold to an IOU, the capacity from such CHP Facility will be determined by a Capacity Demonstration Test. This test will determine the CHP Facility's capability to serve on site and PUC §218(b) loads and provide net exports to the grid, excluding auxiliary/station power load.
  - 5.2.4.4.1 If a CHP Facility elects and is eligible under the Optional As-Available Program, the specific counting rules of that program shall apply for the purpose of ealeulating the subscription cap in the Optional As Available Program.
- 5.2.5 A New CHP Facility for the purpose of Section 5.2 means gas-fired Topping Cycle CHP Facilities and Bottoming Cycle CHP Facilities using waste heat. The capacity of a New CHP Facility to be used to count progress towards the MW Targets shall be established by a Capacity Demonstration Test. The CHP Facility's capacity, as demonstrated by this test, shall exclude auxiliary/station power load.
  - 5.2.5.1 The MWs of capacity for PPAs executed pursuant to the AB 1613 Feed-In Tariff and SGIP shall be the CHP Facility nameplate reported to the lOUs by the Seller, applicant or system owner applicable to such facility.
  - 5.2.5.2 The MWs of capacity for behind the meter CHP Facilities other than SGIP will be the capacity reported to the CEC by the Generating Facility owner and other power plant owners.
- 5.2.6 A Repowered CHP Facility for the purpose of Section 5.2 means gas-fired Topping Cycle CHP Facilities and Bottoming Cycle CHP Facilities using waste heat. The MW to be used to count progress towards the MW Targets shall be determined by a Capacity Demonstration Test after the CHP Facility re-power is completed. The Capacity Demonstration Test will determine the increment of additional project capacity, which will count towards the procuring IOU's MW Target. The total MWs to count progress will be the additional project capacity combined with the Contract Nameplate in each IOU's July 2010 Semi-Annual Report. In the event the Capacity Demonstration Test results in a value lower than the value determined as set forth in the July 2010 Semi-Annual Report, the July 2010 Semi-Annual Report value will count toward the IOU's MW Target.
- 5.3 CHP Program MW Counting Precedent or Use
  - 5.3.1 The Parties acknowledge that the MW counting pursuant to this Section 5, or the specific, individual project capacity counting levels for the CHP Program shall not establish a precedent or be used for any other purpose, or with any other agency for establishing available capacity from any individual CHP Facility.
- 5.4 Justification for Failure to Meet MW Targets

Any IOU that is unable to meet its MW Target must make a showing to justify its inability to meet the MW Target. Lack of sufficient offers can be used as a reason to justify failure to procure the MW Targets and GHG Emissions Reduction Targets. The efficiency of the CHP Facility participating in the IOUs' procurement programs as compared to the Double Benchmark, offer prices in excess of levels as provided herein, and the amount of GHG emissions reductions may be valid justifications for missing the IOU MW Targets and GHG Emissions Reduction Targets. Lack of need or portfolio fit arguments shall not be used as reasons to justify failure to procure the MW Targets, but are reasons to justify an inability to meet the GHG Emissions Reduction Targets.

5.4.1 Offer prices: If the IOU claims that CHP RFO offer prices are excessive, the IOU must refer to independent or publicly-available sources. For example, when making a justification on the basis of price of new or repowered fossil fuel-fired generation, the IOU may compare offer prices to prices reflected in sources such as Cambridge Energy Research Associates' capital cost index, CPUC estimates of new fossil fuel-fired generation, CEC estimates of new fossil fuel-fired generation, forward market prices, or other similar third-party information regarding the cost of new generation CHP Facilities in California.

# 6 GHG Emissions Reduction Targets

- 6.1 Strategy
  - 6.1.1 By approving this Settlement, the CPUC will adopt a strategy to reduce statewide GHG emissions by the following means:
    - 6.1.1.1 Maintaining the existing GHG emissions reduction attributable to the efficient Existing CHP Facilities and reducing GHG emissions from the inefficient Existing CHP Facilities, as defined in 7.1.3, by encouraging the repower, eonversion to Utility Prescheduled Facilities or retirement of such CHP Facilities; and,
    - 6.1.1.2 Increasing the efficiency of the CHP fleet by adding efficient CHP resources to the IOUs' electric portfolios to make progress towards the CARB CHP RRM;
    - 6.1.1.3 Allocating a portion of the CARB CHP RRM to other non-IOU load serving entities to increase the amount of efficient CHP in the State; and,
    - 6.1.1.4 Achieving the GHG Emissions Reduction Targets by December 31, 2020.
- 6.2 IOUs' GHG Emissions Reduction Targets
  - 6.2.1 Existing: Maintaining GHG Emission Reductions from Existing CHP
    - 6.2.1.1 The IOUs shall maintain an equivalent amount of GHG emissions reductions attributable to the gas-fired Topping Cycle CHP Facilities included in each IOU's July 2010 Semi-Annual Reports for PPAs that expire in the Initial Program Period.
  - 6.2.2 New GHG Reductions: 10U GHG Emissions Reduction Targets

- 6.2.2.1 In addition to 6.2.1.1., this Settlement establishes a GHG target of 4.3 MMT based on the CARB Scoping Plan estimates that, by 2020, the State can add 4,000 MW of additional efficient CHP. These 4,000 MW are estimated to reduce GHG emissions by 6.7 MMT. The CARB CHP RRM does not have specific allocations to the IOUs.
- 6.2.2.2 The Parties agree that the portion of the CARB CHP RRM allocated to the lOUs as part of this CHP Program can be derived by multiplying the current, or any revised statewide CARB CHP RRM, by the lOUs' percentage shares of statewide retail electric sales, as determined by the most current CEC data.
- 6.2.2.3 The CARB CHP RRM resulting in emissions reductions of 6.7 MMT and IOU and ESP/CCA retail sales of 72% presently yields a total allocation of 4.8 MMT to the IOUs and ESPs/CCAs. The amount of retail sales attributable to ESPs is currently estimated as 10% of total electricity sales in the IOU Service Territories. Reducing the 4.8 MMT by the amount of ESP/CCA retail electric sales yields a target of 4.3 MMT for the IOUs' bundled service customers.
  - 6.2.2.3.1 IOU and ESP/CCA retail sales within IOU service territories currently represent 72% of total electricity sales in the State (this calculation does not consider customer generation). The percentage of IOU Service Territory retail sales based on CEC data from 2007 is as follows: SCE (45.6 %); PG&E (43.9 %); and SDG&E (10.5 %). It is expected that, as the electric market continues to evolve, the portion of electric sales attributable to each IOU will change. Thus, the proportion of the CARB CHP RRM attributable to each IOU will change.
  - 6.2.2.3.2 As CEC data regarding percentage of retail sales is updated and published, the CPUC staff will calculate and post on the CPUC website an update to the portion of each IOU's GHG Emissions Reduction Target that is based on the CARB CHP RRM.
  - 6.2.2.3.3 During the Initial Program Period, the GHG Emissions Reduction Target is set for the IOUs, but may be adjusted among the IOUs, as a group, and the ESPs and CCAs on an individual basis if each ESP/CCA has responsibility for procuring CHP, or as a group if the CPUC requires the IOUs to purchase CHP power on behalf of the ESP/CCA, based on changes in retail electric sales. This adjustment will be based upon the updated and published CEC data.
- 6.3 ESP and Community Choice Aggregator (CCA) Portion of the CARB CHP RRM
  - 6.3.1 This Settlement adopts an initial allocation of the CARB CHP RRM of 4.3 MMT for the IOUs' bundled service customers and 0.5 MMT (4.8 MMT-4.3 MMT = 0.5 MMT) for the non-IOU load serving entities serving Direct Access (DA) customers and CCA customers in the IOUs' Service Territories (non-IOU LSEs).
  - 6.3.2 The Parties prefer that the non-IOU LSEs procure their respective shares of the CARB CHP RRM by entering into their own PPAs with CHP Facilities. If the non-IOU LSEs

- are not required to procure their respective shares of the CARB CHP RRM then, as set forth further in Section 13.1.2, the IOUs will obtain CHP and each non-IOU LSE will be allocated cost responsibility for GHG reductions attributable to CHP based on its proportion of statewide retail sales.
- 6.3.3 To the extent the amount of electric sales attributable to non-IOU LSEs increases or decreases during the term of the Settlement, the IOU and non-IOU LSE allocations of the CARB CHP RRM shall be adjusted annually by the CPUC Energy Division based on updated and published CEC retail sales data.
- 6.3.4 To the extent a CCA provides service or commences service in an IOU's Service Territory; the IOU's portion of the CARB CHP RRM and corresponding GHG Emissions Reduction Targets shall be adjusted to account for the CCA's service in that Service Territory consistent with the methodology set forth in Section 6.2.2.3.1, above. The remainder of the CARB CHP RRM, currently estimated as 1.9 MMT, should be the responsibility of POUs or entities not regulated by the CPUC.
- 6.3.5 The Parties support proportional MW Targets and GHG Emissions Reduction Targets for all POUs in California at CARB and/or in other statewide or national venues.
- 6.4 Method to Determine Each IOU's GHG Emissions Reduction Target
  - 6.4.1 For the purposes of this section regarding GHG Emissions Reduction Target counting, Existing CHP Facilities are gas-fired Topping Cycle CHP Facilities that exported and delivered electric power to an IOU as listed by QF ID number in each IOU's July 2010 Semi-Annual Report as "Contract Nameplate." The IOUs proportional share of the CARB CHP RRM in effect during the Second Program Period is established in this Settlement as 4.3 MMT, but may be superseded by a modification of the CARB CHP RRM regarding the goal of securing 6.7 MMT of incremental GHG reductions from incremental CHP resources or changes in CEC published retail electric sales.
  - 6.4.2 Add any GHG Emissions Reduction Target shortfall or subtract any surplus from Existing CHP Facilities, as determined at the end of the Initial Program Period.
    - 6.4.2.1 In 2015, after the Initial Program Period, all Parties, in conjunction with CPUC Energy Division staff, will meet and confer to determine the status of the Existing CHP Facilities from each IOU's July 2010 Semi-Annual Reports for purposes of determining any GHG Emissions Reduction Target shortfall or surplus.
    - 6.4.2.2 For Existing CHP Facilities that shut down during the Initial Program Period, the GHG reductions will be calculated against the previous two calendar years of data compared to the Double Benchmark.
    - 6.4.2.3 The net of the GHG Debit or GHG Credit calculated from the CHP Facilities that are no longer operational will be added/subtracted to the share of the CARB CHP RRM allocated to the IOUs as a group.
  - 6.4.3 Coal-fired, wood waste, and renewable CHP will count towards the GHG Emissions Reduction Targets.

- 6.4.4 Calculate each IOU's allocated portion of the CARB CHP RRM based on its percentage of retail electric sales at that time, as determined by updated and published CEC data; and,
- 6.4.5 Add or subtract net changes from each IOU's GHG Credits and GHG Debits from all sources not included in Section 6.4.2 above during the Initial Program Period.
- Each IOU's GHG Emissions Reduction Target for the Second Program Period will be calculated and submitted to the CPUC by the IOUs in their Semi-Annual CHP Program Reports.
- 6.6 The IOUs' GHG Emissions Reduction Target for the Second Program Period is subject to review and revision in the LTTP process.
- 6.7 Changes to the CARB CHP RRM
  - 6.7.1 If CARB, pursuant to an official CARB document modifies the CARB CHP RRM to revise the goal of securing 6.7 MMT of incremental GHG reductions from incremental CHP resources, the GHG Emissions Reduction Targets adopted by this Settlement will be adjusted accordingly, so long as the CPUC adopts such modification in the LTPP process. The GHG Emissions Reduction Targets may also be adjusted by the CPUC in the LTPP process, provided that changes in the GHG Emissions Reduction Targets do not affect the MW Targets specified in this Settlement.
  - 6.7.2 Changes to the GHG Emissions Reduction Targets resulting from a change to the CARB CHP RRM adopted between LTPP decisions may be implemented by the CPUC via a Tier III advice letter process.
  - 6.7.3 If the CARB CHP RRM is modified without a specific allocation by CARB to the IOUs, the methodology set forth in Section 6.2.2.3 for determining the IOU allocation of the CARB CHP RRM will remain the same, and the corresponding GHG Emissions Reduction Targets will be updated consistent with that methodology. During the Second Program Period, the CARB CHP RRM allocations will be adjusted annually by the CPUC Energy Division based upon updated and published CEC retail sales data.
  - 6.7.4 It is expected that new CHP Facilities, including Repowered CHP Facilities, may provide IOUs with meaningful GHG emission reductions to meet the CARB CHP RRM. Such expectations will be incorporated in the evaluation criteria and may change to reflect revisions to CARB's Scoping Plan or related regulations.
  - 6.7.5 AB32 or Federal Delay or Elimination

If AB 32 CARB GHG compliance is delayed or suspended or superseded by a federal program for all or any portion of the Second Program Period, the IOUs shall sustain the GHG reduction benefit derived from applying a fixed double benchmark equaling 8,300 Btu/kWh and an 80% boiler to gas-fired CHP Facilities in the Existing CHP Facilities resources fleet. In addition to sustaining the GHG reduction benefit pursuant to this subsection, the IOUs shall procure any other MW Targets or GHG Emissions Reduction Targets established in the LTPP process for the Second Program Period.

6.8 Advocating at CARB and in other Forums

- 6.8.1 The Parties agree to support and advocate for the terms of this Settlement before CARB and in all other forums.
- 6.8.2 Notwithstanding the above section, the Parties reserve the right to advocate before CARB, the CPUC or any successor agency for modification of: (a) the CARB CHP RRM; (b) the Double Benchmark used to measure GHG emission reductions from CHP; (c) the IOU obligations to procure CHP capacity in the Second Program Period, or GHG Emissions Reduction Targets from CHP Facilities.
- 6.8.3 Parties are free to advocate at CARB or the CPUC in favor of maintaining or modifying the current definition of the Double Benchmark for a CHP Facility for a period longer than the term of the PPA. For example, Parties may advocate for the application of the Double Benchmark for the operating life of the CHP Facility.
- 6.9 Justification for Failure to Meet GHG Emissions Reduction Targets

Any lOU that is unable to meet its GHG Emissions Reduction Target must make a showing to justify its inability to meet the GHG Emissions Reduction Target. Asserted justifications for missing the GHG Emissions Reduction Targets may include that:

- 6.9.1 The efficiency of the CHP Facility participating in the IOU's procurement programs is inefficient compared to the Double Benchmark, except as otherwise measured pursuant to Section 7.3;
- 6.9.2 The offer prices are in excess of levels as provided herein; and
  - 6.9.2.1 Offer prices: If the IOU claims that CHP RFO offer prices are excessive, the IOU must refer to independent or publicly-available sources. For example, when making a justification on the basis of price of new or repowered fossil fuel-fired generation, the IOU may compare offer prices to prices reflected in sources such as Cambridge Energy Research Associates' capital cost index, CPUC estimates of new fossil fuel-fired generation, CEC estimates of new fossil fuel-fired generation, forward market prices, or other similar third-party information regarding the cost of new generation CHP Facilities in California.
- 6.9.3 A lack of need exists.

# 7 GHG Emission Accounting Methodology

- 7.1 GHG Accounting Principles
  - 7.1.1 Progress toward the IOUs' GHG Emissions Reduction Targets will be determined by a GHG Credit or GHG Debit. A "+"counts as a GHG Credit which will count toward the IOU's then-current GHG Emissions Reduction Target from CHP resources. A "-" counts as a GHG Debit, which will count against the IOU's then-current GHG Emissions Reduction Target. This same methodology applies if the CPUC requires CCAs/ESPs to procure on their own behalf.
  - 7.1.2 Except otherwise noted in Section 7.3, the Parties agree to measure the amount of GHG emissions from CHP Facilities as compared to the current Double Benchmark in place at the time of PPA execution or, for a Utility Prescheduled Facility, execution of a new

- PPA or a Legacy PPA Amendment. The Double Benchmark is intended to reflect the GHG emissions that would have occurred if the same amount of electricity and thermal output were obtained from conventional generation resources and a stand-alone boiler. The Double Benchmark measures the additional amount of GHG emissions that otherwise would exist if the CHP Facility did not exist.
- 7.1.3 For the purposes of GHG accounting, an "efficient" CHP refers to one that reduces emissions as compared to the Double Benchmark. An "inefficient" CHP refers to one that increases GHG emissions as compared to the Double Benchmark.
- 7.1.4 CHP PPAs that are required by law to be executed by the IOU outside of the RFO or bilateral procurement processes, meaning PURPA contracts, feed-in tariffs (e.g., AB 1613 Feed-In Tariff) or other mandated programs or PPAs will not count as a GHG Debit to the IOU. Efficient CHP Facilities as compared to the Double Benchmark will be counted as a GHG Credit.
- 7.2 The Double Benchmark, which may be later modified pursuant to this Settlement, is as follows:
  - 7.2.1 The Heat Rate for the electricity generated is 8,300 BTU/kWh HHV at the busbar and excluding line losses.
  - 7.2.2 The thermal efficiency of a standard boiler is 80%.
  - 7.2.3 If the Double Benchmark is modified by CARB in a superseding Scoping Plan, regulation, or by the CPUC in a subsequent CPUC decision, the modified Double Benchmark shall be applied only on a prospective basis. The revised Double Benchmark shall apply to a CHP Facility that executes a PPA on or after the adoption date of the modified Double Benchmark. Changes to the Double Benchmark will not modify the MW Targets in this Settlement.
- 7.3 Detailed GHG Accounting Methodology to Measure Progress Toward the IOUs' GHG Emissions Reduction Targets
  - 7.3.1 Projects counted as a GHG Credit (+)
    - 7.3.1.1 New CHP Facilities: Efficient New CHP Facilities as compared to the Double Benchmark will count as a GHG Credit toward the contracting IOU's GHG Emissions Reduction Target regardless of where the CHP Facility is located. Measurement is based on the Double Benchmark in place at the time of PPA execution compared to the anticipated operations reflected in the PPA.
    - 7.3.1.2 Physical Change From a Repowered CHP Facility, MW Expansion, or Fuel Change (e.g., an existing coal plant converts to a less GHG intensive fuel):

      Counts as a GHG Credit for the IOU. The measurement is the difference between i) the previous two (2) calendar years of operational data compared to the Double Benchmark in place at the time of PPA execution and ii) the anticipated change in operations as identified in the PPA compared to the Double Benchmark.
    - 7.3.1.3 CHP Facility Change In Operations or Conversion to a Utility Prescheduled
      Facility: Counts as a GHG Credit. Measurement is based on the Baseline year

emissions minus the projected PPA emissions and emissions associated with replacing one hundred percent (100%) of the decreased electric generation at a time differentiated Heat Rate. The Baseline year emissions are the average of the previous two (2) calendar years of operational data.

- 7.3.1.4 Existing inefficient CHP Facility shuts down: Counts as a GHG Credit toward the CARB CHP RRM of the IOU that previously contracted with the CHP.
  - 7.3.1.4.1 If the thermal need continues, the measurement is the Double Benchmark in place at the time of the shut-down compared to the previous two (2) calendar years of operational data.
  - 7.3.1.4.2 If the thermal need no longer exists, measurement is based on the Baseline year emissions minus the projected PPA emissions and emissions associated with replacing one hundred percent (100%) of the decreased electric generation at a time differentiated Heat Rate. The baseline year emissions are the average of the previous two (2) calendar years of operational data. The IOU shall demonstrate the thermal need no longer exists.
- 7.3.2 Projects counted as a GHG Debit (-)
  - 7.3.2.1 <u>Inefficient New CHP Facilities</u>: Inefficient New CHP Facilities will count as a GHG Debit toward the contracting lOU's GHG Emissions Reduction Target. Measurement is based on the Double Benchmark in place at the time of PPA execution compared to anticipated operations reflected in the PPA.
  - 7.3.2.2 Shut-down or Retirement of an existing, efficient CHP Facility and the thermal need continues: There is a GHG Debit for the IOU who previously contracted with the CHP equal to the amount of the GHG emissions reduction of the CHP Facility as compared to the Double Benchmark against the previous two calendar years of operational data.
  - 7.3.2.3 Physical Change, a Repowered CHP Facility, MW Expansion or Fuel Change (e.g., an existing coal plant converts to a less GHG intensive fuel): A less efficient facility counts as a GHG Debit toward the IOU target. The measurement before the change is the previous two (2) calendar years of operational data compared to the Double Benchmark in place at the time of PPA execution; the measurement after the change is the anticipated operations reflected in the PPA compared to the Double Benchmark.
- 7.3.3 Projects counted as Neutral
  - 7.3.3.1 Existing CHP Facility with no change in operations: Regardless of contract status (*i.e.*, a new PPA with an Existing CHP Facility or one that sells to the market) the CHP Facility is considered neutral for GHG accounting purposes.
  - 7.3.3.2 Efficient Existing CHP Facility shuts-down and the thermal need is discontinued: If the host facility does not put in boilers, then there is no change to the IOU GHG Emissions Reduction Target.

- 7.3.3.3 Inefficient projects required by law to execute (including PURPA <20 MW, asavailable, and feed-in tariffs: Count as neutral toward the GHG Emissions Reduction Target.
- 7.4 Effective Date for Accounting of Changes in GHG Credits and GHG Debits
  - 7.4.1 The GHG benefit shall be calculated at the time of execution of the CHP PPA (includes RFO, bilateral agreement, Feed-In Tariff, as-available, PURPA <20 MWs). The calculation of the GHG Credit or GHG Debit attributable to the CHP Facility shall not be altered for the term of the PPA for the purposes of counting progress towards the IOUs' GHG Emissions Reduction Targets, regardless of a change to the Double Benchmark or modifications to the CARB Scoping Plan regarding the goal of securing 6.7 MMT of incremental GHG reductions from incremental CHP resources. If revised by a CARB determination and adopted by the CPUC in the LTPP proceeding, the modified Double Benchmark shall only apply to a CHP Facility that executes a PPA on or after the adoption of the modified Double Benchmark.
  - 7.4.2 For an efficient CHP Facility that terminates its operations, the effective date shall be the date operations terminate.
  - 7.4.3 For an existing CHP Facility with a change in operations, the effective date will be the date a new CHP PPA is executed by both Parties or, if there is no new PPA, the date on which the CHP Facility commences its change in operations.
  - 7.4.4 PPAs executed during the period between September 1, 2009 and the Settlement Effective Date count towards the IOUs' MW Targets and GHG Emissions Reduction Targets.
  - 7.4.5 SGIP or behind-the-meter CHP Facilities are counted at the time operations commence.

# 8 CPUC Jurisdictional Entities' Semi-Annual CHP Program Reports

- 8.1 General Description
  - 8.1.1 Each CPUC Jurisdictional Entity shall prepare a semi-annual report (CHP Program Semi-Annual Report) detailing progress towards both MW Targets and GHG Emissions Reduction Targets.
  - 8.1.2 Purpose: To track and report progress towards the CPUC Jurisdictional Entities' GHG Emissions Reduction Targets and MW Targets.
  - 8.1.3 Responsible Parties: The CPUC Jurisdictional Entities will each prepare and update a CHP Program Report that they will submit to the CPUC Energy Division. The CPUC Energy Division will be responsible for verifying the accuracy of the data and collating the data to develop publicly-available reports as provided herein.
  - 8.1.4 Availability: The CPUC Energy Division will post on the CPUC website public versions of the detailed updates for each CPUC Jurisdictional Entity, and maintain and publish a summary tracking progress toward both MW Targets and GHG Emissions Reduction Targets. This report shall be in addition to the IOUs' semi-annual reports on QF status.

- 8.1.5 Format: The CPUC Energy Division will determine the format of the CHP Program Reports. The Parties recommend Microsoft excel worksheet, with tabs for summary, including but not limited to PPAs with All Existing CHP Facilities, PPAs with New CHP Facilities or Repowered CHP Facilities, terminated PPAs, and with retired or shut down CHP Facilities.
- 8.1.6 Frequency: Semi-annual.
- 8.2 Overview of CHP Program Report Content
  - 8.2.1 Summary to include: CHP Facility PPAs or other procurement and termination of CHP Facility PPAs by all CPUC Jurisdictional Entities by year, number of PPAs, capacity, GHG accounting (GHG Credits and GHG Debits).
  - 8.2.2 Tracking in the report will begin upon execution of a new PPA by the Buyer and Seller for all CHP Facilities including existing, repowered or new, and excluding Transition PPAs, and Legacy PPAs (except where an amendment to a Legacy PPA results in an incremental change to GHG Emissions Reductions, in which case the tracking shall begin on the date of the execution of a Legacy PPA Amendment).
  - 8.2.3 CHP Program Reports will include:
    - 8.2.3.1 A tally of all CHP MWs deemed, pursuant to the Settlement, to count towards progress in meeting the IOUs' MW Targets, including a breakdown by IOU Service Territory.
    - 8.2.3.2 A tally toward overall GHG Emissions Reduction Targets, including the GHG emissions reductions attributable to CHP Facilities that currently have PPAs with the IOUs, including breakdown by IOU Service Territory.
    - 8.2.3.3 Each tally shall include MWs, GHG Credits and GHG Debits and shall be broken down by the following mutually exclusive categories:
      - 8.2.3.3.1 Executed CHP Facility PPAs (but not yet delivering),
      - 8.2.3.3.2 Operational CHP Facility PPAs,
      - 8.2.3.3.3 Amended PPAs, and
      - 8.2.3.3.4 Terminated CHP Facility PPAs including facility shut downs.
- 8.3 Report Content for New PPAs
  - 8.3.1 The following table shows the public report content for:
    - 8.3.1.1 Existing CHP Facilities entering into new PPAs (excluding Transition PPAs or Legacy PPA C1 Amendments)
    - 8.3.1.2 New CHP Facilities, Repowered CHP Facilities and Expanded Facilities

Item	Description
All Identification Numbers	Facility ID number and name, as per the PPA/CHP PPA
	CAISO ID number, CEC ID number, Energy Information Administration (EIA) ID number,
	CARB ID number
Location	IOU Service Territory/ESP/CCA etc.
Type of Program under which CHP Facility has been procured	Bilateral, CHP-only RFO, All Source RFO, AB 1613 Feed-In Tariff, As-Available, PURPA ≤ 20 MW, SGIP, Utility Prescheduled Facility PPA, Section 3.07 modified CHP Facilities
Power Product	CHP Facility Contract Nameplate (MW) (define in report)
	Contract Capacity (MW)
	Energy deliveries under the Optional As-Available Program measured in AMW
Fuel Type	For example: natural gas, biomass, pet-coke, etc.
Contract Execution Date	Mm/dd/yyyy
Online Date	Mm/dd/yyyy
GHG Credits/ GHG Debits	Reflect figures pursuant to GHG accounting rules in Section 7 of the Settlement to determine GHG Credits/GHG Debits, and effective dates of change to GHG accounting. This value remains static for counting purposes for the life of the PPA.
Double Benchmark	The Double Benchmark reference point, e.g., 8,300 Btu/kWh and 80% boiler efficiency.
Technology Type	Microturbine, etc.
Topping Cycle or Bottoming	Topping Cycle or Bottoming Cycle Classification

Item	Description
Cycle	

- 8.4 Report Content for Legacy PPA C1 Amendments, Utility Prescheduled Facilities or Other Changes in Operations
  - 8.4.1 The following table shows the public report content for Legacy PPA C1 Amendments, Utility Prescheduled Facilities or other changes in operations.
  - 8.4.2 Amendments to Legacy PPAs for CHP Facilities that convert to Utility Prescheduled Facilities will record the same information as in the table in Section 8.3 plus:

Item	Description
All Identification Numbers	Facility ID number and name, as per the PPA/CHP PPA
	CAISO ID number, CEC ID number, EIA ID number,
	CARB ID number
Location	IOU Service Territory/ESP/CCA etc.
Amendment Date	Mm/dd/yyyy
GHG Credits /GHG Debits	Reflect figures pursuant to GHG accounting rules in Section 7 of the Settlement to determine GHG Credits/GHG Debits, and effective dates of change to GHG accounting.
GHG Credit /GHG Debit Measurement	As per the GHG accounting rules in Section 7.
Effective Date of Change in Operations	Mm/dd/yyyy

- 8.5 Report Content for Terminated PPAs and Facilities that Cease Operations
  - 8.5.1 The following table shows the public report content for Terminated PPAs and facilities that cease operations.
  - 8.5.2 Terminated PPAs with CHP Facilities (includes CHP Facilities with PPAs that are terminated for any reason during the Settlement Term (i.e., expired, retired, or facilities that cease operations).

8.5.2.1 Facilities that cease operations shall indicate whether the thermal need remains or is discontinued.

Item	Description
Facility Name	Name as it appears in the PPA
All Facility ID Numbers	Facility ID number and name, as per the PPA, CAISO ID number, CEC ID number, EIA ID number, CARB ID number
Location	IOU Service Territory/ESP/CCA, etc.
Type of Program under which CHP had been procured	Bilateral, CHP-only RFO, All Source RFO, AB 1613 Feed-In Tariff, As-Available, PURPA ≤ 20 MW, SGIP, Utility Prescheduled Facility PPA, Legacy PPA, Section 3.07 modified facilities
Power Product	Facility Contract Nameplate (MW)
	Energy deliveries under the Optional As-Available PPA measured in AMW
Fuel Type	For example: natural gas, biomass, pet-coke, etc.
PPA Termination or Ceases Operating	Mm/dd/yyyy
Thermal Load Status	Discontinued/Ongoing
GHG Credits /GHG Debits Measurement	As per the GHG accounting rules in Section 7.

# 9 CHP Auditor

# 9.1 Description of CHP Auditor

- 9.1.1 This section addresses the role of the CHP Auditor(s) for the CHP RFOs in the Settlement. This section also sets forth non-disclosure obligations for Confidential Information, the right to seek and secure information, designation criteria, and other matters.
- 9.1.2 Purpose and Role of the CHP Auditor(s). The purpose and role of a CHP Auditor is to be an informed advocate for CHP interests regarding the implementation of the CHP

- Program. CHP Auditor(s) will only be designated under this section if, upon written notice, an IOU does not, or anticipates it will not, meet any of the following: a MW Target established by this Settlement; a GHG Emissions Reduction Target established by this Settlement; or any MW Target or GHG Emissions Reduction Target established by the CPUC in the LTPP after the Settlement Effective Date.
- 9.1.3 Payment of CHP Auditor(s). The CHP Parties may retain the CHP Auditor(s) subject to the conditions and approval processes set forth herein. The CHP Parties will bear the costs of the CHP Auditor. The sponsoring CHP Party (or Parties) for any individual CHP Auditor will bear the costs of such CHP Auditor.
- 9.1.4 Permitted Reporting by the CHP Auditor and the Applicable Non-Disclosure Agreement. The CHP Auditor(s) shall execute and is subject to non-disclosure restrictions set forth in the CHP Auditor Non-Disclosure Agreement (NDA), which is attached as Exhibit 8. With the exception of the twenty-four (24) month restriction in Section 9.5.3, if the CPUC, the courts or another legal authority modifies the requirements applicable to a reviewing representative of IOU Confidential Information, the NDA shall be updated to reflect such modifications (Applicable NDA). The CHP Auditor shall be bound by the terms and conditions of the Applicable NDA executed for a particular RFO on the conditions in the Applicable NDA at the time of execution. Updates to Applicable NDAs, if any, as provided under this section, shall apply only prospectively for NDAs executed after such updates. Subject to the terms and conditions of the Applicable NDA, the CHP Auditor may:
  - 9.1.4.1 Before the CPUC, and all divisions thereof, and the IOU's Procurement Review Group (CPUC forum) report on the IOU's conduct of and procurement decisions arising from a particular CHP RFO, using any information, including the Confidential Information obtained by the CHP Auditor from the IOU.
  - 9.1.4.2 Before any other forum, report, assess, recommend or advocate CHP procurement policies, targets or actions based upon the CHP Auditor's experience auditing a CHP RFO or other RFOs, including the procurement of additional CHP capacity to meet CHP procurement targets established by the CPUC, additional actions to promote the success of the CHP Program, or advocate any other action related to CHP policy or procurement that the Auditor deems appropriate. Actions by the Auditor under this section shall be consistent with the goals and objectives of the Settlement, and the Auditor(s) may not reveal Confidential Information pursuant to the Applicable NDA. For actions under this section the Auditor(s) may express qualitative opinions regarding the CHP Program and related matters based upon the Auditor's experience, provided that the CHP Auditor does not reveal Confidential Information.
  - 9.1.4.3 Before any other forum, the CHP Auditor(s) shall secure the advance approval of any affected IOU of the disclosure of any aggregated data derived from the IOU's Confidential Information that the CHP Auditor proposes to disclose. The affected IOU's advance approval shall be timely in contemplation of the circumstances of the disclosure and the IOU's consent shall not be unreasonably withheld. The IOU's decision on whether to allow use of data must be based on CPUC confidentiality rules in place at that time and on the terms of the Applicable NDA.

- 9.1.4.4 The Auditor(s) may also reference any public information, including any information publicly disclosed by any IOU Independent Evaluator.
- 9.1.4.5 The NDA, among other things, shall provide procedures for the Auditor to comply with lawful orders from a commission, court or other lawful authority compelling disclosure of Confidential Information.

## 9.2 IOU Notice Triggering Audit

- 9.2.1 An IOU will provide written notice to those CHP Parties on the new service list for this Settlement that the CPUC will create for the CPUC review of this Settlement, that it will not meet, or anticipates that it will not meet any of the following: a MW Target established by this Settlement; a GHG Emissions Reduction Target established by this Settlement; or any MW Target or GHG Emissions Reduction Target established by the CPUC in the LTPP after the Settlement Effective Date.
- 9.2.2 An IOU shall provide notice at the earliest practicable time based upon an IOU's expectation that it will not achieve any one of the following in a particular CHP RFO: a MW Target established by this Settlement; a GHG Emissions Reduction Target established by this Settlement; or any MW Target or GHG Emissions Reduction Target established by the CPUC in the LTPP after the Settlement Effective Date; provided that such notice shall be no later than the earlier of (1) Buyer's Advice Letter requesting approval of results from any CHP RFO solicitation, or (2) within five (5) Business Days of any presentation by the Buyer to the Buyer's PRG indicating that the Buyer will not achieve one of the above targets.

#### 9.3 Time Period for Audit Review

- 9.3.1 The CHP Auditor(s) shall have up to ninety (90) calendar days (Audit Period) from the date of the receipt of the Confidential Information that was previously provided to the Independent Evaluator for a particular CHP RFO to present findings to the CPUC. If the CHP Auditor(s) require(s) additional time to present such findings, the Director of the CPUC Energy Division may, upon good cause shown, extend the Audit Period up to one hundred and eighty (180) calendar days. The CHP Auditor shall provide a written request to the Director of the CPUC Energy Division for such extension with a copy of the request simultaneously provided to the affected IOU.
- 9.3.2 Any pending CHP audit will not provide good cause for a delay in the IOU's completion of the procurement process for any CHP RFO.
- 9.3.3 A CHP Audit that exceeds ninety (90) calendar days will not provide good cause for delay of an IOU in commencing the next CHP RFO; provided that the IOU may condition procurement in subsequent RFOs upon final resolution of pending RFO audits in order to avoid procurement in excess of established MW Targets or GHG Emissions Reduction Targets.
- 9.4 Receipt and Review of Confidential Information and Other Relevant Data
  - 9.4.1 The CHP Parties shall provide the IOU with notice of the identity of the designated CHP Auditor(s) for the CHP RFO. Within ten (10) Business Days of the receipt of such notice, the IOU will provide the Confidential Information to the CHP Auditor(s)

that was provided to the Independent Evaluator, if the CHP Auditor has executed and submitted an Applicable NDA as provided by this Section 9 with the particular IOU whose RFO is audited. The CHP Auditor(s) shall not be entitled to receive Confidential Information prior to the time the IOU receives the CHP Auditor's executed NDA.

- 9.4.2 At a minimum the Confidential Information shall include all information provided to the subject CHP RFO's Independent Evaluator. Notwithstanding anything to the contrary in this Settlement, the CHP Auditor is not entitled to review any proprietary models used by an IOU in an RFO, provided that all inputs and outputs of the model used in the RFO shall be provided to the CHP Auditor.
- 9.4.3 In addition to the right to secure access to all information provided by the IOU to the Independent Evaluator working on the particular CHP RFO, the CHP Auditor(s) shall also have the right to secure additional information that is reasonably required to evaluate the IOU procurement decisions in connection with a CHP RFO. The IOUs shall respond in full to all data requests from CHP Auditor(s) within ten (10) Business Days. The Parties agree that the Director of the CPUC Energy Division shall promptly address and resolve any disputes over the relevance of or access to data sought by the CHP Auditor(s).
- 9.4.4 The Parties agree that there is no waiver of rights or positions regarding access to confidential data resulting from future changes by the CPUC, court or other lawful authority on the issue of confidentiality. The obligations and responsibilities of the CHP Auditor(s) with regard to maintaining confidentiality of the Confidential Information shall be specified in the Applicable NDA.
- 9.5 Designation, Notice, Conflict of Interest Review and Number of CHP Auditors
  - 9.5.1 The CHP Parties shall endeavor to agree upon a single CHP Auditor, but may designate no more than two CHP Auditors for each CHP RFO under this section; provided that IEP shall have the right to designate one Auditor, and collectively CCC-CAC-EPUC shall have the right to designate one Auditor.
  - 9.5.2 The CHP Parties shall provide notice to the CPUC Energy Division director (Director) of the name of any designated CHP Auditor(s) for a specific conflict of interest review. The Director shall evaluate the curriculum vitae for any designated auditor, determine if any applicable conflict of interest restrictions exist for the designee(s) within five (5) Business Days and notify the lOU that the Director has determined that the designated CHP Auditor(s) is/are acceptable. The Director will certify the designated CHP Auditor(s) within five (5) Business Days, during which the Director shall review conflict(s) of interest for the CHP Auditor(s) under the following conditions: (1) the direct ownership interest in any bidder in the particular RFO subject to the CHP audit, or (2) a breach of a non-disclosure agreement in an CHP RFO audit review or any other NDA. If the Director identifies any other conflict of interest standard that the Director reasonably believes needs to be applied to the CHP Auditor, the Director shall meet and confer with all affected Parties to discuss the need for such other standard(s).
  - 9.5.3 Each CHP Auditor must certify in writing, prior to the receipt of Confidential Information, that he or she will not engage, beginning on the date of the delivery of Confidential Information, and thereafter for a period of twenty four (24) months in: (a)

a transaction for the generation, purchase, sale or marketing of electrical energy, and/or capacity, and/or related products, including specifically electricity related financial products (meaning derivatives, swaps or options), at wholesale in the State of California, (b) a transaction for the purchase, sale or marketing at wholesale of natural gas commodity, assets, including specifically natural gas related financial products (meaning derivatives, swaps or options), for electric generation, purchase or sale purposes in the State of California, (c) preparing offers and/or bidding strategies, bidding on, or purchasing of electric power plants in the State of California or the substantive supervision of any employees whose duties include such responsibilities with regard to those activities, subject to the following Section 9.5.4, or (d) mergers and/or acquisitions of entities that own or control electric generation and/or natural gas assets or commodity associated with electric generation in the State of California, (e) consulting with or advising others in connection with any activity set forth in subparagraphs (a), (b), (c) or (d) of this Section 9.5.3.

9.5.4 The CHP Auditor(s) may not share the Confidential Information with any third party, including any co-worker or employee, except to provide necessary technical, administrative and clerical support of no more than three (3) individuals for the Auditor's work; provided that such party is also subject to the NDA for the affected IOU and each provision of Section 9. The CHP Auditor may directly supervise employees, office colleagues or co-workers, but shall establish rules to eliminate any substantive supervision of activities identified in Section 9.5.3, above. For example, any third party, including a supervisor, employee, office colleague or co-worker of a CHP Auditor shall not have any substantive involvement in reviewing, providing guidance to or reviewing the results of the analysis derived from the Confidential Information.

# 10 SRAC Energy Pricing Structure

- 10.1 Applicability
  - 10.1.1 The SRAC energy price applies to the following:
    - 10.1.1.1 Transition PPAs;
    - 10.1.1.2 Legacy PPAs, if another option in Section 11 is not selected;
    - 10.1.1.3 QF PPAs; and
    - 10.1.1.4 Optional As-Available PPAs.
- 10.2 Methodologies and Formulae
  - 10.2.1 The following formulas and methodologies for determining the SRAC energy price shall replace the CPUC-adopted SRAC formula as of the first day of the second calendar month beginning after the Settlement Effective Date (SRAC Commencement Date) and are set forth below:
    - 10.2.1.1 Method for Determining the SRAC Energy Price. Subject to Sections 10.2.2 through 10.2.7, as of the SRAC Commencement Date, the SRAC energy price must be calculated using the following formula:

Energy Price \$/kWh = ((Applicable HR \* BTGP/1,000,000) + VOM) \* TOU + LA + GHG Charges

Where:

Applicable HR = The Heat Rate for the specified time-period, per the following table:

Calendar Year(s)	Heat Rate (Btu/kWh)
2011	8,700
2012	8,225
January 1, 2013 through	8,125
December 31, 2014	
January 1, 2015 and beyond	Market Heat Rate

BTGP = Calendar month Burner Tip Gas Price (\$/MMBtu), per D.07-09-040 and CPUC Resolution E-4246 (as such Resolution may be modified from time to time by the CPUC);

VOM = Calendar month avoided variable O&M (\$/kWh), per D.07-09-040 and CPUC Resolution E-4246 (as such Resolution may be modified from time to time by the CPUC);

GHG Charges = means all taxes, charges or fees, assessed with the implementation and regulation of Greenhouse Gas emissions with respect to the Generating Facility imposed by any Governmental Authority, such as the CARB's AB 32 Cost of Implementation Fee (as defined in Title 17 C.C.R. §95200). For example, if the charges are assessed on but not included in fuel consumption or gas costs, the Applicable Heat Rate or Burner Tip Gas Price will be used to derive the dollars per kilowatt-hour charge. On January 1, 2015 or the commencement of the First Compliance Period, the GHG Charges will equal zero in the above formula.

TOU (i.e., time-of-use) = The TOU factors are, as of the SRAC Commencement Date, as follows for each IOU:

	TOU Factors if Buyer is PG&E		
	Summer	Winter	
Peak	1.2564	N/A	
Partial-Peak	1.1535	1.1395	
Off-Peak	0.9155	0.9628	
Super Off-Peak	0.7439	0.8216	

Off-Peak TOU factors will be calculated as a residual – similar to the current method – to preserve the correctness of the monthly hourly weighting.

An example for Period A – Summer is: [Number of hours in month – (1.2564 \* Number of Summer Peak hours in month) – (1.1535 \* Number of Summer Partial-Peak hours in month) – (0.7439 \* Number of Summer Super Off-Peak hours in month)] / Number of Summer Off-Peak hours in month

	TOU Factors if Buyer is SCE		
	Summer	Winter	
On-Peak	1.4251	N/A	
Mid-Peak	see below	1.2185	
Off-Peak	0.8526	see below	
Super Off-Peak		0.7760	

Summer Mid-Peak = (Total # hours in month - (1.4251 \* # of Summer On-Peak hours in month) - (0.8526 \* # of Summer Off-Peak hours in month)) / # of Summer Mid-Peak hours in month

Winter Off-Peak = (Total # hours in month - (1.2185 \* # of Winter Mid-Peak hours in month) - (0.7760 \* # of Winter Super Off Peak hours in month)) / # of Winter Off-Peak hours in month

	TOU Factors if Buyer is SDG&E		
	Summer	Winter	
On-Peak	1.411	1.224	
Semi-Peak	1.106	1.106	
Off-Peak	0.986	0.933	
Super Off-Peak	0.645	0.711	

PG&E and SDG&E may update the TOU factors in their respective SRAC postings at the beginning of each calendar year using the energy-only portion of the time-of-use factors (as adjusted by PG&E and SDG&E, as applicable and if necessary, to reflect their respective CPUC-approved time-of-use periods) set forth in their most recent RPS Program solicitation (e.g., 2012 time-of-use factors are those used in Buyer's 2011 RPS Program solicitation). SCE may update its TOU factors annually through an appropriate filing with the CPUC.

LA (i.e., hourly location adjustment, in \$/kWh) = LMP<sub>OF</sub> - LMP<sub>Trading Hub</sub>

Where the hourly location adjustment will be based on the hourly Day-Ahead prices and actual hourly generation by the Generating Facility for delivery to Buyer as follows:

 $LMP_{QF}$  (in \$/kWh) = The hourly Day-Ahead Locational Marginal Price at the point of interconnection with the CAISO-controlled electric system associated with the Generating Facility; and

LMP<sub>Trading Hub</sub> (in \$/kWh) = The hourly Day-Ahead Locational Marginal Price of the trading hub where the Generating Facility is located (i.e., SP15 Existing Zone Generation Trading Hub (formerly SP15), NP15 Existing Zone Generation Trading Hub (formerly NP15), or ZP26 Existing Zone Generation Trading Hub (formerly ZP26), as applicable, or any successor thereto).

10.2.2 Energy Price During the Floor Test. If there is a cap-and-trade program in California for the regulation of GHG, as established by CARB (and/or by a different Governmental Authority pursuant to federal or state legislation), then, during the Floor

Test Term, the SRAC energy price will be the higher of the two formulas provided in Sections 10.2.2.1 and 10.2.2.2 (the GHG Floor Test):

10.2.2.1 Energy Price \$/kWh = ((Market Heat Rate \* BTGP/1,000,000) + VOM) \* TOU + LA

Where:

Market Heat Rate (Btu/kWh) = As defined in Section 17;

BTGP (\$/MMBtu) = As set forth above;

VOM (\$/kWh) = As set forth above;

TOU =As set forth above; and

LA (\$/kWh) = As set forth above.

OR

10.2.2.2 Energy Price \$/kWh = ((Applicable HR \* (BTGP + GHG Allowance Price) /1,000,000) + VOM) \* TOU + LA + GHG Charges

Where:

Applicable HR = (A) 8,225 Btu/kWh through December 31, 2012, (B) 8,125 Btu/kWh from January 1, 2013 through December 31, 2014; and (C) Actual HR (as defined in Section 17) from January 1, 2015 until the end of the Floor Test Term;

BTGP (\$/MMBtu) = As set forth above;

GHG Allowance Price (\$/MMBtu) = Allowance Cost (\$/MT) \* 117lbs of GHG per MMBtu / 2,204.6 lbs per MT

Where:

Allowance Cost (\$/MT) = The cost of one Allowance, determined using the GHG Auction clearing price from the latest GHG Auction that has taken place during the calendar quarter immediately preceding the date that Buyer's payment is due to Seller; <u>provided</u>, <u>however</u>, that if there is no GHG Auction held during the applicable time-period, then the Allowance Cost is determined in accordance with Section 10.2.6.

VOM(\$/kWh) = As set forth above;

GHG Charges (\$/kWh) = As set forth above;

TOU =As set forth above; and

LA ( $\frac{kWh}{=}$  As set forth above.

- 10.2.3 Free Allowance Reporting and Allocation. If, at any time, Buyer makes a monthly payment to Seller utilizing the GHG Floor Test formula set forth in Section 10.2.2.2, then Buyer shall deduct from the monthly payment to Seller for the applicable month the value of the Free Allowances disclosed in and based on all Free Allowance Notices that have not already been applied to a prior payment to Seller; provided, however, that if Buyer, using reasonable efforts, is unable to process such payment adjustment for the applicable month, then Buyer shall make such payment adjustment to the next monthly payment due to Seller. For any month that Buyer utilizes the formula set forth in Section 10.2.2.1 to make a monthly payment to Seller, Buyer shall maintain a record of the value and quantity of all Free Allowances disclosed in the Free Allowance Notices, if any, and shall deduct the value of such Free Allowances to any subsequent monthly payment due to Seller where Buyer calculates such monthly payment utilizing the formula set forth in Section 10.2.2.2 until such time that the value of all such Free Allowances are expended.
- 10.2.4 In order for Buyer to make the payment adjustment set forth in the immediately preceding paragraph, Seller agrees to deliver to Buyer, within twenty (20) calendar days of receiving any Free Allowances, a Free Allowance Notice for the applicable month, which Free Allowance Notice must include all Additional GHG Documentation. Buyer shall value any such Free Allowances using the same methodology Buyer uses in valuing the Allowance Cost, as set forth above.
- 10.2.5 Buyer will review the Annual GHG Reports described in Section 10.3 to determine if there is any discrepancy in the payments made by Buyer to Seller for GHG Compliance Costs during the course of the applicable calendar year. To the extent Buyer determines that there is any such discrepancy, (i) if Buyer owes Seller an additional payment for GHG Compliance Costs, then Buyer shall make such additional payment in a subsequent monthly payment to Seller under the Applicable PPA, or (ii) if Seller owes Buyer a payment refund for GHG Compliance Costs, then Buyer shall offset such payment refund amount in a subsequent monthly payment to Seller under the Applicable PPA. If the Applicable PPA terminates before Buyer is able to make such additional payment for GHG Compliance Costs or offset such GHG Compliance Costs payment refund from Seller's monthly payments, as applicable, then Buyer or Seller, as applicable, shall pay all remaining payment amounts due within the thirty- (30) day period before the termination of the Applicable PPA.
- 10.2.6 Determining Allowance Costs under the GHG Floor Test if there is No GHG Auction. This Section 10.2.6 is applicable if no GHG Auction has been held during the time-period for which the Allowance Cost variable set forth in Section 10.2.2.2 is to be determined. In such an instance, publicly available indices will be used to determine the price for the applicable period. If no such indices exist, SCE, SDG&E and PG&E, on the one hand, and CAC, EPUC, CCC and IEP, on the other hand, shall negotiate in good faith to reach an agreement on setting the Allowance Cost variable. If, after negotiating for fifteen (15) Business Days, the Parties are not able to reach agreement on setting the Allowance Cost variable, then SCE, SDG&E and PG&E, on the one hand, and CAC, EPUC, CCC and IEP, on the other hand, shall each select, within fifteen (15) calendar days after such failed negotiations, price quotations from two (2) different Reference Market-Makers, for a total of four (4) price quotations. The Allowance Cost variable for the applicable time-period will be determined by taking the average of the four (4) price quotations so selected by the Parties.

10.2.7 Energy Price from the end of the Floor Test Term. As of the end of Floor Test Term, the SRAC energy price must be calculated in accordance with the following formula:

Energy Price  $\frac{h}{h} = ((Market Heat Rate * BTGP/1,000,000) + VOM) * TOU + LA$ 

Where:

Market Heat Rate (Btu/kWh) = as defined in Section 17;

BTGP (\$/MMBtu) = As set forth above;

VOM(\$kWh) = As set forth above;

TOU =As set forth above; and

LA (\$/kWh) = As set forth above.

#### 10.3 Reporting Requirements

- 10.3.1 From the SRAC Commencement Date through the end of the term of the Applicable PPA (and for any period following the termination of the Applicable PPA to the extent relating back to the term of the Applicable PPA), Seller shall provide to Buyer the following information (together, the Annual GHG Reports):
  - 10.3.1.1 On or before the fifth (5th) Business Day following Seller's timely submission to CARB (or any other authorized Governmental Authority having jurisdiction in California) of the CARB Mandatory GHG Emissions Annual Report, or such other annual report submitted to CARB, detailing the GHG emissions of the Generating Facility for the applicable calendar year (as verified by an independent third party, if applicable) (the CARB Annual Report), Seller shall deliver such CARB Annual Report to Buyer; and
  - 10.3.1.2 To the extent not set forth in the CARB Annual Report (or if Seller is no longer required to submit the CARB Annual Report for any reason), then Seller shall submit to Buyer, along with the CARB Annual Report (or, if Seller is no longer required to submit the CARB Annual Report for any reason, then on the sixtieth (60th) Business Day following the end of the applicable calendar year), the following information for the applicable calendar year, which, in each case, must be verifiable and of settlement quality: (1) the Useful Thermal Energy Output of the Generating Facility; (2) total fuel usage of the Generating Facility; (3) the total amount of GHG emissions attributable to the Generating Facility, the electrical energy used to serve the Site Host Load, the Useful Thermal Energy Output of the Generating Facility; (4) the total electrical energy produced by the Generating Facility, the electrical energy used to serve the Site Host Load, and the energy delivered to Buyer; and (5) the number of Allowances (including Free Allowances) held and/or surrendered by Sellcr for such calendar year (during any period where the SRAC energy price is calculated in accordance with Section 10.2.2).
- 10.3.2 If Buyer requires any other information not delineated in Section 10.3.1 in order to comply with any GHG emissions reporting requirements adopted by CARB and/or by

any other Governmental Authority and imposed on Buyer (other than the information that Seller must provide in accordance with Section 10.3.3), then SCE, SDG&E and PG&E, on the one hand, and CAC, EPUC, CCC and IEP, on the other hand, shall promptly meet and confer regarding such other information that Buyer requires and negotiate in good faith to reach a mutually acceptable agreement. Buyer and Seller shall be bound by any agreement as to any information required by Buyer, as described in the foregoing, between PG&E, SCE and SDG&E, on the one hand, and CAC, EPUC, CCC and IEP, on the other hand, in accordance with the foregoing.

- 10.3.3 Each Party shall deliver to the other Party, or before the tenth (10th) Business Day following receipt of a notice from the other Party, such information as such other Party is required to report to any authorized Governmental Authority pursuant to the Settlement.
- 10.3.4 To the extent that the information provided by the disclosing Party in accordance with this Section 10.3 is Confidential Information, the receiving Party shall treat such Confidential Information with the same degree of care that it currently treats the data and information provided by QFs under the existing CHP QF Compliance Monitoring Program.

#### 10.4 Market Disruption Event

10.4.1 If, on or after the date that the Market Heat Rate applies to and is used in the calculation of the energy price and until the end of the term of the Applicable PPA, there occurs a Market Disruption Event, then the Market Heat Rate for the affected Trading Day(s) must be determined by reference to the Market Heat Rate for the first Trading Day thereafter on which no Market Disruption Event exists; provided, however, that if the Market Heat Rate is not so determined within five (5) Trading Days after the Market Disruption Event, then PG&E, SCE and SDG&E, on the one hand, and CAC, EPUC, CCC and IEP, on the other hand, will negotiate in good faith to reach an agreement on a Market Heat Rate (or a method for determining a Market Heat Rate), and if such negotiating parties have not so agreed on or before the twelfth (12th) Trading Day following the first Trading Day on which the Market Disruption Event occurred or existed, then the Market Heat Rate will be determined in good faith by taking the average of the price quotations for energy and relevant Trading Days that are obtained from no more than two (2) Reference Market-Makers selected by PG&E, SCE and SDG&E, on the one hand, and CAC, EPUC, CCC and IEP on the other hand (for a total of four (4) price quotations). Buyer and Seller shall be bound by any agreement as to a Market Heat Rate (or a method for determining a Market Heat Rate) between PG&E, SCE and SDG&E, on the one hand, and CAC, EPUC, CCC and IEP, on the other hand, in accordance with the foregoing.

#### 10.5 Seller's Responsibility

10.5.1 Other than Buyer's payment to Seller for GHG Compliance Costs and GHG Charges as set forth in payment formulae in Section 10.2 above, Seller is solely responsible for all GHG Compliance Costs and all other costs associated with implementation and regulation of GHG emissions with respect to Seller or the Generating Facility.

#### 11 Legacy PPA Matters for All Existing QFs

#### 11.1 Energy and Capacity Pricing

- 11.1.1 Unless a Legacy PPA is otherwise amended to specify an alternate energy price or pricing methodology, SRAC shall be as set forth in Section 10.
- 11.1.2 QFs with Legacy PPAs may execute the Amendment to Legacy PPAs, which shall include energy pricing options identified as Options A, B, C1, C2, and C3, and other terms, as attached hereto as Exhibits 1, 2, and 3.
- 11.1.3 If no capacity price is set in the Legacy PPAs, other than the administratively determined price, the capacity price shall be as set in D.07-09-040.

#### 11.2 Other Matters

- 11.2.1 The Parties' objective is to assure that a CHP or Utility Prescheduled Facility operating under an extension ordered by the Commission in D.07-09-040 will be able to deliver power without interruption pursuant to the extension of the Legacy CHP PPA until the first day of the term, meaning the initial date of the delivery of power, of a new or amended PPA (Subsequent PPA). Extensions of Legacy CHP PPAs ordered by the Commission pursuant to D.07-09-040 shall remain in effect until the date the Seller commences power deliveries under a Subsequent PPA pursuant to this Settlement. For example, the end delivery date of the extension of the Legacy CHP PPA would be Monday and the first day of the term of the Subsequent PPA would be Tuesday. This would allow continued delivery of power without interruption under an approved and effective Subsequent PPA. The Parties shall use all reasonable efforts to meet conditions that would permit transition from the extensions to an approved and effective Subsequent PPA within one hundred and twenty (120) days after the Settlement Effective Date. Absent good cause shown, the extension of the Legacy CHP PPA shall terminate and the term of the Subsequent PPA commence no later than one hundred and twenty (120) days after the Settlement Effective Date. Good cause shall include pending regulatory approvals from the CPUC, CAISO or other Governmental Authority that prevents the delivery of power under a Subsequent PPA. In the event of a dispute, the QF may submit a letter to the Director of the CPUC Energy Division (with a copy to the 10U with a right to respond) setting forth the particular facts that explain why the CHP Facility cannot move to the Subsequent PPA and asking for a further extension. The Director of Energy Division will act on this letter within thirty (30) days and the extension of the Legacy CHP PPA shall remain in effect during the period of review. Such requests shall not be unreasonably repetitive or designed primarily to delay termination of the extension of the Legacy CHP PPA The Parties will comply with Commission orders or actions of the Director of CPUC Energy Division related to the termination of the extension of the Legacy CHP PPA.
- 11.2.2 Legacy PPAs, and extensions of PPAs ordered by the CPUC, shall remain in effect under the terms and conditions specified in such PPAs and extensions of such PPAs, subject to the following section.
- 11.2.3 A QF with a Legacy PPA can bid into a CHP RFO or otherwise make use of any applicable CHP Procurement Process.

- 11.3 Non-Binding Forecasting Requirements for Legacy PPAs
  - 11.3.1 General Requirements. The Buyer and Seller shall make good faith efforts to abide by the forecast requirements and procedure described below and shall agree upon reasonable changes to these requirements and procedures from time to time as necessary to:
    - 11.3.1.1 Support Buyer's compliance with the CAISO's scheduling requirements related to the PPA;
    - 11.3.1.2 Accommodate changes to the Buyer's and/or Seller's respective generation technology and organizational structure; and
    - 11.3.1.3 Address changes in the operating and Scheduling procedures of Seller, Buyer and the CAISO, including automated Forecast and outage submissions.
  - 11.3.2 The Buyer and Seller agree that the Forecasts generated by, or otherwise resulting from, the forecasting requirements and procedures in this Section 11.3 are non-binding on Seller, the Generating Facility or the Site Host.
  - 11.3.3 Seller's Forecasting Submittal Procedures
    - 11.3.3.1 30-Day Forecast.
      - 11.3.3.1.1 No later than 30 days before the first day of the following Initial Forecast Month, Seller shall provide Buyer with a Forecast for the 30-day period commencing on the first day of the Initial Forecast Month using the Web Client.
      - 11.3.3.1.2 If the Web Client becomes unavailable, Seller shall provide Buyer with the Forecast by e-mail at [ Buyer to provide contact information ] or by telephoning Buyer's generation operations center at [ Buyer to provide contact information ].
      - 11.3.3.1.3 The Forecast, and any updated Forecasts provided pursuant to Section 11.3.3 shall (i) not include any anticipated or expected electric energy losses between the meter(s) used for measuring the energy sold to Buyer by Seller and the point of delivery of the energy delivered to Buyer by Seller, and (ii) limit hour-to-hour Forecast changes to no less than 250 kWh during any period when the Web Client is unavailable. Seller shall have no restriction on hour-to-hour Forecast changes when the Web Client is available.
    - 11.3.3.2 Weekly Update to 30-Day Forecast.
      - 11.3.3.2.1 Commencing on or before 5:00 p.m. California time of the Wednesday before the first week covered by the Forecast provided pursuant to Section 11.3.3.1, and on or before 5:00 p.m. California time every Wednesday thereafter for the remainder of the term of the PPA, Seller shall update the Forecast for the 30-day period

- commencing on the Sunday following the weekly Wednesday Forecast update submission.
- 11.3.3.2.2 Seller shall use the Web Client, if available, to supply this weekly update or, if the Web Client is not available, Seller shall provide Buyer with the weekly Forecast update by e-mailing or telephoning Buyer at the e-mail address or telephone number listed in Section 11.3.3.1.

#### 11.3.3.3 Further Update to 30-Day Forecast.

- 11.3.3.3.1 As soon as reasonably practicable and commensurate with Seller's knowledge, Seller shall provide Forecast updates to take into account expected changes in daily, hourly and real-time deliveries from the Generating Facility for any cause, including changes in the Generating Facility ambient conditions, a Forced Outage or a Real-Time Forced Outage, any of which results or is expected to result in a material change to the Generating Facility's deliveries (whether in part or in whole).
- 11.3.3.3.2 This updated Forecast pursuant to this Section 11.3 must be submitted to Buyer via the Web Client by no later than:
  - 11.3,3.3.2.1 5:00 p.m. California time on the day before the day in which Day -Ahead trading occurs in accordance with the Western Electricity Coordinating Council Preschedule Calendar (as found on the Western Electricity Coordinating Council's website) is impacted by the change, if the change is known to Seller at that time,
  - 11.3.3.3.2.2 The Hour-Ahead Scheduling Deadline, if the change is known to Seller at that time,
  - 11.3.3.3.2.3 If the change is not known to Seller by the timeframes indicated in Sections 11.3.3.2.1 and 11.3.3.3.2.2 immediately above, no later than 20 minutes after Seller becomes aware of the event which caused the expected energy production change.
- 11.3.3.3.3 Seller's updated Forecast must contain the following information:
  - 11.3.3.3.3.1 The heginning date and time of the event resulting in a change in the availability of the Generating Facility and expected hourly energy production in MWh/h;
  - 11.3.3.3.3.2 The expected ending date and time of the event;
  - 11.3.3.3.3.3 The expected energy production or available generation capacity, as applicable, in MWh; and
  - 11.3.3.3.4 Any other information required by the CAISO as communicated to Seller by Buyer.

- 11.3.4 Buyer is responsible for all CAISO charges and is entitled to receive all CAISO revenues.
- 11.3.5 Except as set forth in this Section 11.3, there shall be no modification of Seller's existing communication protocols and designated contacts with Buyer, if any, including any requirement to notify Buyer of Generating Facility parallel operation or separation from the electrical system.

#### 12 CAISO Tariff Compliance

- 12.1 CAISO Tariff Compliance for New PPAs
  - 12.1.1 As reflected in the applicable PPAs, all CHP Facilities subject to the CAISO Tariff shall comply with applicable CAISO Tariff provisions as determined by the CAISO no later than the time when the CHP Facility begins deliveries under any PPA entered into pursuant to this Settlement. CAISO approved revenue quality metering and telemetry shall be installed in compliance with CAISO requirements within six (6) months following execution of such contract, subject to any extension granted by the CAISO.
  - 12.1.2 The Parties acknowledge that the CAISO may condition, waive, extend or modify applicable conditions for interconnection, metering or other CAISO jurisdictional matters to the extent permitted by the CAISO Tariff provisions applicable to CHP Facilities. Pending installation of CAISO approved revenue quality metering and telemetry for a CHP Facility, the IOU shall provide the CAISO with any necessary revenue meter and telemetry data as requested by the CAISO.

#### 13 IOU Cost Recovery for CHP Program PPAs

- 13.1 Cost Allocation and Departing Load Charges
  - 13.1.1 The IOUs will enter into CHP PPAs for up to twelve (12) years. CPUC D.04-12-048, D.06-07-029 and D.08-09-012 limit the recovery of power procurement PPA costs to ten (10) years. This Settlement is conditioned on a CPUC Decision approving this Settlement which supersedes the provisions of D.06-07-029 and D.08-09-012 as follows: (1) the relevant costs (either "above market costs" or "net capacity costs," as appropriate) of this CHP Program can be recovered through Non-Bypassable Charges and (2) the same relevant costs of new PPAs entered into pursuant to the CHP PPA, subject to the restrictions herein.
  - 13.1.2 One of the two following methods for allocation of CHP procurement costs shall uniformly apply to all ESPs and CCAs. The choice of the two methods referenced in this section shall be determined by the CPUC on review of this Settlement. The same method shall be used for all three IOUs.
    - 13.1.2.1 If the CPUC determines that all ESPs and CCAs should procure CHP generation for their customers, then the relevant CPUC decisions (potentially including D.04-12-048, D.06-07-030 and/or D.08-09-012) shall be superseded to the extent necessary to authorize the allocation of all above-market costs associated with the IOU portion of the CHP Program on a vintaged basis to

future Direct Access (DA) and CCA customers and all future Departing Load Customers, except for Departing Load Customers served by CHP Facilities as provided in Public Utilities Code §218(b) (CHP Departing Load Customers), through the Cost Responsibility Surcharge (CRS). The calculation of the above-market costs of the CHP Program to be added to the CRS would follow the same methodology adopted in D.06-07-030.

- 13.1.2.2 If the CPUC determines that the IOUs should purchase CHP generation on behalf of DA and CCA customers, then the D.06-07-029 (and D.08-09-012 if necessary) shall be superseded to the extent necessary to authorize the IOUs to recover the net capacity costs associated with the CHP Program from all bundled service, DA and CCA customers and all Departing Load Customers except for CHP Departing Load Customers, on a non-bypassable basis. The net capacity costs of the CHP Program shall be defined as the total costs paid by the IOU under the CHP Program less the value of the energy and any ancillary services supplied to the IOU under the CHP Program. No energy auction shall be required to value such energy and ancillary services. In exchange for paying a share of the net costs of the CHP Program, the LSEs serving DA and CCA customers will receive a pro-rata share of the RA credits procured via the CHP Program.
- 13.1.3 The Parties agree that they will not advocate charging load served by CHP that meets the criteria of Public Utilities Code §218(b) and §216.6 for the costs of the CHP Program through any cost allocation mechanism.
- 13.1.4 The current Non-Bypassable Charges applicable to Customer Generation Departing Load (CGDL) as of December 31, 2009 include only the DWR Bond Charge, Nuclear Decommissioning and the Public Purpose Program charge. The Parties agree to oppose (or not support) any proposal before the CPUC not otherwise required by statute that would result in the application of "new generation" costs as defined in D.04-12-048 and D.08-09-012 or cost allocation mechanism (CAM) charges to CHP CGDL, or that would re-characterize the costs currently recovered through those mechanisms such that they would apply to CHP CGDL in the future. However, Parties reserve the right to argue that future programs not currently within the scope of "new generation" as defined in D.08-09-012<sup>5</sup> should be treated as Public Purpose Programs.
- 13.1.5 In recognition of the new cost recovery mechanisms contemplated by this Settlement, the Parties agree to advocate exclusion from the Competition Transition Charge (CTC) of any above-market costs associated with purchases of power from a CHP Facility via a PPA entered into pursuant to this Settlement. However, the above-market costs of QF procurement via Legacy PPAs may continue to be recovered through CTC for the life of those contracts.
- 13.2 IOUs' Energy Resources Recovery Accounts

<sup>&</sup>quot;New Generation" as defined in D.08-09-012 includes "generation from both fossil fueled and renewable resources contracted for or constructed by the investor-owned utilities subsequent to January 1, 2003." D.08-09-012, note 1.

13.2.1 The IOUs shall recover the cost of all payments made pursuant to PPAs and PPA
Amendments executed under this CHP Program in their respective Energy Resources
Recovery Accounts, subject only to the reasonable administration of these PPAs and
PPA Amendments.

#### 14 Settlement of Pending and Anticipated Litigation

#### 14.1 Retroactive Adjustment of SRAC Prices

14.1.1 If the CPUC adopts the Settlement and FERC terminates, subject to reinstatement, the PURPA must take obligation for QFs over 20 MW, the lOUs will withdraw with prejudice all SRAC retroactive price adjustment claims and challenges and such claims and challenges cannot be revived. As long as the FERC 210(m) suspension stays in place, the Parties may neither raise new claims nor seek retroactive adjustments of SRAC prices paid to QFs during the Settlement Term based on the grounds that SRAC prices violate PURPA avoided cost requirements or based on an argument that alternative pricing theories or methodologies would be preferable to the SRAC methodologies set forth in the Settlement. However, the IOUs may adjust payments to QFs based on other issues that arise in the administration of the PPAs, e.g., billing errors, metering errors, fraud.

#### 14.2 Released Claims

- 14.2.1 QFs over 20 MW will not have the right to new five (5)-year and ten (10)-year PPAs ordered in D. 07-09-040 during the Settlement Term.
- 14.2.2 CCC will withdraw its motion for an order implementing the prospective QF program PPA options adopted in D.07-09-040.
- 14.2.3 All Parties will waive all retroactive claims for energy and capacity adjustment for periods including, but not limited to:
  - 14.2.3.1 SRAC Remand Dispute (R.99-11-021) (December 2000 March, 31 2001),
  - 14.2.3.2 SCE's SRAC Update application (A. 08-11-001) (April 2004 July 31, 2009), and
  - 14.2.3.3 PG&E's and SDG&E's pending refund application (September 2007 July 31, 2009).
- 14.2.4 SCE/TURN will withdraw its Petition for Writ of Review of D.07-09-040 and D.08-07-048 at California Court of Appeal (Case B210398). Parties will withdraw cross claims in that case.
- 14.2.5 SCE, PG&E, and SDG&E will withdraw their application for rehearing of D.09-04-032.
- 14.2.6 SCE, PG&E, SDG&E, and TURN will withdraw their application for rehearing of D.09-04-034.
- 14.2.7 CAC, EPUC, and IEP will withdraw their applications for rehearing of D.09-04-032.

- 14.2.8 CAC and EPUC will withdraw their application for rehearing of D.09-04-034.
- 14.2.9 The CPUC shall close with prejudice its reconsideration of the Administrative Heat Rates as ordered in D.08-11-062.
- 14.2.10 Application or petitions regarding CHP Facilities on Transitional SO1 agreements or extensions will be withdrawn.
- 14.2.11 The IOUs will withdraw Advice Letters PG&E AL 3197-E, SDG&E AL 1958-E, and SCE AL 2200-E to implement new Standard Offer Contracts pursuant to D.07-09-040.
- 14.2.12 The IOUs shall withdraw the Petition for Modification of D.07-I2-052 (LTPP Decision) regarding QF capacity.

#### 14.3 Confidentiality

- 14.3.1 The Parties agree that they shall abide by the CPUC rules regarding Confidential Information as set forth in CPUC decisions and as modified by the CPUC, the courts or the legislature.
- 14.3.2 Parties reserve all rights to advocate for their respective positions before the CPUC, the courts or the legislature regarding the confidentiality of IOU procurement information.

#### 15 Federal Energy Regulatory Commission 210(m) Application

- 15.1 PURPA §210(m) Application at the FERC<sup>6</sup>
  - 15.1.1 The IOUs will submit a joint application under Section 210(m) of PURPA and 18 C.F.R. §§ 292.309 292.310 to FERC ("Joint Application") requesting a termination of the PURPA purchase requirement.
  - 15.1.2 The IOUs will provide the other parties to the Settlement including QFs who are members of the CHP Parties' organizations ("Other Parties to the Settlement") an opportunity to review and provide comment to the IOUs on the Joint Application prior to it being filed and the IOUs will use good faith in considering and incorporating comments consistent with the Settlement.
  - 15.1.3 The Joint Application shall be filed under 210(m)(1)(C). The Joint Application will reference the following four components to demonstrate that the application meets the statutory requirements of Section 210(m)(1)(C): MRTU, RA Capacity, RPS Program, and CHP Program Settlement.
  - 15.1.4 The Other Parties to the Settlement will have the right to intervene and file comments on the Joint Application, but will not protest nor otherwise oppose the termination of the PURPA purchase obligation, consistent with Section 15.1.3 above. Nothing herein prohibits the IOUs from filing a reply to comments consistent with Section 15.1.3.

Order 688, New PURPA Section 210(m) Regulations Applicable to Small Power Production and Cogeneration Facilities, 117 FERC ¶ 61,078 (2006).

- 15.1.5 The Other Parties to the Settlement will provide the IOUs an opportunity to review and provide comment to the Other Parties to the Settlement on their respective comment on the Joint Application prior to the comment being filed and the Other Parties will use good faith in considering and incorporating comments consistent with the Settlement.
- 15.1.6 The Joint Application will not be filed until after the CPUC issues a decision approving or disapproving the Settlement.
- 15.1.7 Subject to the situations described in Sections 15.1.9, 15.1.10, and 15.1.12 below, the IOUs and the Other Parties to the Settlement (collectively, the IOUs and the Other Parties to the Settlement, "the Parties") agree that all advocacy in all forums will be consistent with the Settlement, the Joint Application, and Section 15.1.3.
- 15.1.8 The Parties reserve their rights to seek rehearing and appeal a FERC decision granting or denying the Joint Application.
- 15.1.9 If a non-settling party protests or otherwise opposes the Joint Application, nothing herein prohibits the Parties from replying. In addition, the Parties shall: (1) meet and confer within three (3) days to determine if they can reach agreement on advocacy positions consistent with the Settlement, the Joint Application and Section 15.1.3; and (2) make a good faith effort to agree on bases for advocacy within seven (7) days consistent with the Settlement, the Joint Application, and Section 15.1.3; thereafter, including requesting that FERC convene a Settlement Conference under Rule 601. If the Parties are unsuccessful in reaching resolution of advocacy positions as contemplated by this section, each Party shall have the right to file in support or opposition to the protest consistent with the Settlement, the Joint Application and Section 15.1.3.
- 15.1.10 The Other Parties to the Settlement do not waive the right to file at FERC for a reinstatement of a particular IOU's PURPA purchase obligation if the IOU breaches its obligations under the Settlement or the CHP Program adopted in the Settlement is not successfully implemented, based upon the IOU's failure to meet the targets established by the CPUC pursuant to the Settlement, without justification as provided for in the Settlement. The Other Parties to the Settlement cannot file for a reinstatement until the CPUC or the Energy Division Director makes a determination in writing that the IOU's failure to meet the targets was not justified. The Other Parties to the Settlement shall not file for a reinstatement of a particular IOU's PURPA purchase obligation earlier than the announcement of the results of the second RFO for that IOU. The IOUs retain the right to argue that termination of the PURPA purchase requirement should be sustained and/or that relief should be solely on the basis of Section 210(m)(1)(A) in response to a reinstatement filing.
- 15.1.11 If at any time, after the Section 201(m)(1) termination has been granted by FERC, a filing is made at FERC, or any other applicable authority, by a third party seeking relief from the Section 210(m)(1) termination or reinstatement of the IOU PURPA purchase obligation, or any successor provision, the Parties shall: (1) meet and confer within five (5) days to determine if they can reach agreement on advocacy positions; and (2) make a good faith effort to agree on bases for advocacy within five (5) days thereafter that will preserve the grounds for termination set forth in Section 15.1.3, above. If the Parties are unsuccessful in reaching resolution of advocacy positions as contemplated

- by this section, each party shall have the right to file in support or opposition to the filing seeking relief from the termination.
- 15.1.12 .If the Joint Application is denied, the IOUs reserve the right to file a new application that does not comply with the procedures set forth above.
- 15.2 FERC Reinstatement of PURPA Obligation
  - 15.2.1 If FERC reinstates the 10Us' obligation to purchase pursuant to Section 210(m) then the following applies:
    - 15.2.1.1 SRAC pricing stays in place until changed by the CPUC.
    - 15.2.1.2 The amendments signed by Legacy QFs stay in place for the term of the amendments without change, including SRAC contained in such amendments.
    - 15.2.1.3 Parties may advocate for changes to SRAC pricing.
    - 15.2.1.4 Prior claims for and challenges to retroactive price adjustments cannot be revived, but Parties may advocate for retroactive adjustments to SRAC pricing in accordance with existing Court and CPUC decisions unless and until modified by the Courts or CPUC and may petition or otherwise advocate for changes in rules and decisions on retroactive adjustments.
    - 15.2.1.5 If the CPUC sets SRAC at a market-based price, excluding certain administrative adjustments such as O&M adders, there will be no retroactive pricing for the period such market-based SRAC formula is in effect. However, the IOUs may adjust payments to QFs based on other issues that arise in the administration of the PPAs, e.g., billing errors, metering errors, fraud.
    - 15.2.1.6 If the CPUC sets SRAC at a price based in whole or in part on administrative or non-market based components, excluding certain administrative adjustments such as O&M adders, rather than a market-based price SRAC, Parties may advocate for a return to a market-based SRAC price and may advocate for the retroactive application of an adopted market-based SRAC pricing change in accordance with existing Court and CPUC decisions unless and until modified by the Courts or CPUC.
    - 15.2.1.7 Obligations to conduct additional CHP RFOs or conduct alternative CHP Procurement Processes under the Settlement shall be suspended, including the MW Targets and GHG Emissions Reduction Targets; provided the CPUC may on grounds other than the Settlement direct the procurement of CHP resources. CHP PPAs executed prior to the suspension remain in full force and effect. Any procurement targets to be established by the CPUC in the LTPP remain in place unless and until modified by the CPUC in a subsequent proceeding.
    - 15.2.1.8 Designation of Legally Enforceable Obligation under Section 210(m). The Parties agree that any existing CHP PPA, any existing extension agreements, and any Standard Offer PPA or Pro-Forma PPA executed pursuant to this Settlement will constitute a legally-enforceable obligation as provided in Section 210(m).

#### 16 Conditions Precedent and Subsequent to Settlement Effective Date

- 16.1 Approval of Settlement by CPUC
  - 16.1.1 This Settlement shall be submitted and approved in all open QF dockets. This Settlement resolves all open and pending issues in all the following open dockets: R.04-04-003, R.04-04-025, A.08-11-001, R.99-11-022, R.06-02-013.
- 16.2 Conditions Precedent to Effectiveness of the Settlement
  - 16.2.1 Final and non-appealable FERC approval of the IOUs' application to terminate their PURPA purchase obligation, subject to reinstatement in accordance with Section 210(m);
  - 16.2.2 Final and non-appealable approval of the Settlement by the CPUC as submitted for approval without revisions unacceptable to any Party or in an alternative form that is acceptable to all Parties;
  - 16.2.3 As part of the Settlement approval process, CARB support, in written form, for the Settlement. Examples of such support can be, but are not limited to, pleadings, comments to the CPUC or a letter of support. CARB support must be provided to the CPUC no later than CPUC approval of this Settlement;
  - 16.2.4 The CPUC Decision adopting the Settlement will supersede certain portions of existing CPUC decisions as occurs with any subsequent CPUC decision or order;
  - 16.2.5 D.06-07-029 and D.08-09-012 are superseded as follows: (1) the relevant costs (either "above market costs" or "net capacity costs" as appropriate) of this CHP Program can be recovered through Non-Bypassable Charges consistent with Section 13 herein; and (2) the same relevant costs of new PPAs entered into pursuant to the CHP Program can be recovered through Non-Bypassable Charges for up to twelve (12) years consistent with Section 13 herein.
  - 16.2.6 The Procurement obligations in this Settlement and under the RPS Program supersede and replace the QF MWs in D.07-12-052.
  - 16.2.7 All Parties will execute and be bound by the Settlement on the Settlement Effective Date and shall support and defend this Settlement in all forums, including but not limited to the CPUC, FERC, CEC, CARB and the federal and state Legislature. CCC, CAC and EPUC will sign on behalf of their members and an appendix to the Settlement will eontain the names of all of the current members of each of the organizations and each such member will be bound by the Settlement. IEP members and other QFs will have the option to sign a QF PPA amendment, which will include agreement to be bound by the Settlement. An amendment or execution of the Settlement will not be required to bind the members of CCC, CAC and EPUC. CCC, CAC and EPUC will represent that it is authorized to represent each listed member and that it has provided a copy of the Settlement to each of its members.
  - 16.2.8 To the extent the Generating Facility has Green Attributes associated with the Related Product (as Green Attributes and Related Product are defined in each of the form PPAs

attached to this Settlement), such Green Attributes shall be counted or credited toward the purchasing IOU's RPS Program or any successor program.

- 16.3 Conditions Subsequent to Effectiveness of Settlement
  - 16.3.1 If CARB subsequently adopts regulations directly imposing a MW Target or GHG Emissions Reduction Target that differs from the Settlement for the Second Program Period, the IOUs' obligations to purchase from CHP to meet the MW Target or GHG Emissions Reduction Target will remain in place until such time as the CPUC is able to consider such change in an LTPP or other pertinent proceeding.

#### 17 Glossary of Defined Terms

AB 1613 Feed-In Tariff: As defined in Section 4.4.1.

Actual HR: The Heat Rate that must be used in accordance with and subject to the terms set forth in Section 10.2.2.2, which Heat Rate Buyer shall calculate, on the date of the commencement of the First Compliance Period, using the following formula:

Actual HR = The average of the Daily  $HR_n$  for each delivery or flow date in the two (2) year period immediately preceding the commencement of the First Compliance Period

Where:

Daily  $HR_n = [EP_n - VOM_n] / [GP_n + GT_n]$ 

Where:

- EP<sub>n</sub> = The average of the Day Ahead hourly electric energy prices, as determined by the Integrated Forward Market (as defined in the CAISO Tariff) for (i) SP15 Existing Zone Generation Trading Hub (formerly known as SP15), or its successor, if Buyer is SCE or SDG&E, and (ii) NP15 Existing Zone Generation Trading Hub (formerly known as NP15), or its successor, if Buyer is PG&E;
- $VOM_n$  = Calendar month avoided variable O&M for the applicable month (\$/kWh), per D.07-09-040 and CPUC Resolution E-4246 (as such Resolution may be modified from time to time by the CPUC);
- GP<sub>n</sub>= The applicable daily gas price index, which is (i) Platt's Gas Daily (currently SoCalGas gas indices) if Buyer is SCE or SDG&E, or (ii) Platt's Gas Daily (currently SoCalGas and PG&E Malin gas indices), if Buyer is PG&E; and
- $GT_n$  = The gas transportation rate for the applicable month, per CPUC Resolution E-4246 (as such Resolution may be modified from time to time).
- Additional GHG Documentation: Documentation necessary to allocate Free Allowances to energy delivered by Seller to Buyer, which documentation consists of the following, in each case for the time-period to which the Free Allowances are to apply: (a) the total amount of GHG emissions attributable to the Generating Facility, the electrical energy used to serve the Site Host Load, the Useful Thermal Energy Output of the Generating Facility; and the energy delivered to Buyer; (b) the Useful Thermal Energy Output of the Generating Facility; (c) the total electrical energy produced by the Generating Facility, the electrical energy used to the serve the Site Host Load, and the energy delivered to Buyer; and (d) total fuel usage of the Generating Facility.
- Allowance: A limited tradable authorization (whether in the form of a credit, allowance or other similar right), allocated to, issued to or purchased by, Seller, the Site Host or a Related Entity of Seller, with respect to the Generating Facility, to emit one MT of GHG, in accordance with a cap-and-trade program in California for the regulation of GHG, as

established by CARB (and/or by a different Governmental Authority pursuant to federal or state legislation), and as applied to the GHG emitted by the Generating Facility.

<u>AMW</u>: For purposes of this Settlement, the average MW deliveries from the Optional As-Available PPAs as determined in accordance with Section 4.6.2.10.3.

AMW CAP: There is a cap on energy deliveries from all of the Optional As-Available PPAs measured in AMW. The AMW Cap shall be: PG&E: 75 AMW, SCE: 75 AMW, SDG&E: 10, as defined in Section 4.6.2.10.2.

Annual GHG Reports: As defined in Section 10.3.1.

Applicable NDA: As defined in Section 9.1.4.

Applicable PPA: Any power purchase agreement between Buyer and Seller that requires Buyer to calculate the energy payment due to Seller under such agreement in whole or in part based on SRAC.

Availability Incentive Payments: As defined in the CAISO Tariff.

Availability Standards: As defined in the CAISO Tariff.

Baseline: Specific to GHG Accounting, as defined in Section 7.3.1.3.

Bottoming Cycle CHP: A cogeneration facility in which the energy input to the system is first applied to a useful thermal energy application or process, and at least some of the reject heat emerging from the application or process is then used for power production, and as otherwise provided in 18 CFR Section 292, et seq.

Burner Tip Gas Price: As determined in CPUC Decision 07-09-040 and CPUC Resolution E-4246 (as such Resolution may be modified from time to time by the CPUC).

Business Day: Any day except a Saturday, Sunday, the Friday after the United States
Thanksgiving holiday, or a Federal Reserve Bank holiday that begins at 8:00 a.m. and
ends at 5:00 p.m. local time for the Party sending a notice or payment or performing a
specified action.

<u>Buyer</u>: The party responsible for purchasing electric energy, capacity or other power products delivered by Seller in a PPA.

CAISO: The California Independent System Operator Corporation or successor entity.

<u>CARB</u>: The California Air Resources Board, or any successor entity.

CARB Mandatory GHG Emissions Annual Report: Reporting of greenhouse gases by major sources is required by the California Global Warming Solutions Act. The Air Resources Board approved a mandatory reporting regulation in December 2007, which became effective January 2009. See Section 10.3.1.1.

CARB CHP Recommended Reduction Measure or RRM: As referenced on pages 43-44 of the "Approved Scoping Plan" adopted by the CARB at its December 11, 2008 Board meeting. http://www.arb.ca.gov/cc/scopingplan/document/adopted scoping plan.pdf

- CARB Scoping Plan: the "Approved Scoping Plan" adopted by the CARB at its December 11, 2008 Board meeting.

  <a href="http://www.arb.ca.gov/cc/scopingplan/document/adopted\_scoping\_plan.pdf">http://www.arb.ca.gov/cc/scopingplan/document/adopted\_scoping\_plan.pdf</a>
- CAISO Tariff: The CAISO Operating Agreement and Tariff, including the rules, protocols, procedures and standards attached thereto, as the same may be amended or modified from time to time and approved by the FERC, or any successor entity.
- CAISO-Approved Quantity: The total quantity of electric energy that Buyer Schedules with the CAISO and the CAISO approves in its final schedule which is published in accordance with the CAISO Tariff.
- CCA or Community Choice Aggregator: Any city, county, or city and county, or group of cities, counties, or cities and counties, whose governing board or boards elect to combine the loads of their residents, businesses, and municipal facilities in a community wide electricity buyers' program, as specified in Public Utilities Code §331.5.
- CEC: California Energy Commission.
- <u>CHP</u>: Combined Heat and Power system or cogeneration means the sequential use of energy for the production of electrical and useful thermal energy (Public Utilities Code §216.6). As defined in Section 1.1.1.
- CHP Auditor: As provided in Section 9.
- CHP Facility or CHP Facilities: A facility that meets the federal definition of a qualifying cogeneration facility under 18 C.F.R. §292.205.
- CHP Party or CHP Parties: Any of the following associations, or an individual member of or group of individual members of such associations California Cogeneration Council (CCC), Cogeneration Association of California (CAC), Energy Producers and Users Coalition (EPUC); the Independent Energy Producers Association (IEP); and Sellers that execute the Legacy PPA Amendment.
- CHP Program Semi-Annual Report: As defined in Section 8.1.1.
- CHP Procurement Processes: Procurement processes for CHP resources by the IOUs that count toward the MW and/or GHG Emissions Reduction Targets as specified in the Settlement. These include the CHP RFO PPAs, bilaterally negotiated and executed CHP PPAs, the AB 1613 Feed-In Tariff, QF Standard Offer CHP PPAs for eligible CHP Facilities pursuant to PURPA, Optional As Available PPAs for eligible CHP Facilities, certain IOU-owned CHP, existing CHP that converts to a Utility Prescheduled Facility, and new behind the meter CHP.
- <u>CHP OF Compliance Monitoring Program</u>: The existing CPUC QF Compliance Monitoring Program, established by Decision 91-05-007.
- CHP RFOs: As defined in Section 4.2.1.
- CHP RFO Pro-Forma PPA: As defined in Section 4.2.6.

<u>Capacity Demonstration Test</u>: Buyer shall conduct a Capacity Demonstration Test of the Generating Facility under the CHP PPA to determine the MW to count towards the MW Targets. The Capacity Demonstration Test shall be conducted as defined in the attached PPAs.

Confidential Information: All oral or written communications exchanged between the Parties on or after the Settlement Effective Date relating to the implementation of the CHP PPA, including information related to Seller's compliance with operating and efficiency standards applicable to a qualifying cogeneration facility or fuel use standards applicable to a qualifying small power production facility. Confidential Information does not include (i) information which is in the public domain as of the Settlement Effective Date or which comes into the public domain after the Settlement Effective Date from a source other than from the other Party, (ii) information which either Party can demonstrate in writing was already known to such Party on a non-confidential basis before the Settlement Effective Date, (iii) information which comes to a Party from a bona fide third-party source not under an obligation of confidentiality, or (iv) information which is independently developed by a Party without use of or reference to Confidential Information or information containing Confidential Information.

CPUC: the California Public Utilities Commission.

CPUC Jurisdictional Entities: the IOUs, ESPs and CCAs.

D.: A CPUC Decision.

DA: Direct Access. As defined in Section 6.3.1.

<u>Day-Ahead</u>: As defined and may be modified in the CAISO Tariff.

<u>Day-Ahead Market</u>: The forward market for energy and ancillary services to be supplied during the settlement period of a particular Trading Day that is conducted by the CAISO, and other Scheduling Coordinators. This market closes with the CAISO's acceptance of the final Day-Ahead Sehedule.

<u>Day-Ahead Schedule</u>: A schedule prepared by a Scheduling Coordinator or the CAISO before the beginning of a Trading Day. This schedule indicates the levels of generation and demand scheduled for each settlement period of that Trading Day.

<u>Departing Load Customers</u>: Includes CGDL, TMDL and NMDL customers as defined in SCE's existing DL-NBC, CGDL-CRS, TMDL and NMDL tariff schedules; as CGDL, TMDL, and NMDL in PG&E's existing E-DCG, E-NMDL and E-TMDL tariff schedules; and in SDG&E's existing E-DEPART tariff schedules.

Designated Capacity Product: As defined in Section 3.4.1.2.

<u>Direct Aceess Customers</u>: Customers located within the IOU Service Territory of an IOU who purchase electricity from an ESP.

<u>Double Benchmark</u>: The Double Benchmark measures the amount of GHG emissions that otherwise would exist if the CHP Facility did not exist. The Double Benchmark, which may be later modified pursuant to this Settlement, contains the following assumptions:

- (a) the heat rate for the electricity generated is 8,300 BTU/kWh at the busbar and excluding line losses; and (2) the thermal efficiency of a standard boiler is 80%.
- Efficiency Performance Deficiency: Failure to meet annual efficiency requirements under the terms of the PPA, as defined in Section 4.2.9.
- Efficiency Performance Requirement: A contractual requirement to meet certain levels of efficiency as defined in the PPA or AB 1613 Feed-In Tariff.
- Emergency Condition: As set forth in the Transmission Provider's LGIA, SGIA or other distribution-level FERC-jurisdictional interconnection agreement with Seller, as applicable; *provided, however*, that if Seller interconnects pursuant to Tariff Rule 21, "Emergency Condition" means "Emergency", as defined in such Tariff Rule 21.
- Emissions Performance Standard or EPS: The Emissions Performance Standard adopted by the CPUC in D.07-01-039 and as it may be revised in subsequent decisions.
- ESP or Electric Service Provider: An entity that is licensed by the CPUC to provide electric power service to Direct Access Customers under Public Utilities Code §§218.3 and 394, or subsequent statutes.
- Existing CHP Facility(ies): An Existing CHP Facility is one that was operational before the Settlement Effective Date.
- Expanded CHP Facilities: Means (i) an existing topping-cycle CHP Facility that, on or after the Settlement Effective Date, has added at least one new combustion turbine and increased the Power Rating of the Generating Facility by not less than 90% of the Power Rating of the largest existing combustion turbine at the Generating Facility, or (ii) an existing bottoming-cycle CHP Facility that has increased its total Power Rating by at least 30% as compared to the Power Rating before such expansion.
- <u>EWG or Exempt Wholesale Generator</u>: An unregulated power generator that is allowed to sell wholesale power as an independent energy producer, and is exempt from the Public Utility Holding Company Act of 1935.
- FERC: The Federal Energy Regulatory Commission, or successor agency.
- <u>First Compliance Period</u>: The first period of time for compliance with a cap-and-trade program in California for the regulation of GHG, as established by CARB (and/or by a different Governmental Authority pursuant to federal or state legislation). There will be no more than a single First Compliance Period.
- Floor Test Term: From the date the First Compliance Period commences for a period of three (3) years.
- Free Allowance: Any Allowance freely allocated to Seller or the Generating Facility by CARB or an authorized Governmental Authority (or any entity authorized by such Governmental Authority).
- Free Allowance Notice: The notice, delivered by Seller to Buyer in accordance with the Applicable PPA, that sets forth the aggregate quantity of Free Allowances received by

Seller during the applicable time-period and sets forth the allocation of such Free Allowances in accordance with the following:

- (i) The allocation of Free Allowances by CARB (or any other Governmental Authority) to the energy generated by the Generating Facility and delivered to Buyer during the applicable time-period; or
- (ii) If CARB (or any other Governmental Authority) does not allocate Free Allowances received by Seller as described in subsection (i) above, then Seller shall set forth in the Free Allowance Notice the quantity of Free Allowances allocated to the energy generated by the Generating Facility and delivered to Buyer during the applicable time-period  $(FA_d)$  utilizing the following formula:

$$FA_d = FA_t * [G_e/(G_e + G_t)] * [E_d/(E_{sh} + E_d)]$$

Where:

- $FA_t$  = Total number of Free Allowances received by Seller with respect to the Generating Facility for the applicable time-period;
- $G_e$  (in MTs) = Emissions of GHG attributed to the total amount of electric energy produced by the Generating Facility for the applicable time-period (ealculated in accordance with the formula set forth in Section 95112 of the California Code of Regulations, or any successor thereto, which calculation must be set forth in the Free Allowance Notice);
- $G_t$  (in MTs) = Emissions of GHG attributed to the Useful Thermal Energy Output produced by the Generating Facility for the applicable time-period (calculated in accordance with the formula set forth in Section 95112 of the California Code of Regulations, or any successor thereto, which calculation must be set forth in the Free Allowance Notice);
- $E_d$  (in kWh) = Energy generated by the Generating Facility and delivered to Buyer for the applicable time-period; and
- $E_{sh}$  (in kWh) = Electric energy generated by the Generating Facility and used to serve the Site Host Load for the applicable time-period; or
- (iii) If the CARB (or any other Governmental Authority) does not allocate the Free Allowances received by Seller, as described in (i) above, and there is no available formula in any applicable rule or regulation for the calculation of  $G_e$  and  $G_t$ , as described in (ii) above, then Seller shall include in the Free Allowance Notice the total amount of emissions of GHG attributed to the electric energy period ( $G_e$ , in MTs) and the Useful Thermal Energy Output ( $G_t$ , in MTs) produced by the Generating Facility for the applicable time-period based on the two following formulas:
  - G<sub>e</sub> = G \* (Useful Power Output / (Useful Power Output + Useful Thermal Energy Output))

 $G_t = G * (Useful Thermal Energy Output / (Useful Power Output + Useful Thermal Energy Output))$ 

Where:

- G (in MTs) = Total emissions of GHG produced by the Generating Facility for the applicable time-period;
- Useful Power Output (in MMBtu) = As defined in 18 CFR §292.202(g), or any successor thereto;
- Useful Thermal Energy Output (in MMBtu) = As defined in 18 CFR §292.202(h), or any successor thereto;
- Upon determining  $G_e$  and  $G_t$  in subsection (iii) above, Seller shall then calculate for and provide the quantity of Free Allowances attributed to energy generated by the Generating Facility and delivered to Buyer for the applicable time-period  $(FA_d)$  using the formula set forth in subsection (ii) of this definition.

Forced Outage: As set forth in the CAISO Tariff.

- <u>Forecast</u>: The hourly forecast of (i) the total electric energy production of the Generating Facility (in MWh) when the Generating Facility is not PIRP-eligible, or (ii) the available total generation capacity of the Generating Facility (in MW) when the Generating Facility is PIRP-eligible, in each case net of the Site Host Load and Station Use.
- Generating Facility: Seller's electric energy production facility set forth in the PPA.
- GHG Auction: Any auction or other sale-by-bid event applicable to California and by an authorized Governmental Authority (or any entity authorized by such Governmental Authority) for the sale of Allowances.
- GHG Charges: Means all taxes, charges or fees, assessed with the implementation and regulation of Greenhouse Gas emissions with respect to the Generating Facility imposed by any Governmental Authority, such as the CARB's AB 32 Cost of Implementation Fee (as defined in Title 17 C.C.R. §95200). For example, if the charges are assessed on but not included in fuel consumption or gas costs, the Applicable Heat Rate or Burner Tip Gas Price will be used to derive the dollars per kilowatt-hour charge. On January 1, 2015 or the commencement of the First Compliance Period, the GHG Charges will equal zero in the above formula.
- GHG Compliance Costs: The cost of Allowances, as determined in accordance with Section 10.2.
- GHG Credit: A positive GHG Emissions Reduction from a Generating Facility as defined in Section 7 of the Settlement.
- GHG Debit: An increase in GHG emissions from a Generating Facility as defined in Section 7 of the Settlement.
- GHG Emissions Reduction Target: The procurement targets defined in Section 6.

- GHG Floor Test: As defined in Section 10,2.2.
- Governmental Authority: Any governmental authority responsible for the regulation of GHG in California, including (i) any federal, state, local, municipal or other governmental authority, (ii) any governmental, regulatory or administrative agency, commission, lawfully exercising or entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, or (iii) any court or governmental tribunal.
- Greenhouse Gas or GHG: Emissions released into the atmosphere of carbon dioxide (CO<sub>2</sub>), nitrous oxide (N<sub>2</sub>O) and methane (CH<sub>4</sub>), which are produced as the result of combustion or transport of fossil fuels. Other greenhouse gases may include hydrofluorocarbons (HFCs), perfluorocarbons (PFCs) and sulfur hexafluoride (SF<sub>6</sub>), which are generated in a variety of industrial processes. Greenhouse gases may be defined or expressed in terms of a MT of CO<sub>2</sub>-equivalent, in order to allow comparison between the different effects of gases on the environment; provided, however, that the definition of the term 'Greenhouse Gas', as set forth in the immediately preceding sentence, shall be deemed revised to include any update or other change to such term by CARB and/or any other Governmental Authority.
- Heat Rate: For purposes of this Settlement, the value obtained, in BTU per kWh, when the fuel input, on a Higher Heating Value basis, in BTU is divided by generation, net of Station Use, in kWh.
- <u>Higher Heating Value or HHV</u>: The high or gross heat content of the fuel with the heat of vaporization included; the water vapor is assumed to be in a liquid state.
- Hour-Ahead Scheduling Deadline: Thirty (30) minutes before the deadline established by the CAISO for the submission of schedules for the applicable hour.
- IE or Independent Evaluator: The duties of an IE are generally described in CPUC D.07-12-052 at pp. 136-142.
- <u>Initial Program Period</u>: The period commencing on the Settlement Effective Date and concluding forty-eight (48) months thereafter.
- Initial Forecast Month: The first calendar month for which Seller provides to Buyer a 30-day
  Forecast pursuant to Section 11.3, which must be the first calendar month commencing
  no earlier than 30 days after Settlement Effective Date and no later than 60 days after
  such date.
- IOU: An investor owned-utility, including PG&E, SCE and SDG&E.
- IOUs: PG&E, SCE and SDG&E collectively.
- Joint Application: As defined in Section 15.1.1.
- <u>July 2010 Semi-Annual Report</u>: Each IOU's July 2010 Cogeneration and Small Power Production Semi-Annual Reports, as defined in Section 6.2.1.1.

<u>Legacy PPA(s)</u>: All Existing QF or CHP PPAs, including extensions of such PPAs, in force and effect on the Settlement Effective Date with any lOU that has not expired by its terms and conditions, excluding PPAs entered into pursuant to the RPS Program.

<u>Legacy PPA Amendment</u>: An amendment to a Legacy PPA which is attached to this Term Sheet as Exhibits 1, 2, and 3.

LGIA (Large Generator Interconnection Agreement or Standard Large Generator Interconnection Agreement): As set forth in the CAISO Tariff.

<u>Legacy PPA C1 Amendment:</u> An amendment to a Legacy PPA which allows the IOU to schedule operations of the CHP Facility or Utility Prescheduled Facility.

Locational Marginal Price: As defined and may be modified in the CAISO Tariff.

<u>LTPP</u>: The CPUC's Long-Term Procurement Plan proceeding for the IOUs, or its successor proceeding.

Market: The MIF calculation adopted in D.07-09-040 and Resolution E-4246 with 100% Market Heat Rate and 0% Administrative Heat Rate. Market Heat Rates shall be defined as provided in Resolution E-4246. The Parties agree to rely on the 12-month forward indices adopted by the CPUC in Resolution E-4246, unless the direction to use such indices is modified by the CPUC.

Market Disruption Event: With respect to any MHR Source, any of the following events: (i) the permanent discontinuation or material suspension of trading in the exchange or in the market specified for determining a Market Heat Rate; (ii) the temporary or permanent discontinuance or unavailability of the MHR Source; or (iii) the temporary or permanent closing of any exchange specified for determining a Market Heat Rate. For purposes of this definition, "temporary" means five (5) or more continuous Trading Days.

Market Heat Rate: The 12-month forward market heat rate, calculated for each calendar pricing month utilizing the methodology set forth in CPUC D.07-09-040 and CPUC Resolution E-4246 (as such Resolution may be modified from time to time by the CPUC) for (i) SP15 Existing Zone Generation Trading Hub (formerly known as SP15), or its successor, if Buyer is SCE or SDG&E, and (ii) NP15 Existing Zone Generation Trading Hub (formerly known as NP15), or its successor, if Buyer is PG&E.

Megawatt (MW): One thousand kilowatts (1,000 kW) or one million (1,000,000) watts.

Megawatt Hour (MWh): One thousand kilowatt-hours.

MRTU: The Market Redesign and Technology Upgrade implemented by CAISO on April 1, 2009.

MHR Source: The relevant publications used to determine the Market Heat Rate.

MMT: Million Metric Tons.

MT(s): Metric Ton(s)

MW Target(s): As provided in Section 5.

NDA: Non Disclosure Agreement as defined in Section 9.1.4 and attached as Exhibit 8.

Net MW Targets: MW Targets in each RFO in the Initial Program Period.

New MW Target(s): As defined in Section 5.1.4.1.

New CHP Facility: A CHP Facility that became operational after the Settlement Effective Date.

Non-Availability Charges: As defined in the CAISO Tariff.

Non-Bypassable Charge: A charge to Departing Load Customers to recover the costs of PPAs entered into by an IOU for their benefit while they were IOU bundled customers.

Non-IOU LSE: A load serving entity other than PG&E, SCE, and SDG&E. See Section 6.3.1.

Optional As-Available Program: Available for generators that deliver less than 20 MW, on average to the Buyer, as provided in Section 4.6.

Optional As-Available PPA: A PPA option for CHP generators that deliver less than 20 MW, on average, to the Buyer, as provided in Section 4.6, and the form of which is attached to this Settlement as Exhibit 7.

Other Parties to the Settlement: As defined in Section 15.1.2.

<u>PIRP (Participating Intermittent Resource Program)</u>: The CAISO's intermittent resource program initially established pursuant to Amendment 42 of the CAISO Tariff in FERC Docket ER02-922-000, or any successor program that Buyer determines accomplishes a similar purpose.

PG&E: Pacific Gas and Electric Company.

<u>Party</u>: A single party to the Settlement from among CCC, CAC, EPUC, IEP, PG&E, SCE, SDG&E, TURN, and DRA.

Parties: CCC, CAC, EPUC, IEP, PG&E, SCE, SDG&E, TURN, DRA.

PNode: As defined in the CAISO Tariff.

<u>POUs</u>: Publicly-owned utilities including municipal utilities (utilities owned by branches of local government) and/or co-ops (utilities owned cooperatively by customers).

<u>Power Rating</u>: The electrical power output value indicated on the generating equipment nameplate.

PPA: A power purchase agreement between a Seller and Buyer.

<u>PRG</u>: The group of non-market participants established in D.02-08-071 as an advisory group to review and assess the details of the IOUs' overall procurement strategy, RFOs, specific proposed procurement contracts and other procurement processes.

- PURPA: The Public Utility Regulatory Policies Act of 1978, as amended, 16 U.S.C. §§823a, 2601, et seq.
- **OCRs**: IOU Quarterly Compliance Reports submitted to the CPUC.
- Qualifying Facility or QF: An electric energy generating facility that complies with the qualifying facility definition established by PURPA and any FERC rules as amended from time to time (18 Code of Federal Regulations Part 292, Section 292.203 et seq.) implementing PURPA and has filed with FERC (i) an application for FERC certification, pursuant to 18 Code of Federal Regulations Part 292, Section 292.207(b)(1), which FERC has granted, or (ii) a notice of self-certification pursuant to 18 Code of Federal Regulations Part 292, Section 292.207(a).
- QF PPA: A PPA with standard terms and conditions for a QF with a nameplate capacity of 20 MWs or less,
- Real-Time Forced Outage: A Forced Outage which occurs only after 5:00 p.m. California time on the day before a Trading Day impacted by such Forced Outage.
- Reference Market-Maker: A leading dealer in the energy market that is not a Related Entity of either Party and that is selected by a Party in good faith among dealers of the highest credit standing which satisfy all the criteria that such Party applies generally at the time in deciding whether to offer or to make an extension of credit. Such dealer may be represented by a broker.
- Related Entity: With respect to either Party, any individual or entity that, directly or indirectly, through one or more intermediaries, controls, or is controlled by, or is under common control with such Party. For purposes of the Applicable PPA, "control" means the direct or indirect ownership of fifty percent (50%) or more of the outstanding capital stock or other equity interests having ordinary voting power.
- <u>Repowered CHP Facilities</u>: A CHP Facility that, on or after the Settlement Effective Date, has had its prime mover(s) replaced or refurbished, as follows:
  - If the CHP Facility contains combustion turbines, then each combustion turbine must be replaced with:
    - o A new combustion turbine that has been certified as new by the original manufacturer of the equipment <u>provided</u>, <u>however</u>, that the CHP Facility that has replaced its combustion turbines with a substantially identical engine (e.g., as is common during major overhauls of aeroderivative combustion turbines or as part of a spare engine program), does not qualify as a Repowered CHP Facility; or
    - A refurbished combustion turbine, so long as such refurbished combustion turbine has been certified by the entity that refurbished such combustion turbine (which may be the manufacturer) to achieve Heat Rate and total power output performance guarantees comparable to a new combustion turbine, prior to operational degradation; or
  - If the CHP Facility contains only steam turbines, then each steam turbine must be replaced with a refurbished steam turbine, which refurbishment must bave been accomplished with new or near-new condition parts, and which include (i) a

replacement of all stop and throttle control valves, seals, bearing, rotors, and turbine blades of each steam turbine, and (ii) a replacement or rebuilding of the stationary part of each steam turbine back to new condition, including seal system, lube oil system and all associated piping and auxiliary equipment.

In addition to the above requirement, (1) the repowering of the Generating Facility as described in this definition must be completed before the Term State Date, (2) Section 1.02(a) of the CHP PPAs must provide that the Generating Facility will be a Repowered CHP Facility on the Term Start Date, and (3) Seller must provide to Buyer a written certification, including all supporting data, from a qualified independent engineer, which certification must provide that the total useful life of the CHP Facility (including, as applicable, the combustion turbine(s), the steam turbine(s), the electrical generator(s) and the heat recovery steam generator) will operate for at least the Term of this Agreement, subject to industry standard maintenance practices.

- Resource Adequacy Benefit or RA Benefit: The resource adequacy benefits associated with the generating capability of the CHP Facility or Utility Prescheduled Facilities that can be used to meet the IOU's obligations under any CPUC resource adequacy Rulings.

  Resource Adequacy Benefits shall include any local, zonal or otherwise locational attributes associated with the CHP or Utility Prescheduled Facilities.
- RPS Program: The State of California Renewable Portfolio Standard Program, as codified at California Public Utilities Code §399.11, et seq.
- SCE: Southern California Edison Company.
- <u>Schedule</u>: The action of Scheduling Coordinator, or its designated representatives, of notifying, requesting, and confirming to the CAISO, the CAISO-Approved Quantity of electric energy.
- <u>Scheduling Coordinator</u>: Scheduling coordinators (SCs) submit schedules and bids and provide settlement-ready meter data to the CAISO.
- SDD: Scheduling and Delivery Deviation as set forth in Exhibit K of each of the QF PPA, Transition PPA, Optional As-Available PPA, and CHP RFO PPA.
- SDG&E: San Diego Gas & Electric Company.
- Second Program Period: The period of time commencing from the end of the Initial Program Period and concluding on December 31, 2020.
- Self-Generation: A generation facility dedicated to serving on site load or a particular retail customer, often located on the customer's premises. The facility may either be owned directly by the generator or owned by a third party with a contractual arrangement to provide electricity to meet some or all of the customer's load.
- <u>Self Generation Incentive Program (SGIP)</u>: SGIP provides rebates for qualifying distributed energy systems installed on the customer's side of the utility meter.
- <u>Seller</u>: The Party responsible for generating and delivering electric energy to Buyer in accordance with a PPA.

- <u>Service Territory</u>: The state, area or region served principally by a single IOU, subject to service provided by ESPs and CCAs to customers in such state, area or region.
- Settlement: That certain Settlement Agreement, dated October 8, 2010, filed by the Settling Parties in CPUC R.04-04-025, R.04-04-003, A.08-11-001, R.99-11-022, and R.06-02-013.
- Settlement Effective Date: The date that is the later of the two following events: (1) final and non-appealable CPUC approval of the Settlement in its entirety in a form that is agreeable to each Party; and (2) a final and non-appealable order from FERC approving the IOUs' Joint Application to terminate their PURPA purchase requirement under Section 210(m) of the Energy Policy Act of 2005, 117 FERC ¶61,078 (2006).
- Settlement Term: The period beginning on the Settlement Effective Date and ending December 31, 2020.
- SGIA (Small Generator Interconnection Agreement): The form of Interconnection Request (as defined in the CAISO Tariff) pertaining to a Small Generating Facility (as defined in the CAISO Tariff), which is attached to the CAISO Tariff as Appendix T.
- Site Host: The entity or entities purchasing or otherwise using the Site Host Load or Useful Thermal Energy Output from the Generating Facility.
- Site Host Load: The electric energy and capacity produced by or associated with the Generating Facility that serves electrical loads (other than Station Use) of Seller or one or more third parties conducted pursuant to California Public Utilities Code §218(b).
- SRAC or Short Run Avoided Cost: Short Run Avoided Cost, as set forth in Section 10.
- SRAC Commencement Date: As defined in Section 10.2.
- Standard Offer Contract or SOC: The QF PPA that was developed by the Parties in order to comply with D.07-09-040 and submitted for approval by the IOUs through Advice Letters PG&E AL 3197-E, SDG&E AL 1958-E, and SCE AL 2200-E, each as proposed to be modified by the CPUC in proposed final Resolution E-4242 dated September 10, 2009 and revised on October 20, 2009.
- <u>State CHP Program or CHP Program</u>: The program established in this Settlement for CPUCjurisdiction entities to procure from CHP Facilities and Utility Prescheduled Facilities.
- Station Use: The electric energy produced by the Generating Facility that is (i) used within the Generating Facility to power the lights, motors, control system and other electrical loads that are necessary for Operation, (ii) consumed within the Generating Facility's electricity energy distribution system as losses needed to deliver electricity to the Site Host Load, and (iii) consumed within the generator collection system as losses between the generator(s) and the high voltage side of the Generating Facility output transformer(s)."

System Emergency: As set forth in the CAISO Tariff.

- <u>Tariff Rule 21</u>: The interconnection standards of the Transmission Provider for distributed generation adopted by the CPUC in D.00-11-001 and D.00-12-037, as modified by the CPUC.
- <u>Tier 2 Advice Letter</u>: As defined in the CPUC's General Order 96-B or its successor.
- Tier 3 Advice Letter: As defined in the CPUC's General Order 96-B or its successor.
- <u>Topping Cycle CHP Facility</u>: A cogeneration facility in which the energy input to the facility is first used to produce useful power output, and at least some of the reject heat from the power production process is then used to provide useful thermal energy, and as otherwise provided in 18 CFR §292.205, et seq.
- <u>Trading Day</u>: The day in which Day-Ahead trading occurs in accordance with the Western Electricity Coordinating Council's Preschedule Calendar (as found on the WECC's website).
- <u>Transition PPA</u>: A short-term PPA between a CHP currently selling to an IOU under a Legacy PPA or extension thereof that begins on the expiration of the Legacy PPA and ends at the conclusion of the Transition Period, the form of which is attached to this Settlement as Exhibit 4.
- <u>Transition Period</u>: The period beginning on the Settlement Effective Date and ending on July 1, 2015.
- <u>Transmission Provider</u>: Any individual or entity responsible for the interconnection of the Generating Facility with the interconnecting utility's electrical system or the CAISO Controlled Grid or transmitting the Metered Energy (as defined in the PPAs attached to this Scttlement) on behalf of the Seller from the Generating Facility to the Delivery Point (as defined in the PPAs attached to this Settlement).
- <u>Useful Thermal Energy Output</u>: As defined in 18 CFR §292.202(h) and modified by the Energy Policy Act of 2005, or any successor thereto.
- <u>Utility Prescheduled Facility</u>: An Existing CHP Facility that has changed operations to convert to a utility controlled scheduled dispatchable generation facility, including but not limited to an EWG.
- Web Client: A Buyer provided Web based system or an email address designated by Buyer.

#### CERTIFICATE OF SERVICE BY ELECTRONIC MAIL OR U.S. MAIL

I, the undersigned, state that I am a citizen of the United States and am employed in the City and County of San Francisco; that I am over the age of eighteen (18) years and not a party to the within cause; and that my business address is Pacific Gas and Electric Company, Law Department B30A, 77 Beale Street, San Francisco, CA 94105.

I am readily familiar with the business practice of Pacific Gas and Electric Company for collection and processing of correspondence for mailing with the United States Postal Service. In the ordinary course of business, correspondence is deposited with the United States Postal Service the same day it is submitted for mailing.

On the 8th day of October 2010, I caused to be served a true copy of:

## JOINT MOTION FOR APPROVAL OF QUALIFYING FACILITY AND COMBINED HEAT AND POWER PROGRAM SETTLEMENT AGREEMENT

[XX] By Electronic Mail – serving the enclosed via e-mail transmission to each of the parties listed on the official service list for R.99-11-022, R.04-04-003, R.04-04-025, R.06-02-013 and A.08-11-001 with an e-mail address.

[XX] By U.S. Mail – by placing the enclosed for collection and mailing, in the course of ordinary business practice, with other correspondence of Pacific Gas and Electric Company, enclosed in a sealed envelope, with postage fully prepaid, addressed to those parties listed on the official service list for R.99-11-022, R.04-04-003, R.04-04-025, R.06-02-013 and A.08-11-001 without an e-mail address.

I certify and declare under penalty of perjury under the laws of the State of California that the foregoing is true and correct.

Executed on this 8th day of October 2010 at San Francisco, California.

PAMELA J. DAWSON-SMITH

### THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA EMAIL SERVICE LIST CRUS DOCKET NO. PROCESSES ASSAURANT PROCESSES

CPUC DOCKET NO. R0404003-R0404025-A0811001-R9911022-R0602013

ab1@cpuc.ca.gov;abb@eslawfirm.com;achang@efficiencycouncil.org;ACT6@pge.com;AEG@cpuc.ca.gov;agc@cpuc.ca.gov;agrimaldi@mckennalong.com;alexm@calpine.com;alho@pge.com;aliddell@icfi.com;allwazeready@aol.com;ALR4@pge.com;amber.wyatt@sce.com;amber@iepa.com;andrea.morrison@directenergy.com;andrew.dalton@valero.com;andy.vanhorn@vhcenergy.com;angela.kim@fticonsulting.com;anogee@ucsusa.org;ATrial@SempraUtilities.com;atrowbridge@daycartermurphy.com;AxL3@pge.com;ayk@cpuc.ca.gov;b.buchynsky@dgc-

us.com;bbc@cpuc.ca.gov;bcragg@goodinmacbride.com;bdicapo@caiso.com;bernardo@braunle gal.com;beth.fox@sce.com;beth@beth411.com;bfinkelstein@turn.org;bhines@svlg.net;bill@jbse nergy.com;billjulian@sbcglobal.net;bjl@bry.com;bkc7@pge.com;blaising@braunlegal.com;bmcc@mccarthylaw.com;bmd@cpuc.ca.gov;bmeister@energy.state.ca.us;bobgex@dwt.com;bpowers@powersengineering.com;brbarkovich@earthlink.net;brianhaney@useconsulting.com;bruce.fost er@sce.com;btang@ci.azusa.ca.us;californiadockets@pacificorp.com;car@cpuc.ca.gov;carla.pet erman@gmail.com;carlo.zorzoli@enel.it;carol.schmid-

fraee@sce.com;carol.schmidfrazee@sce.com;Case.Admin@sce.com;cathy.karlstad@sce.com;cbk@eslawfirm.com;ccasselman@pilotpowergroup.com;cem@newsdata.com;CentralFiles@SempraUtilities.com;CFPena@SempraUtilities.com;chilen@NVEnergy.com;chilen@sppc.com;chris.ohara@nrgenergy.com;Claufenb@energy.state.ca.us;cleni@energy.state.ca.us;clyde.murley@comcast.net;cmkehrein@ems-

ca.com;cneedham@edisonmission.com;cpollina@winston.com;CPUCCases@pge.com;cpucdockets@keyesandfox.com;crmd@pge.com;crochlin@socalgas.com;csmoots@perkinscoie.com;ctorchia@chadbourne.com;cynthia.brady@constellation.com;CZammit@SempraUtilities.com;daipm@daioildale.com;DAKing@SempraUtilities.com;DAKinports@SempraUtilities.com;david.reynolds@ncpa.com;david@branchcomb.com;davidmorse9@gmail.com;DBP@cpuc.ca.gov;dbr@cpuc.ca.gov;dcarroll@downeybrand.com;ddavie@wellhead.com;deana.ng@sce.com;deb@a-

klaw.com;dehling@klng.com;dfredericks@dgpower.com;dgrandy@caonsitegen.com;dgulino@rid gewoodpower.com;dhuard@manatt.com;Diane.Fellman@nrgenergy.com;Dick@DavisHydro.com;djh@cpuc.ca.gov;dkk@eslawfirm.com;Dkolk@compenergy.com;dmarcus2@sbcglobal.net;dmcfarlan@mwgen.com;DNiehaus@SempraUtilities.com;Don.Vawter@AES.com;doug.kiviat@morganstanley.com;douglass@energyattorney.com;douglass@energyattorney.com;douglass@energyattorney.com;dsaul@pacificsolar.net;dtateosian@powereng.com;dug@cpuc.ca.gov;dvidaver@energy.state.ca.us;dwood8@cox.net;dwoods@whitecase.com;dws@r-c-s-

inc.com;ecrem@ix.netcom.com;edchang@flynnrci.com;edf@cpuc.ca.gov;editorial@californiaenergycircuit.net;edwardoneill@dwt.com;ej\_wright@oxy.com;ek@a-

klaw.com;eks@cpuc.ca.gov;eleuze@rrienergy.com;ELL5@pge.com;emello@sppc.com;epoole@adplaw.com;e-recipient@caiso.com;e-

recipient@caiso.com;eric@strategyi.com;etiedemann@kmtg.com;ewheless@lacsd.org;filings@a

klaw.com;fmobasheri@aol.com;fortlieb@sandiego.gov;gabriellilaw@sbcglobal.net;gary.allen@sce.com;garyi@enxco.com;gaw@cpuc.ca.gov;GBaker@SempraUtilities.com;GDixon@SempraUtilities.com;gig@cpuc.ca.gov;glw@eslawfirm.com;gmorris@emf.net;grosenblum@caiso.com;gtd@cpuc.ca.gov;gustavo.luna@aes.com;gwung@mwe.com;GXL2@pge.com;GxZ5@pge.com;hchoy@isd.co.la.ca.us;Henry.Nanjo@dgs.ca.gov;hoerner@redefiningprogress.org;HYao@SempraUtilities.com;hypower@pacbell.net;irene.stillings@energycenter.org;iryna.kwasny@doj.ca.gov;j.eric.isken@sce.com;jackmack@suesec.com;janet.combs@sce.com;janet.combs@sce.com;janice@strategenconsulting.com;janreid@coastecon.com;jarmstrong@gmssr.com;jbloom@winston.com;jbloom@winston.com;jbloom@winston.com;jbloom@winston.com;jennifer.porter@energycenter.org;jesus.arredondo@nrgenergy.com;jesus.arredond

inc.com;jkarp@winston.com;JKloberdanz@SempraUtilities.com;jlehman@anaheim.net;jleslie@luce.com;jluckhardt@downeybrand.com;jm3@cpuc.ca.gov;jmcmahon@8760energy.com;jmh@cpuc.ca.gov;joc@cpuc.ca.gov;jody\_london\_consulting@earthlink.net;joe.paul@dynegy.com;joh@cpuc.ca.gov;johnrredding@earthlink.net;jon.jacobs@paconsulting.com;jordan.white@pacificorp.co

# THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA EMAIL SERVICE LIST CPUC DOCKET NO. R0404003-R0404025-A0811001-R9911022-R0602013

m;joyw@mid.org;JPacheco@SempraUtilities.com;jscancarelli@crowell.com;jshields@ssjid.com;jsp5@pge.com;jsqueri@gmssr.com;jst@cpuc.ca.gov;judypau@dwt.com;julie.martin@bp.com;julie.n.dumoulin-

smith@ubs.com;jweil@aglet.org;jwoodwar@energy.state.ca.us;JYamagata@SempraUtilities.com;k.abreu@sbcglobal.net;karen.lee@sce.com;karen@klindh.com;karp@pge.com;Kathryn.Wig@nrgenergy.com;kcj5@pge.com;KCordova@SempraUtilities.com;kdusel@navigantconsulting.com;kdw@cpuc.ca.gov;kdw@woodruff-expert-

services.com;keith.mccrea@sablaw.com;kenneth.swain@navigantconsulting.com;kerry.hattevik @nexteraenergy.com;kgriffin@energy.state.ca.us;kho@cpuc.ca.gov;khojasteh.davoodi@navy.mil;kjk@kjkammerer.com;kjohnson@caiso.com;kjsimonsen@ems-

ca.com;klatt@energyattorney.com;klatt@energyattorney.com;KMelville@SempraUtilities.com;kmi lls@cfbf.com;kmkiener@cox.net;KMorton@SempraUtilities.com;kmudge@covad.com;kowalewsk ia@calpine.com;kpp@cpuc.ca.gov;l\_brown369@yahoo.com;lau@cpuc.ca.gov;LauckhartR@bv.com;laura.genao@sce.com;lcottle@winston.com;leon.bass@sce.com;lettenson@nrdc.org;lgk2@pge.com;liddell@energyattorney.com;lisa\_weinzimer@platts.com;lisa weinzimer@sbcglobal.net;lkostrzewa@edisonmission.com;lmackey@lspower.com;lmh@eslawfirm.com;loe@cpuc.ca.gov;lra@cpuc.ca.gov;LSchavrien@SempraUtilities.com;luluw@newsdata.com;LUrick@SempraUtilities.com;luta1@bp.com;lwong@energy.state.ca.us;lys@a-

klaw.com;MAGq@pge.com;map@cpuc.ca.gov;marcie.milner@shell.com;Marshall.Clark@dgs.ca.gov;martinhomec@gmail.com;mary.lynch@constellation.com;mbrubaker@consultbai.com;mc3@cpuc.ca.gov;mclaughlin@braunlegal.com;mdjoseph@adamsbroadwell.com;mdozier@caiso.com;mecsoft@pacbell.net;mflorio@turn.org;mgreen@palco.com;mhharrer@sbcglobal.net;michael.backstrom@sce.com;michael.evans@shell.com;michael.hindus@pillsburylaw.com;michaelboyd@sbcglobal.net;mike.montoya@sce.com;mike.tierney@nrgenergy.com;mjaske@energy.state.ca.us;mjd@cpuc.ca.gov;mkh@cpuc.ca.gov;mmiller@energy.state.ca.us;Monica.Schwebs@bingham.com;mpa@a-klaw.com;mpa@a-

klaw.com;mpryor@energy.state.ca.us;mrh2@pge.com;mrw@mrwassoc.com;mrw@mrwassoc.com;mrw@mrwassoc.com;mshames@ucan.org;msw@cpuc.ca.gov;mth@cpuc.ca.gov;myuffee@mwe.com;nao@cpuc.ca.gov;neburgess@sycamore.com;nes@a-

klaw.com;nlong@nrdc.org;norman.furuta@navy.mil;npedersen@hanmor.com;nrader@igc.org;os hirock@pacbell.net;paulfenn@local.org;pcmcdonnell@earthlink.net;pduvair@energy.state.ca.us; pepper@cleanpowermarkets.com;phanschen@mofo.com;pherrington@edisonmission.com;pheuer-

cv@comcast.net;phil@reesechambers.com;philha@astound.net;philm@scdenergy.com;pholley @covantaenergy.com;pmaxwell@navigantconsulting.com;ppl@cpuc.ca.gov;psd@cpuc.ca.gov;ps toner@lgc.org;pucservice@manatt.com;PVillegas@SempraUtilities.com;pzs@cpuc.ca.gov;r.forgi one@intpower.com;raj.pankhania@ci.hercules.ca.us;ralf1241a@cs.com;rantonopoulos@calpine. com;rcox@pacificenvironment.org;regrelcpuccases@pge.com;ren@ethree.com;REO5@pge.com ;RFG2@pge.com;rfp@eesconsulting.com;rfreeh123@sbcglobal.net;rhwiser@lbl.gov;rick\_noger @praxair.com;rick noger@praxair.com;rkmoore@gswater.com;rls@cpuc.ca.gov;rmccann@umic h.edu;rmiller@energy.state.ca.us;rnevis@daycartermurphy.com;robyn.naramore@sce.com;rocky .ho@fticonsulting.com;roger@berlinerlawpllc.com;ron.dahlin@ge.com;rott@rrienergy.com;rsande rs@hlpower.com;rschmidt@bartlewells.com;rshapiro@chadbourne.com;rwalther@pacbell.net;sa eed.farrokhpay@ferc.gov;salleyoo@dwt.com;sarveybob@aol.com;saw0@pge.com;sberlin@mcc arthylaw.com;sbeserra@sbcglobal.net;scott.tomashefsky@ncpa.com;sdavies@caiso.com;sdross i@calpx.com;Sean.Beatty@mirant.com;seb@cpuc.ca.gov;seboyd@tid.org;SEHC@pge.com;sep hra.ninow@energycenter.org;sesco@optonline.net;sfr@sandaq.org;sfrichardson@winston.com;s hi@cpuc.ca.gov;sisser@goodcompanyassociates.com;sisser@goodcompanyassociates.com;SJ P@cpuc.ca.gov;ska@cpuc.ca.gov;skg@cpuc.ca.gov;skh@cpuc.ca.gov;slefton@aptecheng.com; slg0@pge.com;sls@a-

klaw.com;SMK@cpuc.ca.gov;SNelson@SempraUtilities.com;snuller@ethree.com;srovetti@sfwat er.org;sscb@pge.com;ssmyers@att.net;steve.koerner@elpaso.com;stevegreenwald@dwt.com;steven.huhman@morganstanley.com;steven.schleimer@barclayscapital.com;steven@iepa.com;st

## THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA EMAIL SERVICE LIST CPUC DOCKET NO. R0404003-R0404025-A0811001-R9911022-R0602013

eveng@destrategies.com;svn@cpuc.ca.gov;taj8@pge.com;tam.hunt@gmail.com;tblair@sandieg o.gov;tbo@cpuc.ca.gov;tcarlson@rrienergy.com;tciardella@nvenergy.com;tcr@cpuc.ca.gov;tcx @cpuc.ca.gov;tdarton@pilotpowergroup.com;tdillard@sppc.com;tdp@cpuc.ca.gov;ted@energy-solution.com;tim.hemig@nrgenergy.com;timea.Zentai@navigantconsulting.com;todil@mckennalong.com;tomb@crossborderenergy.com;tomk@mid.org;tomk@mid.org;toms@i-cpg.com;tory.weber@sce.com;TRoberts@SempraUtilities.com;tsolomon@winston.com;tyf@cpuc.ca.gov;unc@cpuc.ca.gov;vjb@cpuc.ca.gov;vjw3@pge.com;vwood@smud.org;wamer@kirkwood

law.com;wem@igc.org;wesley.spowhn@pillsburylaw.com;will.mitchell@cpv.com;william.tomlinson@elpaso.com;WKeilani@SempraUtilities.com;wolff@smwlaw.com;wsm@cpuc.ca.gov;WTobin@SempraGlobal.com;wvm3@pge.com;yxg4@pge.com;

.com;WBlattner@SempraUtilities.com;wbooth@booth-

### THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA SERVICE LIST CPUC DOCKET NO. R0404003-R0404025-A0811001-R9911022-R0602013

Total number of addressees: 455

ALICE GONG

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST. MC B9A SAN FRANCISCO CA 94105

FOR: PACIFIC GAS AND ELECTRIC COMPANY

Email: AxL3@pge.com Status: INFORMATION

ANGELA TORR

PACIFIC GAS & ELECTRIC COMPANY

MC N13E 245 MARKET ST

SAN FRANCISCO CA 94105 Email: ACT6@pge.com Status: INFORMATION

GEORGE ZAHARIUDAKIS

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, RM. 904, MC B9A SAN FRANCISCO CA 94105 Email: GxZ5@pge.com Status: INFORMATION

JOHN PAPPAS UTILITY ELECTRIC PORTFOLIO

**MANAGEMENT** 

PACIFIC GAS AND ELECTRIC COMPANY

245 MARKET ST, MC N12G SAN FRANCISCO CA 94105 Email: jsp5@pge.com Status: INFORMATION

MARK R. HUFFMAN ATTORNEY

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST / PO BOX 7442 (B30A)

SAN FRANCISCO CA 94105

FOR: PACIFIC GAS AND ELECTRIC COMPANY

Email: mrh2@pge.com Status: INFORMATION

ROGER GOLDSTEIN

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, MC B9A SAN FRANCISCO CA 94105 Email: RFG2@pge.com Status: INFORMATION

VALERIE WINN PROJECT MANAGER

PACIFIC GAS & ELECTRIC 245 MARKET ST, MC N12G SAN FRANCISCO CA 94105

Email: vjw3@pge.com Status: INFORMATION ALICE L. REID LAW DEPARTMENT PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, RM 3081-B30A SAN FRANCISCO CA 94105 Email: ALR4@pge.com

Status: INFORMATION

ED LUCHA

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, MC B9A, RM 991 SAN FRANCISCO CA 94105 Email: ELL5@pge.com Status: INFORMATION

JENNY GLUZGOLD

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST MCB9A SAN FRANCISCO CA 94105 Email: yxg4@pge.com Status: INFORMATION

KIMBERLY C. JONES

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, MC B9A SAN FRANCISCO CA 94105 Email: kcj5@pge.com Status: INFORMATION

REGULATORY FILE ROOM

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, B30A / PO BOX 7442 SAN FRANCISCO CA 94105 Email: CPUCCases@pge.com Status: INFORMATION

**RUTH OSCAR** 

PACIFIC GAS AND ELECTRIC COMPANY

245 MARKET ST, N12G SAN FRANCISCO CA 94105 Email: REO5@pge.com Status: INFORMATION

ANDREW L. HARRIS

PACIFIC GAS & ELECTRIC COMPANY

PO BOX 770000 MAIL CODE B9A SAN FRANCISCO CA 94177

Email: alho@pge.com Status: INFORMATION

### THE PUBLIC UTILITIES COMMISSION OF THE STATE OF CALIFORNIA SERVICE LIST CPUC DOCKET NO. R0404003-R0404025-A0811001-R9911022-R0602013

Total number of addressees: 455

BRIAN K. CHERRY DIRECTOR, REGULATORY RELATIONS

PACIFIC GAS AND ELECTRIC COMPANY (39) 77N BEALE ST., PO BOX 770000, MC B10C

SAN FRANCISCO CA 94177 Email: bkc7@pge.com Status: INFORMATION

GRACE LIVINGSTON-NUNLEY ASSISTANT PROJECT MANAGER

PACIFIC GAS AND ELECTRIC COMPANY

PO BOX 770000 MAIL CODE B9A SAN FRANCISCO CA 94177 Email: GXL2@pge.com Status: INFORMATION

LUCY FUKUI

PACIFIC GAS AND ELECTRIC COMPANY PO BOX 770000, RM. 918 - MC B9A SAN FRANCISCO CA 94177

FOR: PACIFIC GAS AND ELECTRIC COMPANY

Email: lgk2@pge.com Status: INFORMATION

SHAUN HALVERSON

PACIFIC GAS AND ELECTRIC COMPANY

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000

FOR: PACIFIC GAS AND ELECTRIC COMPANY

Email: SEHC@pge.com Status: INFORMATION

SHIRLEY WOO ATTORNEY

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE ST, B30A SAN FRANCISCO CA 94105

FOR: Pacific Gas and Electric Company

Email: saw0@pge.com

Status: PARTY

CHARLES R. MIDDLEKAUFF

PACIFIC GAS AND ELECTRIC COMPANY

PO BOX 7442

SAN FRANCISCO CA 94120

FOR: Pacific Gas and Electric Company

Email: crmd@pge.com

Status: PARTY

MARY A. GANDESBERY

PACIFIC GAS AND ELECTRIC COMPANY

PO BOX 7442, MC B30A-3005 SAN FRANCISCO CA 94120-7442 FOR: Pacific Gas and Electric Company

Email: MAGq@pge.com

Status: PARTY

CASE COORDINATION

PACIFIC GAS AND ELECTRIC COMPANY

PO BOX 770000; MC B9A SAN FRANCISCO CA 94177 Email: regrelcpuccases@pge.com

Status: INFORMATION

KATHERINE RYZHAYA

**PACIFIC GAS & ELECTRIC COMPANY** 

MAIL CODE B9A PO BOX 770000

SAN FRANCISCO CA 94177

FOR: PACIFIC GAS & ELECTRIC COMPANY

Email: karp@pge.com Status: INFORMATION

SEBASTIEN CSAPO

PG&E PROJECT MGR. MAIL CODE B9A

PO BOX 770000

SAN FRANCISCO CA 94177 Email: sscb@pge.com Status: INFORMATION

**TOM JARMAN** 

PACIFIC GAS AND ELECTRIC COMPANY

77 BEALE SATREET, RM. 909, MC B9A SAN FRANCISCO CA 94105-1814

FOR: PACIFIC GAS AND ELECTRIC COMPANY

Email: taj8@pge.com Status: INFORMATION

STEPHEN L. GARBER ATTORNEY

PACIFIC GAS AND ELECTRIC COMPANY 77 BEALE ST; MCB30A

77 BEALE ST; MCB30A SAN FRANCISCO CA 94105

FOR: Pacific Gas and Electric Company

Email: slg0@pge.com Status: PARTY

WILLIAM V. MANHEIM ATTORNEY

PACIFIC GAS AND ELECTRIC COMPANY

PO BOX 7442, LAW DEPT. SAN FRANCISCO CA 94120

FOR: PACIFIC GAS AND ELECTRIC COMPANY

Email: wvm3@pge.com

Status: PARTY

Laurence Chaset

CALIF PUBLIC UTILITIES COMMISSION

LEGAL DIVISION

505 VAN NESS AVE RM 5131 SAN FRANCISCO CA 94102-3214

Email: lau@cpuc.ca.gov Status: INFORMATION

Total number of addressees: 455

SOPHIA PARK

CALIFORNIA PUBLIC UTILITIES COMMISSION

EMAIL ONLY EMAIL ONLY CA 0 Email: SJP@cpuc.ca.gov

Status: STATE-SERVICE

Peter Lai

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 320 WEST 4TH ST STE 500 LOS ANGELES CA 90013 Email: ppl@cpuc.ca.gov Status: STATE-SERVICE

ANNE GILLETTE

CALIFORNIA PUBLIC UTILITIES COMMISSION

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000 Email: AEG@cpuc.ca.gov Status: STATE-SERVICE

SARA KAMINS

CALIFORNIA PUBLIC UTILITIES COMMISSION

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000 Email: SMK@cpuc.ca.gov Status: STATE-SERVICE

Amy C. Yip-Kikugawa

CALIF PUBLIC UTILITIES COMMISSION

**EXECUTIVE DIVISION** 505 VAN NESS AVE RM 2106 SAN FRANCISCO CA 94102-3214

Email: ayk@cpuc.ca.gov Status: STATE-SERVICE

Bishu Chatterjee

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: bbc@cpuc.ca.gov Status: STATE-SERVICE

Darwin Farrar

CALIF PUBLIC UTILITIES COMMISSION

DIVISION OF ADMINISTRATIVE LAW JUDGES 505 VAN NESS AVE RM 5041 SAN FRANCISCO CA 94102-3214

Email: edf@cpuc.ca.gov Status: STATE-SERVICE James Loewen

**CALIF PUBLIC UTILITIES COMMISSION** 

**ENERGY DIVISION** 

320 WEST 4TH ST STE 500 LOS ANGELES CA 90013 Email: loe@cpuc.ca.gov Status: STATE-SERVICE

Wade McCartney

**CALIF PUBLIC UTILITIES COMMISSION** 

POLICY & PLANNING DIVISION

770 L ST, STE 1050 SACRAMENTO CA 95814 Email: wsm@cpuc.ca.gov Status: STATE-SERVICE

DAVID PECK

CALIFORNIA PUBLIC UTILITIES COMMISSION

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000 Email: DBP@cpuc.ca.gov Status: STATE-SERVICE

Amy C. Baker

CALIF PUBLIC UTILITIES COMMISSION

ENERGY DIVISION

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: ab1@cpuc.ca.gov Status: STATE-SERVICE

Andrew Campbell

CALIF PUBLIC UTILITIES COMMISSION

**EXECUTIVE DIVISION** 505 VAN NESS AVE RM 5203 SAN FRANCISCO CA 94102-3214

Email: agc@cpuc.ca.gov Status: STATE-SERVICE

Bruce DeBerry

CALIF PUBLIC UTILITIES COMMISSION

DIVISION OF ADMINISTRATIVE LAW JUDGES 505 VAN NESS AVE RM 5043

SAN FRANCISCO CA 94102-3214

Email: bmd@cpuc.ca.gov Status: STATE-SERVICE

Donna J. Hines

**CALIF PUBLIC UTILITIES COMMISSION ELECTRICITY PLANNING & POLICY BRANCH** 

505 VAN NESS AVE RM 4102 SAN FRANCISCO CA 94102-3214

Email: djh@cpuc.ca.gov Status: STATE-SERVICE

Total number of addressees: 455

Douglas M. Long CALIF PUBLIC UTILITIES COMMISSION DIVISION OF ADMINISTRATIVE LAW JUDGES

505 VAN NESS AVE RM 5023 SAN FRANCISCO CA 94102-3214

Email: dug@cpuc.ca.gov Status: STATE-SERVICE

Gregory A. Wilson CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214 FOR: CPUC - ENERGY DIVISION

Email: gaw@cpuc.ca.gov Status: STATE-SERVICE

Jaclyn Marks

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: jm3@cpuc.ca.gov Status: STATE-SERVICE

Jerry Oh

**CALIF PUBLIC UTILITIES COMMISSION** 

WATER BRANCH 505 VAN NESS AVE RM 3200 SAN FRANCISCO CA 94102-3214

Email: joh@cpuc.ca.gov Status: STATE-SERVICE

Julie Halligan

CALIF PUBLIC UTILITIES COMMISSION

CONSUMER PROTECTION AND SAFETY DIVISION

505 VAN NESS AVE RM 2203 SAN FRANCISCO CA 94102-3214

Email: jmh@cpuc.ca.gov Status: STATE-SERVICE

Kathryn Auriemma

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: kdw@cpuc.ca.gov Status: STATE-SERVICE

Laura A. Martin

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: Ira@cpuc.ca.gov Status: STATE-SERVICE Elizabeth Stoltzfus

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: eks@cpuc.ca.gov Status: STATE-SERVICE

Gretchen T. Dumas

CALIF PUBLIC UTILITIES COMMISSION

**LEGAL DIVISION** 

505 VAN NESS AVE RM 4300 SAN FRANCISCO CA 94102-3214

Email: gtd@cpuc.ca.gov Status: STATE-SERVICE

Jeorge S. Tagnipes

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: jst@cpuc.ca.gov Status: STATE-SERVICE

Joe Como

CALIF PUBLIC UTILITIES COMMISSION

DRA - ADMINISTRATIVE BRANCH 505 VAN NESS AVE RM 4101 SAN FRANCISCO CA 94102-3214

Email: joc@cpuc.ca.gov Status: STATE-SERVICE

Karen P. Paull

**CALIF PUBLIC UTILITIES COMMISSION** 

**LEGAL DIVISION** 

505 VAN NESS AVE RM 4300 SAN FRANCISCO CA 94102-3214

Email: kpp@cpuc.ca.gov Status: STATE-SERVICE

Ke Hao Ouyang

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY PRICING AND CUSTOMER PROGRAMS** 

BRANCH

505 VAN NESS AVE RM 4104 SAN FRANCISCO CA 94102-3214

Email: kho@cpuc.ca.gov Status: STATE-SERVICE

Mark S. Wetzell

**CALIF PUBLIC UTILITIES COMMISSION** 

DIVISION OF ADMINISTRATIVE LAW JUDGES

505 VAN NESS AVE RM 5009 SAN FRANCISCO CA 94102-3214

Email: msw@cpuc.ca.gov Status: STATE-SERVICE

Total number of addressees: 455

Matthew Deal

**CALIF PUBLIC UTILITIES COMMISSION** 

POLICY & PLANNING DIVISION 505 VAN NESS AVE RM 5119 SAN FRANCISCO CA 94102-3214

Email: mjd@cpuc.ca.gov Status: STATE-SERVICE

Michael Colvin

**CALIF PUBLIC UTILITIES COMMISSION** 

POLICY & PLANNING DIVISION 505 VAN NESS AVE RM 5119 SAN FRANCISCO CA 94102-3214

Email: mc3@cpuc.ca.gov Status: STATE-SERVICE

Paul Douglas

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: psd@cpuc.ca.gov Status: STATE-SERVICE

Robert Kinosian

CALIF PUBLIC UTILITIES COMMISSION

EXECUTIVE DIVISION 505 VAN NESS AVE RM 5202 SAN FRANCISCO CA 94102-3214

Email: gig@cpuc.ca.gov Status: STATE-SERVICE

Sepideh Khosrowjah

CALIF PUBLIC UTILITIES COMMISSION

EXECUTIVE DIVISION 505 VAN NESS AVE RM 5204 SAN FRANCISCO CA 94102-3214

Email: skh@cpuc.ca.gov Status: STATE-SERVICE

Steven K. Haine

CALIF PUBLIC UTILITIES COMMISSION

ENERGY DIVISION

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: shi@cpuc.ca.gov Status: STATE-SERVICE

Teresa Hortinela

CALIF PUBLIC UTILITIES COMMISSION

ELECTRICITY PLANNING & POLICY BRANCH 505 VAN NESS AVE RM 4102

505 VAN NESS AVE RM 4102 SAN FRANCISCO CA 94102-3214

Email: mth@cpuc.ca.gov Status: STATE-SERVICE Melissa Semcer

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: unc@cpuc.ca.gov Status: STATE-SERVICE

Mikhail Haramati

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: mkh@cpuc.ca.gov Status: STATE-SERVICE

Peter Skala

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: ska@cpuc.ca.gov Status: STATE-SERVICE

Sean A. Simon

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: svn@cpuc.ca.gov Status: STATE-SERVICE

Simon Baker

**CALIF PUBLIC UTILITIES COMMISSION** 

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

Email: seb@cpuc.ca.gov Status: STATE-SERVICE

Sudheer Gokhale

CALIF PUBLIC UTILITIES COMMISSION

**ELECTRICITY PLANNING & POLICY BRANCH** 

505 VAN NESS AVE RM 4102 SAN FRANCISCO CA 94102-3214

Email: skg@cpuc.ca.gov Status: STATE-SERVICE

Terrie D. Prosper

CALIF PUBLIC UTILITIES COMMISSION

EXECUTIVE DIVISION
505 VAN NESS AVE RM 5301
SAN FRANCISCO CA 94102-3214

Email: tdp@cpuc.ca.gov Status: STATE-SERVICE

Total number of addressees: 455

Theresa Cho

CALIF PUBLIC UTILITIES COMMISSION

EXECUTIVE DIVISION 505 VAN NESS AVE RM 5207 SAN FRANCISCO CA 94102-3214

Email: tcx@cpuc.ca.gov Status: STATE-SERVICE

Traci Bone

CALIF PUBLIC UTILITIES COMMISSION

LEGAL DIVISION 505 VAN NESS AVE RM 5027 SAN FRANCISCO CA 94102-3214

Email: tbo@cpuc.ca.gov Status: STATE-SERVICE

Valerie Beck

CALIF PUBLIC UTILITIES COMMISSION

ELECTRIC GENERATION PERFORMANCE BRANCH

505 VAN NESS AVE AREA 2-D SAN FRANCISCO CA 94102-3214

Email: vjb@cpuc.ca.gov Status: STATE-SERVICE

Robert L. Strauss

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

FOR: Energy Division
Email: rls@cpuc.ca.gov
Status: STATE-SERVICE

Pearlie Sabino

CALIF PUBLIC UTILITIES COMMISSION

ENERGY COST OF SERVICE & NATURAL GAS BRANCH

505 VAN NESS AVE RM 4209 SAN FRANCISCO CA 94102-3214

FOR: OFFICE OF RATEPAYER ADVOCATES (ORA)

Email: pzs@cpuc.ca.gov Status: STATE-SERVICE

DON VAWTER

**AES ALAMITOS, LLC** 

690 N. STUDEBAKER RD.
LONG BEACH CA 90803
Email: Don.Vawter@AES.com

Status: INFORMATION

JAMES WEIL DIRECTOR
AGLET CONSUMER ALLIANCE

PO BOX 1916

SEBASTOPOL CA 95473

FOR: AGLET CONSUMER ALLIANCE

Email: jweil@aglet.org Status: PARTY Thomas Roberts

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY PRICING AND CUSTOMER PROGRAMS** 

BRANCH

505 VAN NESS AVE RM 4104 SAN FRANCISCO CA 94102-3214

Email: tcr@cpuc.ca.gov Status: STATE-SERVICE

Tracy Y. Fok

CALIF PUBLIC UTILITIES COMMISSION

UTILITY AUDIT, FINANCE & COMPLIANCE BRANCH

505 VAN NESS AVE RM 3-B SAN FRANCISCO CA 94102-3214

Email: tyf@cpuc.ca.gov Status: STATE-SERVICE

Donald J. Brooks

CALIF PUBLIC UTILITIES COMMISSION

**ENERGY DIVISION** 

505 VAN NESS AVE AREA 4-A SAN FRANCISCO CA 94102-3214

FOR: Energy Division Email: dbr@cpuc.ca.gov Status: STATE-SERVICE

Marion Peleo

CALIF PUBLIC UTILITIES COMMISSION

LEGAL DIVISION

505 VAN NESS AVE RM 4107 SAN FRANCISCO CA 94102-3214

FOR: ORA

Email: map@cpuc.ca.gov

Status: PARTY

Steve Linsey

CALIF PUBLIC UTILITIES COMMISSION

CONSUMER ISSUES ANALYSIS BRANCH 505 VAN NESS AVE RM 2013

SAN FRANCISCO CA 94102-3214

FOR: ORA

Email: car@cpuc.ca.gov Status: STATE-SERVICE

**GUSTAVO LUNA** 

**AES CORPORATION** 

690 N. STUDEBAKER RD. LONG BEACH CA 90803 Email: gustavo.luna@aes.com

Status: INFORMATION

SEEMA SRINIVASAN ALCANTAR & KAHL, LLP

33 NEW MONTGOMERY ST, STE 1850

SAN FRANCISCO CA 94105 Email: sls@a-klaw.com

Total number of addressees: 455

DANIEL W. DOUGLASS
DOUGLASS & LIDDELL

21700 OXNARD ST, STE 1030
WOODLAND HILLS CA 91367-8102
FOR: Alliance for Retail Energy Markets
Email: douglass@energyattorney.com

Status: PARTY

WENDY LIVORNESE

RIDGEWOOD POWER CORPORATION

947 LINWOOD AVE RIDGEWOOD NJ 7450

FOR: Altamont Cogen Corporation, Monterey Power

Company

Status: INFORMATION

FRANK ANNUNZIATO PRESIDENT AMERICAN UTILITY NETWORK INC.

10705 DEER CANYON DR. ALTA LOMA CA 91737-2483 Email: allwazeready@aol.com Status: INFORMATION

EDWARD G. POOLE
ANDERSON & POOLE

601 CALIFORNIA ST, STE 1300 SAN FRANCISCO CA 94108-2812 Email: epoole@adplaw.com

Status: INFORMATION

DAVID REYNOLDS MEMBER SERVICES MANAGER NORTHERN CALIFORNIA POWER AGENCY

651 COMMERCE DRIVE
ROSEVILLE CA 95678-6420
FOR: ASPEN SYSTEMS CORP
Email: david.reynolds@ncpa.com

Status: INFORMATION

REED V. SCHMIDT BARTLE WELLS ASSOCIATES 1889 ALCATRAZ AVE

BERKELEY CA 94703-2714

FOR: BARTLE WELLS ASSOCIATES

Email: rschmidt@bartlewells.com

Status: INFORMATION

BARRY LOVELL
BERRY PETROLEUM COMPANY
5201 TRUXTUN AVE., STE 300
BAKERSFIED CA 93309

FOR: BERRY PETROLEUM COMPANY

Email: bjl@bry.com Status: INFORMATION GREGORY KLATT

DOUGLASS & LIDDELL

21700 OXNARD ST, STE 1030

WOODLAND HILLS CA 91367-8102

FOR: Alliance for Retail Energy Markets/The Filing Parties

Email: klatt@energyattorney.com

Status: PARTY

GOODWIN SCOTT AMERICAN ENERGY, INC. 3330 CLAYTON ROAD, STE B CONCORD CA 94519

Status: INFORMATION

JANIS LEHMAN

ANAHEIM PUBLIC UTILITIES
201 S. ANAHEIM BLVD., STE 1101

ANAHEIM CA 92805

FOR: Anaheim Public Utilities Email: jlehman@anaheim.net Status: INFORMATION

STEVEN A. LEFTON VP POWER PLANT PROJECTS APTECH ENGINEERING SERVICES INC.

PO BOX 3440

SUNNYVALE CA 94089-3440

FOR: APTECH ENGINEERING SERVICES INC.

Email: slefton@aptecheng.com Status: INFORMATION

STEVEN S. SCHLEIMER DIRECTOR, COMPLIANCE &

REGULATORY AFFAIRS BARCLAYS BANK, PLC 200 PARK AVE, FIFTH FLR NEW YORK NY 10166

Email: steven.schleimer@barclayscapital.com

Status: INFORMATION

RYAN WISER BERKELEY LAB

1 CYCLOTRON ROAD, MS-90-4000

BERKELEY CA 94720 Email: rhwiser@lbl.gov Status: INFORMATION

ZANE GLENN

BIG VALLEY POWER, LLC. 1615 CONTINENTAL ST, STE 100

REDDING CA 96001 Status: INFORMATION

Total number of addressees: 455

J. RICHARD LAUCKHART BLACK & VEATCH

10995 GOLD CENTER DRIVE, STE 100 RANCHO CORDOVA CA 95670

Email: LauckhartR@bv.com Status: INFORMATION

BRUCE MCLAUGHLIN ATTORNEY

BRAUN & BLAISING, P.C. 915 L ST STE 1270

SACRAMENTO CA 95814

Email: mclaughlin@braunlegal.com Status: INFORMATION

MAURICE BRUBAKER
BRUBAKER & ASSOCIATES

PO BOX 412000

1215 FERN RIDGE PARKWAY, STE 208

ST. LOUIS MO 63141

Email: mbrubaker@consultbai.com

Status: INFORMATION

IRENE M. STILLINGS EXECUTIVE DIRECTOR CALIF. CTR. FOR SUSTAINABLE ENERGY

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000

Email: irene.stillings@energycenter.org

Status: INFORMATION

JENNIFER PORTER POLICY ANALYST

**CALIFORNIA CENTER FOR SUSTAINABLE ENERGY** 

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000

Email: jennifer.porter@energycenter.org

Status: INFORMATION

BETH VAUGHAN

CALIFORNIA COGENERATION COUNCIL

4391 N. MARSH ELDER COURT

CONCORD CA 94521
Email: beth@beth411.com
Status: INFORMATION

JERRY R. BLOOM ATTORNEY WINSTON & STRAWN LLP

333 SOUTH GRAND AVE, 38TH FLR LOS ANGELES CA 90071-1543

FOR: California Cogeneration Council

Email: jbloom@winston.com

Status: PARTY

DAVID BRANCHCOMB

BRANCHCOMB ASSOCIATES, LLC

9360 OAKTREE LANE ORANGEVILLE CA 95662 Email: david@branchcomb.com

Status: INFORMATION

RYAN BERNARDO

BRAUN BLAISING MCLAUGHLIN, P.C.

915 L ST, STE 1270 SACRAMENTO CA 95814 Email: bernardo@braunlegal.com

Status: INFORMATION

AUDREY CHANG

CA ENERGY EFFICIENCY INDUSTRY COUNCIL

EMAIL ONLY CA 0

Email: achang@efficiencycouncil.org

Status: INFORMATION

CHI DOAN

CALIF. DEPT OF WATER RESOURCES

3310 EL CAMINO AVE., RM LL94 SACRAMENTO CA 95821 Status: STATE-SERVICE

SEPHRA A. NINOW

CALIFORNIA CENTER FOR SUSTAINABLE ENERGY

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000

Email: sephra.ninow@energycenter.org

Status: INFORMATION

THOMAS W. SOLOMON ATTORNEY

WINSTON & STRAWN LLP 101 CALIFORNIA ST, 39TH FLR SAN FRANCISCO CA 94111-5894

FOR: California Cogeneration Council

Email: tsolomon@winston.com Status: INFORMATION

CATHERINE POLLINA

WINSTON STRAWN LLP 101 CALIFORNIA ST, STE 3900

SAN FRANCISCO CA 94111-5894
FOR: California Cogeneration Council

Email: cpollina@winston.com

Total number of addressees: 455

LULU WEINZIMER

**CALIFORNIA ENERGY CIRCUIT** 

695 9TH AVE. NO.2

SAN FRANCISCO CA 94118

Email: lisaweinzimer@sbcglobal.net

Status: INFORMATION

**CONTANCE LENI** 

**CALIFORNIA ENERGY COMMISSION** 

1516 NINTH ST., MS-20 SACRAMENTO CA 95814

Email: cleni@energy.state.ca.us

Status: INFORMATION

CLARE LAUFENBERG

**CALIFORNIA ENERGY COMMISSION** 

1516 NINTH ST, MS 46 SACRAMENTO CA 95814

Email: Claufenb@energy.state.ca.us

Status: STATE-SERVICE

LANA WONG

**CALIFORNIA ENERGY COMMISSION** 

1516 NINTH ST., MS-20 SACRAMENTO CA 95814

Email: Iwong@energy.state.ca.us

Status: STATE-SERVICE

MICHAEL JASKE

**CALIFORNIA ENERGY COMMISSION** 

1516 9TH ST, MS-39 SACRAMENTO CA 95814

Email: mjaske@energy.state.ca.us

Status: STATE-SERVICE

**NANCY TRONAAS** 

**CALIFORNIA ENERGY COMMISSION** 

1516 9TH ST. MS-20

SACRAMENTO CA 95814-5512

Status: STATE-SERVICE

KAREN GRIFFIN EXECUTIVE OFFICE CALIFORNIA ENERGY COMMISSION

1516 9TH ST, MS 39 SACRAMENTO CA 95814

FOR: California Energy Commission Email: kgriffin@energy.state.ca.us

Status: STATE-SERVICE

J.A. SAVAGE

**CALIFORNIA ENERGY CIRCUIT** 

3006 SHEFFIELD AVE. OAKLAND CA 94602

Email: editorial@californiaenergycircuit.net

Status: INFORMATION

PIERRE H. DUVAIR, PH.D

**CALIFORNIA ENERGY COMISSION** 

**1516 NINTH ST** 

SACRAMENTO CA 95814

Email: pduvair@energy.state.ca.us

Status: INFORMATION

KEVIN KENNEDY SUPERVISOR, SPECIAL PROJECTS

**CALIFORNIA ENERGY COMMISSION** 

1516 9TH ST, MS-48 SACRAMENTO CA 95814

Status: STATE-SERVICE

MARC PRYOR

**CALIFORNIA ENERGY COMMISSION** 

1516 9TH ST, MS 20 SACRAMENTO CA 95814

Email: mpryor@energy.state.ca.us

Status: STATE-SERVICE

JIM WOODWARD ELECTRICITY SUPPLY ANALYSIS

DIVISION

**CALIFORNIA ENERGY COMMISSION** 

1516 NINTH ST, MS 20

SACRAMENTO CA 95814-5512

Email: jwoodwar@energy.state.ca.us

Status: STATE-SERVICE

BRADLEY MEISTER

**CALIFORNIA ENERGY COMMISSION** 

1516 9TH ST, MS-26

SACRAMENTO CA 95814

FOR: California Energy Commission Email: bmeister@energy.state.ca.us

Status: STATE-SERVICE

ROSS MILLER

CALIFORNIA ENERGY COMMISSION

1516 9TH ST

SACRAMENTO CA 95814

FOR: California Energy Commission Email: rmiller@energy.state.ca.us

Status: STATE-SERVICE

Total number of addressees: 455

DAVID VIDAVER

**CALIFORNIA ENERGY COMMISSION** 

1516 NINTH ST, MS-20

**SACRAMENTO CA 95814-5512** 

FOR: California Energy Commission Email: dvidaver@energy.state.ca.us

Status: STATE-SERVICE

**CALIFORNIA ENERGY MARKETS** 

425 DIVISIDERO ST., STE. 303 SAN FRANCISCO CA 94117-2242

FOR: CALIFORNIA ENERGY MARKETS

Email: cem@newsdata.com Status: INFORMATION

GRANT A. ROSENBLUM STAFF COUNSEL

**CALIFORNIA ISO** 

151 BLUE RAVINE ROAD FOLSOM CA 95630

FOR: California Independent System Operator Corp.

Email: grosenblum@caiso.com

Status: PARTY

LEGAL & REGULATORY DEPARTMENT

**CALIFORNIA ISO** 

151 BLUE RAVINE ROAD FOLSOM CA 95630

Email: e-recipient@caiso.com Status: INFORMATION

SIDNEY DAVIES

CALIFORNIA ISO

151 BLUE RAVINE ROAD FOLSOM CA 95630

Email: sdavies@caiso.com Status: INFORMATION

WILLIAM H. BOOTH ATTORNEY

LAW OFFICES OF WILLIAM H. BOOTH

67 CARR DRIVE

MORAGA CA 94556

FOR: California Large Energy Consumers Association

Email: wbooth@booth-law.com

Status: PARTY

EDWARD W. O'NEILL ATTORNEY

DAVIS WRIGHT TREMAINE LLP

505 MONTGOMERY ST, STE 800 SAN FRANCISCO CA 94111-6533

FOR: California Large Energy Consumers Association

Email: edwardoneill@dwt.com

Status: PARTY

MARY ANN MILLER ELECTRICITY ANALYSIS OFFICE

**CALIFORNIA ENERGY COMMISSION** 

1516 9TH ST, MS 20

SACRAMENTO CA 96814-5512

FOR: CALIFORNIA ENERGY COMMISSION

Email: mmiller@energy.state.ca.us

Status: STATE-SERVICE

KAREN MILLS

**CALIFORNIA FARM BUREAU FEDERATION** 

2300 RIVER PLAZA DRIVE SACRAMENTO CA 95833 Email: kmills@cfbf.com Status: INFORMATION

CALIFORNIA ISO

151 BLUE RAVINE ROAD FOLSOM CA 95630

Email: e-recipient@caiso.com Status: INFORMATION

MIKE D. DOZIER

**CALIFORNIA ISO** 

151 BLUE RAVINE ROAD

FOLSOM CA 95630

Email: mdozier@caiso.com Status: INFORMATION

BALDASSARO DI CAPO, ESQ.

CALIFORNIA ISO

LEGAL AND REGULATORY DEPARTMENT

151 BLUE RAVINE ROAD FOLSOM CA 95630

Email: bdicapo@caiso.com

Status: PARTY

BARBARA R. BARKOVICH

BARKOVICH & YAP, INC.

44810 ROSEWOOD TERRACE

MENDOCINO CA 95460

FOR: California Large Energy Consumers Association

Email: brbarkovich@earthlink.net

Status: PARTY

KEITH R. MCCREA ATTORNEY

SUTHERLAND, ASBILL & BRENNAN, LLP

1275 PENNSYLVANIA AVE., NW WASHINGTON DC 20004-2415

FOR: California Manufacturers & Technology Association

Email: keith.mccrea@sablaw.com

Total number of addressees: 455

SCOTT BLAISING

BRAUN BLAISING MCLAUGHLIN P.C.

915 L ST, STE. 1270 SACRAMENTO CA 95814

FOR: CALIFORNIA MUNICIPAL UTILITIES ASSN.

Email: blaising@braunlegal.com

Status: PARTY

DOUGLAS M. GRANDY, P.E. CALIFORNIA ONSITE

**GENERATION DG TECHNOLOGIES** 1220 MACAULAY CIRCLE CARMICHAEL CA 95608

FOR: California Onsite Generation Email: dgrandy@caonsitegen.com

Status: INFORMATION

SARAH BESERRA

CALIFORNIA REPORTS.COM

39 CASTLE HILL COURT VALLEJO CA 94591

Email: sbeserra@sbcglobal.net

Status: INFORMATION

LYNNE M. BROWN

CALIFORNIANS FOR RENEWABLE ENERGY INC.

24 HARBOR ROAD

SAN FRANCISCO CA 94124 Email: I brown369@yahoo.com

Status: PARTY

**ROBERT SARVEY** 

RACE

501 W. GRANTLINE RD **TRACY CA 95376** 

FOR: CALIFORNIANS FOR RENEWABLE ENERGY, INC.

Email: sarvevbob@aol.com Status: INFORMATION

**KEVIN BOUDREAUX** 

**CALPINE CORPORATION** 717 TEXAS AVE STE 1000 **HOUSTON TX 77002** Status: INFORMATION

**ROSEMARY ANTONOPOULOS CALPINE CORPORATION** 4160 DUBLIN BLVD., STE 100

DUBLIN CA 94568

Email: rantonopoulos@calpine.com

Status: INFORMATION

KAREN LINDH

**CALIFORNIA ONSITE GENERATION** 

7909 WALERGA ROAD, NO. 112, PMB 119

ANTELOPE CA 95843 Email: karen@klindh.com Status: INFORMATION

SUSAN ROSSI ATTORNEY

CALIFORNIA POWER EXCHANGE CORPORATION

201 S LAKE AVE STE 409 PASADENA CA 91101-4807

FOR: CALIFORNIA POWER EXCHANGE

Email: sdrossi@calpx.com

Status: PARTY

NANCY RADER EXECUTIVE DIRECTOR **CALIFORNIA WIND ENERGY ASSOCIATION** 

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000

FOR: California Wind Energy Association

Email: nrader@igc.org Status: PARTY

MAURICE CAMPBELL MEMBER

CALIFORNIANS FOR RENEWABLE ENERGY, INC.

1100 BRUSSELS ST.

SAN FRANCISCO CA 94134

FOR: CALIFORNIANS FOR RENEWABLE ENERGY, INC.

Email: mecsoft@pacbell.net Status: INFORMATION

MICHAEL E. BOYD PRESIDENT

CALIFORNIANS FOR RENEWABLE ENERGY, INC.

5439 SOQUEL DRIVE SOQUEL CA 95073

FOR: CALIFORNIANS FOR RENEWABLE ENERGY, INC.

Email: michaelboyd@sbcglobal.net

Status: PARTY

AVIS KOWALEWSKI

**CALPINE CORPORATION** 4160 DUBLIN BLVD, STE 100

DUBLIN CA 94568

Email: kowalewskia@calpine.com

Status: INFORMATION

ALEXANDER B. MAKLER VP AND MANAGING COUNSEL

CALPINE CORPORATION 4160 DULBIN BLVD., STE 100

DUBLIN CA 94568

FOR: CALPINE CORPORATION Email: alexm@calpine.com

Total number of addressees: 455.

MARTIN HOMEC EMAIL ONLY

EMAIL ONLY CA 00000-0000

FOR: CARE

Email: martinhomec@gmail.com

Status: INFORMATION

MARY LYNCH VP - REGULATORY AND LEGISLATIVE

**AFFAIRS** 

**CONSTELLATION ENERGY COMMODITIES GRP** 

2377 GOLD MEADOW WAY, STE 100

GOLD RIVER CA 95670

FOR: CECG

Email: mary.lynch@constellation.com

Status: PARTY

ANN G. GRIMALDI

MCKENNA LONG & ALDRIDGE LLP

101 CALIFORNIA ST, 41ST FLR SAN FRANCISCO CA 94111

FOR: Center for Energy and Economic Development

Email: agrimaldi@mckennalong.com

Status: PARTY

ANAN H. SOKKER LEGAL ASSISTANT

CHADBOURNE & PARKE LLP 1200 NEW HAMPSHIRE AVE. NW

WASHINGTON DC 20036 Status: INFORMATION

CLAIRE E, TORCHIA

**CHADBOURNE & PARKE LLP** 

350 SOUTH GRAND AVE., STE 3300

LOS ANGELES CA 90071

Email: ctorchia@chadbourne.com

Status: INFORMATION

JEANNE M. SOLE DEPUTY CITY ATTORNEY CITY AND COUNTY OF SAN FRANCISCO

1 DR. CARLTON B. GOODLETT PLACE, RM. 375

SAN FRANCISCO CA 94102-4682 Email: jeanne.sole@sfgov.org Status: INFORMATION

THOMAS BLAIR
CITY OF SAN DIEGO

9601 RIDGEHAVEN COURT, STE, 120/MS11

SAN DIEGO CA 92123 Email: tblair@sandiego.gov Status: INFORMATION PHILLIP W. REESE

CALIFORNIA BIOMASS ENERGY ALLIANCE, LLC

PO BOX 8

3379 SOMIS ROAD SOMIS CA 93066 FOR: CBEA

Email: phil@reesechambers.com

Status: PARTY

TIMOTHY R. ODIL

MCKENNA LONG & ALDRIDGE LLP

1400 WEWATTA ST., STE, 700

DENVER CO 80202

FOR: Center for Energy and Economic Development

Email: todil@mckennalong.com

Status: PARTY

DALE W. MAHON

CENTRAL HYDROELECTRIC CORPORATION

9951 GRANT LINE ROAD ELK GROVE CA 95624-1441

FOR: CENTRAL HYDROELECTRIC CORPORATION

Status: PARTY

ROBERT SHAPIRO

CHADBOURNE & PARKE LLP

1200 NEW HAMPSHIRE AVE. NW

WASHINGTON DC 20036

Email: rshapiro@chadbourne.com

Status: INFORMATION

STEPHEN A. S. MORRISON

**CITY & COUNTY OF SAN FRANCISCO** 

CITY HALL, SUITE 234

1 DR CARLTON B. GOODLET PLACE SAN FRANCISCO CA 94102-4682

FOR: CITY & COUNTY OF SAN FRANCISCO

Status: PARTY

**BOB TANG ASSISTANT DIRECTOR** 

CITY OF AZUSA

729 NORTH **AZU**SA AVE AZUSA CA 91702-9500

Email: btang@ci.azusa.ca.us

Status: INFORMATION

FRITZ ORTLIEB ATTORNEY

**CITY OF SAN DIEGO** 

1200 THIRD AVE, STE 1200 SAN DIEGO CA 92101

FOR: City of San Diego Email: fortlieb@sandiego.gov

Total number of addressees: 455

JANIS C. PEPPER

CLEAN POWER MARKETS, INC.

PO BOX 3206

LOS ALTOS CA 94024

FOR: CLEAN POWER MARKETS, INC. Email: pepper@cleanpowermarkets.com

Status: INFORMATION

MARC D. JOSEPH ATTORNEY

ADAMS, BROADWELL, JOSEPH & CARDOZO

601 GATEWAY BLVD., STE. 1000 SOUTH SAN FRANCISCO CA 94080

FOR: Coalition of California Utility Employees and California

Unions for Reliable Energy

Email: mdjoseph@adamsbroadwell.com

Status: PARTY

DONALD BROOKHYSER

ALCANTAR & KAHL LLP

1300 SW FIFTH AVE, STE 1750

PORTLAND OR 97201

FOR: Cogeneration Association of California

Email: deb@a-klaw.com Status: INFORMATION

LINDA SHERIF ATTORNEY
ALCANTAR & KAHL LLP

33 NEW MONTGOMERY ST, STE 1850

SAN FRANCISCO CA 94015

FOR: COGENERATION ASSOCIATION OF CALIFORNIA

(CAC) and EPUC

Email: lys@a-klaw.com

Status: PARTY

TED POPE DIRECTOR

**COHEN VENTURES, INC./ENERGY SOLUTIONS** 

1610 HARRISON ST. OAKLAND CA 94612

Email: ted@energy-solution.com

Status: INFORMATION

DAVID X. KOLK, PH.D.

COMPLETE ENERGY SERVICE, INC.

41422 MAGNOLIA ST MURRIETA CA 92562

Email: Dkolk@compenergy.com

Status: INFORMATION

ANDREW BROWN ATTORNEY ELLISON & SCHNEIDER, LLP 2600 CAPITOL AVE, STE 400

SACRAMENTO CA 95816-5905

FOR: Constellation Energy Commodities Group, Inc./ Sierra

Pacific Power Company

Email: abb@eslawfirm.com

Status: PARTY

KAREN TERRANOVA ALCANTAR & KAHL, LLP

33 NEW MONTGOMERY ST, STE 1850

SAN FRANCISCO CA 94105

FOR: COALINGA COGENERATION CO.

Email: filings@a-klaw.com Status: INFORMATION

L. JAN REID

COAST ECONOMIC CONSULTING

3185 GROSS ROAD SANTA CRUZ CA 95062

Email: janreid@coastecon.com

Status: PARTY

MICHAEL ALCANTAR ATTORNEY ALCANTAR & KAHL, LLP

1300 SW FIFTH AVE, STE 1750

PORTLAND OR 97201

FOR: Cogeneration Association of California

Email: mpa@a-klaw.com

Status: PARTY

MARSHALL D. CLARK MANAGER

COGENERATION CONTRACT SERVICES PO BOX 989052, MS-408; ORIM RM 1-435 WEST SACRAMENTO CA 95798-9052

Email: Marshall.Clark@dgs.ca.gov

Status: STATE-SERVICE

WILLIAM MITCHELL

COMPETITIVE POWER VENTURES, INC.

55 2ND ST, STE 525

SAN FRANCISCO CA 94105
Email: will.mitchell@cpv.com
Status: INFORMATION

ERIC C. WOYCHIK

STRATEGY INTEGRATION LLC

9901 CALODEN LANE
OAKLAND CA 94605
FOR: Comverge, Inc.
Email: eric@strategyi.com

Clater DADTY

Status: PARTY

CYNTHIA BRADY

CONSTELLATION ENERGY RESOURCES, LLC

EMAIL ONLY

EMAIL ONLY IL 00000-0000

FOR: Constellation Energy Resources, LLC

Email. cynthia.brady@constellation.com

Total number of addressees: 455

MICHAEL D. EVANS SHELL ENERGY AMERICA (US) L.P. 4445 EASTGATE MALL, STE 100

SAN DIEGO CA 92120 FOR: Coral Power L.L.C. Email: michael.evans@shell.com

Status: PARTY

ROGER BERLINER ATTORNEY BERLINER LAW PLLC 700 12TH ST NW, STE 700 WASHINGTON DC 20006

FOR: County of Los Angeles
Email: roger@berlinerlawpllc.com

Status: PARTY

ED J. WHELESS DIVISION ENGINEER COUNTY SANITATION DIST. OF L.A. COUNTY SOLID WASTER MANAGEMENT DEPT PO BOX 4998

WHITTIER CA 90607-7411 Email: ewheless@lacsd.org Status: INFORMATION

PATRICK HOLLEY
COVANTA ENERGY CORPORATION

2829 CHILDRESS DR. ANDERSON CA 96007-3563

FOR: COVANTA ENERGY CORP Email: pholley@covantaenergy.com

Status: INFORMATION

ANDY FRIEDL CP KELCO

2025 E. HARBOR DRIVE SAN DIEGO CA 92113 Status: INFORMATION

CARL K. OSHIRO ATTORNEY CSBRT/CSBA

100 PINE ST, STE 3110 SAN FRANCISCO CA 94111 Email: oshirock@pacbell.net

Status: PARTY

JUDY PAU Davis wright tremaine LLP

EMAIL ONLY

EMAIL ONLY CA 00000-0000 Email: judypau@dwt.com Status: INFORMATION MICHAEL P. ALCANTAR ALCANTAR & KAHL LLP

33 NEW MONTGOMERY ST, STE 1850

SAN FRANCISCO CA 94105

FOR: Counsel to the Cogeneration Association of California

Email: mpa@a-klaw.com

Status: PARTY

HOWARD CHOY

**COUNTY OF LOS ANGELES** 

1100 NORTH EASTERN AVE, RM 300

LOS ANGELES CA 90063
FOR: County of Los Angeles
Email: hchoy@isd.co.la.ca.us

Status: PARTY

KATHERINE MUDGE

COVAD COMMUNICATIONS COMPANY 7000 N. MOPAC EXPRESSWAY, FLR 2

AUSTIN TX 78731

Email: kmudge@covad.com Status: INFORMATION

KAREN HENRY

COVANTA POWER PACIFIC, INC.

PO BOX 278

CROWS LANDING CA 95313 Status: INFORMATION

KATHLEEN ESPOSITO

CRESTED BUTTE CATALYSTS LLC

**PO BOX 668** 

CRESTED BUTTE CO 81224 Status: INFORMATION

JAMES L. MCARTHUR **DAI OILDALE, INC** 3300 MANOR DRIVE

BAKERSFIELD CA 93308
Email: daipm@daioildale.com
Status: INFORMATION

ROBERT B. GEX ATTORNEY, DAVIS WRIGHT TREMAINE LLP 505 MONTGOMERY ST, STE 800 SAN FRANCISCO CA 94111-6533

Email: bobgex@dwt.com Status: INFORMATION

Total number of addressees: 455

STEVEN F. GREENWALD ATTORNEY DAVIS WRIGHT TREMAINE, LLP 505 MONTGOMERY ST, STE 800 SAN FRANCISCO CA 94111-6533 Email: stevegreenwald@dwt.com

Status: INFORMATION

HENRY NANJO ASSISTANT CHIEF COUNSEL, LEGAL

**SERVICES** 

**DEPARTMENT OF GENERAL SERVICES** 

707 3RD ST, STE 7-330 WEST SACRAMENTO CA 95605

Email: Henry.Nanjo@dgs.ca.gov

Status: INFORMATION

CRAIG WILLIAMS

DG FAIRHAVEN POWER, LLC 1660 UNION ST, STE 200 SAN DIEGO CA 92101 Status: INFORMATION

**BO BUCHYNSKY** 

DIAMOND GENERATING CORPORATION

333 SOUTH GRAND AVE., STE 1570

LOS ANGELES CA 90071

Email: b.buchynsky@dgc-us.com

Status: INFORMATION

ANDREA MORRISON

**DIRECT ENERGY SERVICES, LLC** 

415 DIXON ST

ARROYO GRANDE CA 93420

Email: andrea.morrison@directenergy.com

Status: INFORMATION

Noel Obiora

CALIF PUBLIC UTILITIES COMMISSION

LEGAL DIVISION

505 VAN NESS AVE RM 4107 SAN FRANCISCO CA 94102-3214 FOR: Division of Ratepayer Advocates

Email: nao@cpuc.ca.gov

Status: PARTY

JANE E. LUCKHARDT ATTORNEY

DOWNEY BRAND LLP

621CAPITOL MALL, 18TH FLR SACRAMENTO CA 95814

Email: jluckhardt@downeybrand.com

Status: PARTY

RALPH R. NEVIS

DAY CARTER & MURPHY LLP

3620 AMERICAN RIVER DR., STE 205

SACRAMENTO CA 95864

Email: mevis@daycartermurphy.com

Status: INFORMATION

IRYNA KWASNY

**DEPT. OF WATER RESOURCES-CERS DIVISION** 

3310 EL CAMINO AVE., STE.120 SACRAMENTO CA 95821

Email: iryna.kwasny@doj.ca.gov

Status: STATE-SERVICE

DALE E. FREDERICKS

DG POWER INTERNATIONAL LLC

PO BOX 4400

WALNUT CREEK CA 94596-0400

Email: dfredericks@dgpower.com

Status: INFORMATION

GREGORY S.G. KLATT

**DOUGLASS & LIDDELL** 

411 E. HUNTINGTON DRIVE, STE 107-356

ARCADIA CA 91006-8102

FOR: Direct Access Customer Coalition

Email: klatt@energyattorney.com

Status: INFORMATION

STEVEN A. GREENBERG

**DISTRIBUTED ENERGY STRATEGIES** 

4100 ORCHARD CANYON LANE

VACAVILLE CA 95688

FOR: DISTRIBUTED ENERGY STRATEGIES

Email: steveng@destrategies.com

Status: INFORMATION

DON LIDDELL

**DOUGLASS & LIDDELL** 

2928 2ND AVE

SAN DIEGO CA 92103

Email: liddell@energyattomey.com

Status: INFORMATION

MELANIE GILLETTE

**DUKE ENERGY NORTH AMERICA** 

980 NINTH ST, STE 1420 SACRAMENTO CA 95814 Status: INFORMATION

Page 15 of 33

Total number of addressees: 455

JEFFREY GRAY ATTORNEY
DAVIS WRIGHT TREMAINE
505 MONTGOMERY ST, STE 800
SAN FRANCISCO CA 94111-6533
FOR: Dynamis Incorporated

Email: jeffgray@dwt.com

Status: PARTY

LAWRENCE KOSTRZEWA REGIONAL VP, DEVELOPMENT EDISON MISSION ENERGY

18101 VON KARMAN AVE., STE 1700

IRVINE CA 92612-1046

Email: /kostrzewa@edisonmission.com

Status: INFORMATION

CRYSTAL NEEDHAM SENIOR DIRECTOR, COUNSEL EDISON MISSION ENERGY

18101 VON KARMAN AVE, STE 1700

IRVINE CA 92612-1046

Email: cneedham@edisonmission.com

Status: PARTY

STEVE KOERNER SENIOR CONSEL

EL PASO CORPORATION 2 NORTH NEVADA AVE

COLORADO SPRINGS CO 80903 Email: steve.koerner@elpaso.com

Status: INFORMATION

CHASE B. KAPPEL

**ELLISON SCHNEIDER & HARRIS LLP** 

2600 CAPITOL AVE, STE 400 SACRAMENTO CA 95816-5905 Email: cbk@eslawfirm.com Status: INFORMATION

JEFFERY D. HARRIS ATTORNEY ELLISON, SCHNEIDER & HARRIS 2600 CAPITOL AVE, STE 400 SACRAMENTO CA 95816-5905

Email: idh@eslawfirm.com

Status: PARTY

REN ORENS
ENERGY AND ENVIRONMENTAL ECONOMICS

101 MONTGOMERY ST STE 1500 SAN FRANCISCO CA 94104-4133

Email: ren@ethree.com Status: INFORMATION JOSEPH PAUL SENIOR CORPORATE COUNSEL

DYNEGY-WEST GENERATION 1000LOUISIANA ST, STE. 5800

HOUSTON TX 77002 FOR: Dynegy

Email: joe.paul@dynegy.com

Status: PARTY

PHILIP HERRINGTON REGIONAL VP, BUSINESS

MANAGEMENT

EDISON MISSION ENERGY

18101 VON KARMAN AVE, STE 1700

IRVINE CA 92612-1046

Email: pherrington@edisonmission.com

Status: INFORMATION

ANNE FALCON

EES CONSULTING, INC.

570 KIRKLAND AVE KIRLAND WA 98033

Email: rfp@eesconsulting.com Status: INFORMATION

FRED MOBASHERI CONSULTANT ELECTRIC POWER GROUP, LLC

201 SOUTH LAKE AVE., STE 400

PASADENA CA 91101 Email: fmobasheri@aol.com Status: STATE-SERVICE

LYNN HAUG ATTORNEY

**ELLISON, SCHNEIDER & HARRIS, LLP** 

2600 CAPITOL AVE, STE 400 SACRAMENTO CA 95816-5905 Email: Imh@eslawfirm.com Status: INFORMATION

CARLO ZORZOLI

**ENEL NORTH AMERICA, INC.** 

1 TECH DRIVE, STE 220 ANDOVER MA 1810

FOR: ENEL NORTH AMERICA, INC.

Email: carlo.zorzoli@enel.it Status: INFORMATION

SNULLER PRICE

**ENERGY AND ENVIRONMENTAL ECONOMICS** 

101 MONTGOMERY, STE 1600 SAN FRANCISCO CA 94104 Email: snuller@ethree.com Status: STATE-SERVICE

Total number of addressees: 455

KEVIN J. SIMONSEN **ENERGY MANAGEMENT SERVICES** 

646 EAST THIRD AVE DURANGO CO 81301

Email: kisimonsen@ems-ca.com

Status: INFORMATION

**NORA SHERIFF** ALCANTAR & KAHL, LLP

33 NEW MONTGOMERY ST, STE 1850

SAN FRANCISCO CA 94105

FOR: Energy Producers & Users Coalition

Email: nes@a-klaw.com

Status: PARTY

SARA STECK MYERS ATTORNEY LAW OFFICES OF SARA STECK MYERS

122 - 28TH AVE

SAN FRANCISCO CA 94121

FOR: ENRON WIND CORP., CENETER FOR ENERGY

EFFICIENCY AND RENEWABLE TECHNOLOGIES

(CEERT)

Email: ssmyers@att.net

Status: PARTY

SCOTT GOODWIN FAR WEST POWER CORPORATION

3330 CLAYTON ROAD, STE B CONCORD CA 94519

Status: INFORMATION

NORMAN J. FURUTA

**FEDERAL EXECUTIVE AGENCIES** 

1455 MARKET ST., STE 1744 SAN FRANCISCO CA 94103-1399

FOR: Federal Executive Agencies Email: norman.furuta@navy.mil

Status: PARTY

JAMES MCMAHON 29 DANBURY ROAD NASHUA NH 3064

Email: jmcmahon@8760energy.com

Status: INFORMATION

MARK HARRER 56 ST, TIMOTHY CT. DANVILLE CA 94526

Email: mhharrer@sbcglobal.net

Status: INFORMATION

CAROLYN KEHREIN

**ENERGY MANAGEMENT SERVICES** 

2602 CELEBRATION WAY WOODLAND CA 95776

Email: cmkehrein@ems-ca.com

Status: PARTY

**EVELYN KAHL** 

**ALCANTAR & KAHL, LLP** 

33 NEW MONTGOMERY ST, STE 1850

SAN FRANCISCO CA 94015

FOR: ENERGY PRODUCERS AND USERS COALITION

(EPUC)

Email: ek@a-klaw.com

Status: PARTY

GARY M. IZING

ENXCO DEVELOPMENT CORP.

4000 EXECUTIVE PARKWAY, STE. 100

SAN RAMON CA 94583-4381

Email: garyi@enxco.com

Status: INFORMATION

SAEED FARROKHPAY

FEDERAL ENERGY REGULATORY COMMISSION

110 BLUE RAVINE RD., STE 107

FOLSOM CA 95630

Email: saeed.farrokhpay@ferc.gov

Status: INFORMATION

**ED CHANG** 

FLYNN RESOURCE CONSULTANTS, INC.

2165 MOONSTONE CIRCLE EL DORADO HILLS CA 95762 Email: edchang@flynnrci.com

Status: INFORMATION

KENNETH E. ABREU 853 OVERLOOK COURT SAN MATEO CA 94403

Email: k.abreu@sbcglobal.net

Status: INFORMATION

PHILIPPE AUCLAIR 11 RUSSELL COURT

WALNUT CREEK CA 94598

Email: philha@astound.net Status: INFORMATION

Total number of addressees: 455

DOCKET COORDINATOR 5727 KEITH ST. OAKLAND CA 94618

Email: cpucdockets@keyesandfox.com

Status: INFORMATION

DAVID E. MORSE 140 B ST, STE 5-355 **DAVIS CA 95616** 

Email: davidmorse9@gmail.com

Status: INFORMATION

EDWARD C. REMEDIOS 33 TOLEDO WAY SAN FRANCISCO CA 94123-2108 Email: ecrem@ix.netcom.com Status: INFORMATION

ANGELA KIM FTI CONSULTING 1 FRONT ST STE 1600 SAN FRANCISCO CA 94111-5353 FOR: FTI CONSULTING Email: angela.kim@fticonsulting.com

Status: INFORMATION

JOHN C. GABRIELLI **GABRIELLI LAW OFFICE** 430 D ST

**DAVIS CA 95616** 

FOR: GABRIELLI LAW OFFICE Email: gabriellilaw@sbcglobal.net

Status: INFORMATION

JACK MCNAMARA GEO-ENERGY PARTNERS-1983 LTD. PO BOX 1380

AGOURA HILLS CA 91376 Email: jackmack@suesec.com Status: INFORMATION

RONALD MOORE **GOLDEN STATE WATER/BEAR VALLEY ELECTRIC** 

630 EAST FOOTHILL BLVD SAN DIMAS CA 91773

Email: rkmoore@gswater.com Status: INFORMATION

DAVID MARCUS PO BOX 1287 BERKELEY CA 94701

Email: dmarcus2@sbcglobal.net

Status: INFORMATION

KEITH G. JOHNSON SENIOR MARKET AND PRODUCT

DEVELOPER

151 BLUE RAVINE ROAD FOLSOM CA 95682

Email: kjohnson@caiso.com Status: INFORMATION

ROBIN J. WALTHER 1380 OAK CREEK DRIVE, NO. 316 PALO ALTO CA 94304-2016 Email: rwalther@pacbell.net

Status: INFORMATION

**ROCKY HO** FTI CONSULTING 1 FRONT ST STE 1600

SAN FRANCISCO CA 94111-5353

FOR: FTI CONSULTING

Email: rocky.ho@fticonsulting.com

Status: INFORMATION

RON DAHLIN GENERAL MANAGER

CARDINAL COGEN, INC. 288 CAMPUS DRIVE WEST STANFORD CA 94305 FOR: GE Energy Email: ron.dahlin@ge.com

Status: INFORMATION

RICHARD BENEDETTI GEORGIA - PACIFIC CORPORATION

90 W. REDWOOD AVE. FORT BRAGG CA 95437 Status: INFORMATION

ISSER STEVE VICE PRESIDENT/GENERAL COUNSEL

**GOOD COMPANY ASSOCIATES** 816 CONGRESS AVE, STE 1400 **AUSTIN TX 78701** 

Email: sisser@goodcompanyassociates.com

Total number of addressees: 455

STEVE ISSER VP, GENERAL COUNSEL GOOD COMPANY ASSOCIATES 816 CONGRESS AVE., STE 1400

AUSTIN TX 78701

FOR: Good Company Associates

Email: sisser@goodcompanyassociates.com

Status: PARTY

NORMAN A. PEDERSEN HANNA AND MORTON LLP

444 SOUTH FLOWER ST, STE 1500 LOS ANGELES CA 90071-2916 Email: npedersen@hanmor.com

Status: INFORMATION

GREGGORY L. WHEATLAND ATTORNEY ELL**ISON,** SCHNEIDER & HARRIS, LLP

2600 CAPITOL AVE, STE 400 SACRAMENTO CA 95816-5905 FOR: Hercules Municipal Utility Email: glw@eslawfirm.com

Status: PARTY

TAM HUNT HUNT CONSULTING

EMAIL ONLY EMAIL ONLY CA 0

Email: tam.hunt@gmail.com Status: INFORMATION

ALICE LIDDELL
ICF INTERNATIONAL

620 FOLSOM ST, STE, 200 SAN FRANCISCO CA 94107 Email: aliddell@icfi.com

Status: INFORMATION

DAVID OLSEN

IMPERIAL VALLEY STUDY GROUP 3804 PACIFIC COAST HIGHWAY

VENTURA CA 93001 Status: INFORMATION

STEVEN KELLY

INDEPENDENT ENERGY PRODUCERS ASSOCIATION

1215 K ST, STE 900 SACRAMENTO CA 95814

FOR: INDEPENDENT ENERGY PRODUCERS ASSN

Email: steven@iepa.com

Status: PARTY

GREGG MORRIS DIRECTOR GREEN POWER INSTITUTE 2039 SHATTUCK AVE, STE 402

BERKELEY CA 94704 FOR: Green Power Institute Email: gmorns@emf.net

Status: PARTY

RAJ N. PANKHANIA HERCULES MUNICIPAL UTILITY

111 CIVIC DRIVE HERCULES CA 94547

Email: raj.pankhania@ci.hercules.ca.us

Status: INFORMATION

RALPH SANDERS HL POWER COMPANY 732-025 WENDEL ROAD WENDEL CA 96136

Email: rsanders@hlpower.com Status: INFORMATION

HYPOWER, INC

1150 BALLENA BLVD, 220 ALAMEDA CA 94501 FOR: Hypower, Inc.

Email: hypower@pacbell.net

Status: PARTY

KIMBERLY KIENER
IMPERIAL IRRIGATION DISTRICT

504 CATALINA BLVD. SAN DIEGO CA 92106 Email: kmkiener@cox.net Status: INFORMATION

AMBER RIESENHUBER ENERGY ANALYST INDEPENDENT ENERGY PRODUCERS ASSOC.

1215 K ST, STE 900 SACRAMENTO CA 95814 Email: amber@iepa.com Status: INFORMATION

BRIAN T. CRAGG

GOODIN, MACBRIDE, SQUERI, DAY & LAMPREY

505 SANSOME ST, STE 900 SAN FRANCISCO CA 94111

FOR: Independent Energy Producers Association

Email: bcragg@goodinmacbride.com

Total number of addressees: 455

**DOUGLAS K. KERNER ATTORNEY ELLISON, SCHNEIDER & HARRIS, LLP** 

2600 CAPITOL AVE, STE 400 **SACRAMENTO CA 95816-5905** 

FOR: Independent Energy Producers Association

Email: dkk@eslawfirm.com

Status: PARTY

ROBERT FORGIONE INTERNATIONAL POWER TECHNOLOGY, INC.

1042 W. HEDDING ST., STE 100

SAN JOSE CA 95126 Email: r.forgione@intpower.com

Status: INFORMATION

TOM SKUPNJAK **CPG ENERGY** 5211 BIRCH GLEN RICHMOND TX 77469

FOR: Juniper Generation Email: toms@i-cpg.com

Status: PARTY

DENNIS M.P. EHLING ATTORNEY KIRKPATRICK & LOCKHART NICHOLSON GRAHAM

10100 SANTA MONICA BLVD., 7TH FLR

LOS ANGELES CA 90067 Email: dehling@klng.com Status: INFORMATION

**RORY COX** 

RATEPAYERS FOR AFFORDABLE CLEAN ENERGY

251 KEARNY ST, 2ND FLR SAN FRANCISCO CA 94102

FOR: Local Power

Email: rcox@pacificenvironment.org

Status: PARTY

**LOCAL POWER** 22888 HIGHWAY 1

MARSHALL CA 94940-9701

Status: PARTY

JOHN W. LESLIE, ESQ.

LUCE, FORWARD, HAMILTON & SCRIPPS, LLP

**EMAIL ONLY** EMAIL ONLY CA 0

FOR: LUCE, FORWARD, HAMILTON & SCRIPPS, LLP

Email: jleslie@luce.com Status: INFORMATION

ROBERT E. BURT

INSULATION CONTRACTORS ASSN.

3479 ORANGE GROVE AVE., STE. A NORTH HIGHLANDS CA 95660

FOR: Insulation Contractors Association

Status: INFORMATION

JODY S. LONDON

JODY LONDON CONSULTING

PO BOX 3629

OAKLAND CA 94609

Email: jody london\_consulting@earthlink.net

Status: INFORMATION

TONY WAKIM

KENNEDY/JENKS CONSULTANTS

622 FOLSOM ST

SAN FRANCISCO CA 94107 Status: INFORMATION

G. PATRICK STONER PROGRAM DIRECTOR LOCAL GOVERNMENT COMMISSION

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000 Email: pstoner@lgc.org Status: INFORMATION

PAUL FENN **LOCAL POWER** 

35 GROVE ST

SAN FRANCISCO CA 94102 Email: paulfenn@local.org

Status: INFORMATION

LYNNE MACKEY

LS POWER DEVELOPMENT

400 CHESTERFIELD CTR., STE 110

ST. LOUIS MO 63017

Email: Imackev@lspower.com Status: INFORMATION

RICHARD MCCANN PH.D

M.CUBED

2655 PORTAGE BAY, STE 3

**DAVIS CA 95616** 

Email: mccann@umich.edu Status: INFORMATION

Total number of addressees: 455

DAVID L. HUARD

MANATT, PHELPS & PHILLIPS, LLP ONE EMBARCADERO CENTER, STE 2900 SAN FRANCISCO CA 94111-3736

Email: dhuard@manatt.com Status: INFORMATION

SCOTT HATANAKA

MARTINEZ COGEN LIMITED PARTNERSHIP

550 SOLANO WAY MARTINEZ CA 94553 Status: INFORMATION

C. SUSIE BERLIN

MCCARTHY & BERLIN LLP

100 W. SAN FERNANDO ST., STE 501

SAN JOSE CA 95113

Email: sberlin@mccarthylaw.com

Status: INFORMATION

DAN L. CARROLL ATTORNEY DOWNEY BRAND, LLP

621 CAPITOL MALL, 18TH FLR SACRAMENTO CA 95814

FOR: Merced Irrigation District and Modesto Irrigation

Email: dcarroll@downeybrand.com

Status: PARTY

JAMES ROSS

RCS INC.

500 CHESTERFIELD CENTER, STE 320

CHESTERFIELD MO 63017

FOR: Midway Sunset Cogeneration Email: jimross@r-c-s-inc.com

Status: PARTY

SEAN BEATTY

MIRANT CALIFORNIA, LLC

696 WEST 10TH ST PITTSBURG CA 94565

FOR: Mirant California, LLC/Mirant Delta, LLC/Mirant

Potrero, LLC

Email: Sean.Beatty@mirant.com

Status: PARTY

KERRY HATTEVIK DIRECTOR OF REG. AND MARKET

**AFFAIRS** 

**NEXTERA ENERGY** 829 ARLINGTON BLVD.

EL CERRITO CA 94530

FOR: Mirant Corporation, NRG Energy Email: kerry.hattevik@nexteraenergy.com

Status: PARTY

RANDALL W. KEEN

MANATT, PHLEPS & PHILLIPS, LLP

11355 WEST OLYMPICS BLVD. LOS ANGELES CA 90064 Email: pucservice@manatt.com

Status: INFORMATION

BARRY F, MCCARTHY ATTORNEY MCCARTHY & BERLIN, LLP

100 W. SAN FERNANDO ST., STE 501

SAN JOSE CA 95113

Email: bmcc@mccarthylaw.com

Status: INFORMATION

ARTHUR V. O'DONNELL

CALIFORNIA ENERGY MARKETS

9 ROSCOE ST

SAN FRANCISCO CA 94110-5921

FOR: Media

Email: luluw@newsdata.com Status: INFORMATION

ANN L. TROWBRIDGE

DAY CARTER MURPHY LLC

3620 AMERICAN RIVER DRIVE, STE 205

SACRAMENTO CA 95864

FOR: Merced Irrigation District/Sacramento Municipal Utility

District/California Clean DG Coalition

Email: atrowbridge@daycartermurphy.com

Status: PARTY

DOUGLAS MCFARLAN VP, PUBLIC AFFAIRS

MIDWEST GENERATION EME

440 SOUTH LASALLE ST., STE 3500

CHICAGO IL 60605

Email: dmcfarlan@mwgen.com

Status: INFORMATION

LISA A. COTTLE ATTORNEY

WINSTON & STRAWN LLP

101 CALIFORNIA ST, 39TH FLR SAN FRANCISCO CA 94111

FOR: Mirant California, LLC, Mirant Delta, LLC, and Mirant

Potrero,LLC

Email: lcottle@winston.com

Status: PARTY

THOMAS S. KIMBALL

MODESTO IRRIGATION DISTRICT

1231 11TH ST

MODESTO CA 95354

Email: tomk@mid.org

Total number of addressees: 455

THOMAS S KIMBALL MODESTO IRRIGATION DISTRICT

1231 11TH ST

MODESTO CA 95352-4060

FOR: MODESTO IRRIGATION DISTRICT

Email: tomk@mid.org Status: INFORMATION

MICHAEL A. YUFFEE

MCDERMOTT WILL & EMERY LLP

600 THIRTEENTH ST. NW WASHINGTON DC 20005-3096

FOR: Morgan Stanley Capital Group Inc.

Email: mvuffee@mwe.com Status: INFORMATION

DOUGLAS R. KIVIAT EXECUTIVE DIRECTOR MORGAN STANLEY / COMMODITIES

2000 WESTCHESTER AVE PURCHASE NY 10577

Email: doug.kiviat@morganstanley.com

Status: INFORMATION

PETER W. HANSCHEN ATTORNEY MORRISON & FOERSTER, LLP

101 YGNACIO VALLEY ROAD, STE 450

WALNUT CREEK CA 94596 Email: phanschen@mofo.com Status: INFORMATION

JOHN DUTCHER **MOUNTAIN UTILITIES** 3210 CORTE VALENCIA **FAIRFIELD CA 94534-7875** 

FOR: MOUNTAIN UTILITIES Email: ralf1241a@cs.com Status: INFORMATION

DAVID HOWARTH

MRW & ASSOCIATES, LLC 1814 FRANKLIN ST, STE 720 OAKLAND CA 94612

Email: mrw@mrwassoc.com Status: INFORMATION

JANINE L. SCANCARELLI ATTORNEY

**CROWELL & MORING LLP** 275 BATTERY ST, 23RD FLR SAN FRANCISCO CA 94111 FOR: NATIONAL GRID USA

Email: jscancarelli@crowell.com

Status: INFORMATION

JOY A. WARREN REGULATORY ADMINISTRATOR MODESTO IRRIGATION DISTRICT

1231 11TH ST

MODESTO CA 95354

FOR: Modesto Irrigation District

Email: joyw@mid.org Status: PARTY

GRACE C. WUNG

MCDERMOTT WILL & EMERY LLP

28 STATE ST **BOSTON MA 2109** 

FOR: Morgan Stanley Capital Group, Inc.

Email: gwung@mwe.com Status: INFORMATION

STEVEN HUHMAN

MORGAN STANLEY CAPITAL GROUP INC.

2000 WESTCHESTER AVE **PURCHASE NY 10577** 

Email: steven.huhman@morganstanley.com

Status: INFORMATION

WAYNE AMER PRESIDENT **MOUNTAIN UTILITIES (906)** 

**PO BOX 205** 

KIRKWOOD CA 95646

Email: wamer@kirkwood.com Status: INFORMATION

MRW & ASSOCIATES, LLC

**EMAIL ONLY** 

EMAIL ONLY CA 0

Email: mrw@mrwassoc.com Status: INFORMATION

WILLIAM A. MONSEN

MRW & ASSOCIATES, LLC 1814 FRANKLIN ST, STE 720

OAKLAND CA 94612

Email: mrw@mrwassoc.com Status: INFORMATION

**NOAH LONG** 

NATURAL RESOURCES DEFENSE COUNCIL

111 SUTTER ST, 20TH FLR SAN FRANCISCO CA 94104 Email: nlong@nrdc.org

Total number of addressees: 455

KAY DAVOODI ACQ UTILITY RATES AND STUDIES

NAVAL FACILITIES ENGINEERING COMMAND HQ

1322 PATTERSON AVE., SE - BLDG 33 WASHINGTON DC 20374-5018 Email: khojasteh.davoodi@navy.mil

Status: INFORMATION

KENNY SWAIN

**NAVIGANT CONSULTING** 

3100 ZINFANDEL DRIVE, STE 600 RANCHO CORDOVA CA 95670

Email: kenneth.swain@navigantconsulting.com

Status: INFORMATION

JENNIFER BARNES

NAVIGANT CONSULTING, INC.

**EMAIL ONLY** 

EMAIL ONLY CA 00000-0000

Email: Jennifer.Barnes@Navigantconsulting.com

Status: INFORMATION

JULIE L. MARTIN

**NORTH AMERICA GAS AND POWER** 

BP ENERGY COMPANY
501 WESTLAKE PARK BLVD.
HOUSTON TX 77079
Empile in the martin@ha.com

Email: julie.martin@bp.com Status: INFORMATION

RICHARD D. ELY DAVIS HYDRO

27264 MEADOWBROOK DRIVE

**DAVIS CA 95618** 

FOR: Northern California Small Hydro Assn.

Email: Dick@DavisHydro.com

Status: PARTY

JESUS ARREDONDO NRG ENERGY INC. 4600 CARLSBAD BLVD. CARLSBAD CA 99208

Email: jesus.arredondo@nrgenergy.com

Status: INFORMATION

KATHRYN WIG NRG ENERGY, INC.

EMAIL ONLY CA 0

Email: Kathryn.Wig@nrgenergy.com

Status: PARTY

TIMEA ZENTAI

**NAVIGANT CONSULTING** 

1990 NORTH CALIFORNIA AVE., STE 700

WALNUT CREEK CA 94596

Email: timea.Zentai@navigantconsulting.com

Status: INFORMATION

KIRBY DUSEL

NAVIGANT CONSULTING, INC.

3100 ZINFANDEL DRIVE, STE 600 RANCHO CORDOVA CA 95670

Email: kdusel@navigantconsulting.com

Status: INFORMATION

PAUL D. MAXWELL

NAVIGANT CONSULTING, INC.

3100 ZINFANDEL DRIVE, STE 600 RANCHO CORDOVA CA 95670-6078

Email: pmaxwell@navigantconsulting.com

Status: INFORMATION

SCOTT TOMASHEFSKY

**NORTHERN CALIFORNIA POWER AGENCY** 

651 COMMERCE DRIVE ROSEVILLE CA 95678-6420

Email: scott.tomashefsky@ncpa.com

Status: INFORMATION

LARA ETTENSON

NATURAL RESOURCES DEFENSE COUNCIL

111 SUTTER ST, 20TH FLR SAN FRANCISCO CA 94104

FOR: NRDC

Email: lettenson@nrdc.org

Status: PARTY

E. JESUS ARREDONDO DIRECTOR, REGULATORY AND

GOVERNMENTAL NRG ENERGY, INC. 3741 GRESHAM LANE SACRAMENTO CA 95835 FOR: NRG ENERGY, INC.

Email: jesus.arredondo@nrgenergy.com

Status: INFORMATION

MIKE TIERNEY
NRG ENERGY

1817 ASTON AVE, STE 104 CARLSBAD CA 92008

Email: mike.tierney@nrgenergy.com

Total number of addressees: 455

CHRISTOPHER C. O'HARA ASSISTANT GENERAL

COUNSEL-REGULATORY

**NRG ENERGY** 

211 CARNEGIE CENTER DRIVE

PRINCETON NJ 8540

Email: chris.ohara@nrgenergy.com

Status: PARTY

CHRISTOPHER A. HILEN

**NV ENERGY** 

6100 NEIL ROAD, MS A35

**RENO NV 89511** 

Email: chilen@NVEnergy.com

Status: INFORMATION

BILL JULIAN

OFFICE OF STATE SENATOR MARTHA ESCUTIA

STATE CAPITOL, RM 5080 SACRAMENTO CA 95814 Email: billiulian@sbcqlobal.net

Status: STATE-SERVICE

JONATHAN JACOBS

PA CONSULTING GROUP

75 NOVA DRIVE

PIEDMONT CA 94610-1037

Email: jon.jacobs@paconsulting.com

Status: INFORMATION

MICHAEL GREEN PLANT MANAGER

PACIFIC LUMBER COMPANY 125 MAIN ST

SCOTIA CA 95565

FOR: PACIFIC LUMBER COMPANY

Email: mgreen@palco.com

Status: PARTY

CATHIE ALLEN DIR., REGULATORY AFFAIRS

**PACIFICORP** 

825 NE MULTNOMAH ST, STE 2000

PORTLAND OR 97232

Email: californiadockets@pacificorp.com

Status: INFORMATION

CINDY L. CASSELMAN

PILOT POWER GROUP, INC. (1365)

EMAIL ONLY
EMAIL ONLY CA 0

Email: ccasselman@pilotpowergroup.com

Status: INFORMATION

TARYN CIARDELLA SR. LEGAL SECRETARY

NV ENERGY EMAIL ONLY

EMAIL ONLY NV 0

Email: tciardella@nvenergy.com

Status: INFORMATION

E.J. WRIGHT

OCCIDENTAL POWER SERVICES, INC.

111 WEST OCEAN BLVD LONG BEACH TX 90802 Email: ej\_wright@oxy.com Status: INFORMATION

ATTN.: BUSINESS MANAGER

OGDEN POWER PACIFIC, INC.

2829 CHILDRESS DR. ANDERSON CA 96007 Status: INFORMATION

DON WOOD

PACIFIC ENERGY POLICY CENTER

4539 LEE AVE LA MESA CA 91941 Email: dwood8@cox.net Status: INFORMATION

JORDAN A. WHITE SENIOR ATTORNEY

**PACIFICORP** 

1407 W. NORTH TEMPLE, STE 320

SALT LAKE CITY UT 84116

FOR: PacifiCorp

Email: jordan.white@pacificorp.com

Status: PARTY

WESLEY M. SPOWHN

PILLSBURY WINTHROP SHAW PITTMAN LLP

50 FREMONT ST

SAN FRANCISCO CA 94105

FOR: PILLSBURY WINTHROP SHAW PITTMAN LLP

Email: wesley.spowhn@pillsburylaw.com

Status: PARTY

THOMAS R. DARTON

**PILOT POWER GROUP, INC. (1365)** 

EMAIL ONLY EMAIL ONLY CA 0

Email: tdarton@pilotpowergroup.com

Total number of addressees: 455

EDWARD J. TIEDEMANN ATTORNEY KRONICK, MOSKOVITZ, TIEDEMANN & GIRARD

400 CAPITOL MALL, 27TH FLR SACRAMENTO CA 95814-4416

FOR: PLACER COUNTY WATER AGENCY

Email: etiedemann@kmtg.com Status: INFORMATION

DAVID TATEOSIAN POWER ENGINEERS PO BOX 2037

PO BOX 2037 MARTINEZ CA 94553

Email: dtateosian@powereng.com

Status: INFORMATION

WILLIAM E. POWERS POWERS ENGINEERING 4452 PARK BLVD., STE. 209 SAN DIEGO CA 92116

FOR: POWERS ENGINEERING Email: bpowers@powersengineering.com

Status: INFORMATION

RICK C. NOGER PRAXAIR, INC. (1370)

2430 CAMINO RAMON DRIVE, STE. 300

SAN RAMON CA 94583

Email: rick\_noger@praxair.com

Status: INFORMATION

ANDREW HOERNER REDEFINING PROGRESS 1904 FRANKLIN ST, 6TH FLR OAKLAND CA 94612

Email: hoerner@redefiningprogress.org

Status: PARTY

MONICA SCHWEBS BINGHAM MCCUTCHEN LLP THREE EMBARCADERO CENTER SAN FRANCISCO CA 94111-4067

FOR: Representing Ripon Cogeneration LLC Email: Monica.Schwebs@bingham.com

Status: PARTY

PAULETTE HEUER VICE PRESIDENT-OPERATIONS RIPON COGENERATION LLC

8500 CYPRESSWOOD DRIVE, STE 202

SPRING TX 77379

Email: pheuer-cv@comcast.net Status: INFORMATION

LISA WEINZIMER
PLATTS MCGRAW-HILL
695 NINTH AVE, NO. 2
SAN FRANCISCO CA 94118
Email: lisa weinzimer@platts.com

Status: INFORMATION

JAMES D. SQUERI

GOODIN, MACBRIDE, SQUERI, RITCHIE & DAY

505 SANSOME ST, STE 900 SAN FRANCISCO CA 94111 FOR: Powerex Corp.

Email: jsqueri@gmssr.com

Status: PARTY

RICK NOGER PRAXAIR, INC.

2430 CAMINO RAMON DRIVE, STE. 300

SAN RAMON CA 94583

FOR: PRAXAIR PLAINFIELD, INC. Email: rick\_noger@praxair.com

Status: PARTY

DONALD SCHOENBECK

RCS, INC.

900 WASHINGTON ST, STE 780 VANCOUVER WA 98660 Email: dws@r-c-s-inc.com

Status: INFORMATION

TIM HEMIG

NRG ENERGY, INC.

1817 ASTON AVE, STE 104 CARLSBAD CA 92008

FOR: REGIONAL ENVIRONMENTAL BUSINESS NRG

**ENERGY** 

Email: tim.hemig@nrgenergy.com

Status: INFORMATION

DANIEL V. GULINO

RIDGEWOOD POWER MANAGEMENT, LLC

947 LINWOOD AVE RIDGEWOOD NJ 7450

FOR: RIDGEWOOD POWER MANAGEMENT, LLC

Email: dgulino@ridgewoodpower.com

Status: INFORMATION

TRENT A. CARLSON RRI ENERGY, INC. 1000 MAIN ST HOUSTON TX 77001

FOR: RRI Energy, Inc.

Email: tcarlson@rrienergy.com

Total number of addressees: 455

ROBERT OTT RRI ENERGY, INC 1000 MAIN ST

HOUSTON TX 77002 Email: rott@rrienergy.com Status: INFORMATION

WAYNE TOMLINSON RUBY PIPELINE, LLC

2 NORTH NEVADA AVE, 14TH FLR COLORADO SPRINGS CO 80903 Email: william.tomlinson@elpaso.com

Status: INFORMATION

SUSAN FREEDMAN SENIOR REGIONAL ENERGY

PLANNER

SAN DIEGO ASSOCIATION OF GOVERNMENTS

401 B ST, STE 800 SAN DIEGO CA 92101 Email: sfr@sandag.org Status: INFORMATION

KELLY M. MORTON SAN DIEGO GAS & ELECTRIC

101 ASH ST

SAN DIEGO CA 92123

FOR: SAN DIEGO GAS & ELECTRIC Email: KMorton@SempraUtilities.com

Status: PARTY

GEORGETTA J. BAKER SAN DIEGO GAS & ELECTRIC/SOCAL GAS

101 ASH ST, HQ 13 SAN DIEGO CA 92101

FOR: San Diego Gas & Electric Company and Southern

California Gas Company

Email: GBaker@SempraUtilities.com

Status: PARTY

CINDY ZAMMIT SAN DIEGO GAS & ELECTRIC CO.

EMAIL ONLY
EMAIL ONLY CA 0

Email: CZammit@SempraUtilities.com

Status: INFORMATION

DEAN A. KINPORTS SAN DIEGO GAS & ELECTRIC COMPANY 8306 CENTURY PARK COURT CP32D

SAN DIEGO CA 92123

Email: DAKinports@SempraUtilities.com

Status: INFORMATION

ERIC LEUZE
RRI ENERGY, INC
4174 RIVA RIDGE DRIVE
FAIR OAKS CA 95628
Empil: clause@riangery.com

Email: eleuze@rrienergy.com Status: INFORMATION

VIKKI WOOD

SACRAMENTO MUNICIPAL UTILITY DISTRICT

6301 S ST, MS A204

SACRAMENTO CA 95817-1899 Email: vwood@smud.org Status: INFORMATION

LISA URICK

SAN DIEGO GAS & ELECTRIC COMPANY

101 ASH ST, HQ-12B SAN DIEGO CA 92101

FOR: San Diego Gas & Electric Email: LUrick@SempraUtilities.com

Status: PARTY

JOHN A. PACHECO

SAN DIEGO GAS & ELECTRIC COMPANY

101 ASH ST, HQ12B SAN DIEGO CA 92101-3017 FOR: San Diego Gas & Electric Email: JPacheco@SempraUtilities.com

Status: PARTY

PEDRO VILLEGAS

SAN DIEGO GAS & ELECTRIC/ SO. CAL. GAS

EMAIL ONLY CA 0

FOR: SAN DIEGO GAS & ELECTRIC/ SO. CAL. GAS

Email: PVillegas@SempraUtilities.com

Status: INFORMATION

ALLEN K. TRIAL

SAN DIEGO GAS & ELECTRIC COMPANY

101 ASH ST, HQ-12B SAN DIEGO CA 92101

Email: ATrial@SempraUtilities.com

Status: INFORMATION

GINA M. DIXON

SAN DIEGO GAS & ELECTRIC COMPANY 8330 CENTURY PARK COURT, MS CP32D

SAN DIEGO CA 92123

Email: GDixon@SempraUtilities.com

Total number of addressees: 455

JOSEPH R. KLOBERDANZ SAN DIEGO GAS & ELECTRIC COMPANY 8326 CENTURY PARK COURT

SAN DIEGO CA 92123

Email: JKloberdanz@SempraUtilities.com

Status: INFORMATION

STEVE RAHON

SAN DIEGO GAS & ELECTRIC COMPANY

8330 CENTURY PARK COURT, CP32C

SAN DIEGO CA 92123-1548

Email: LSchavrien@SempraUtilities.com

Status: INFORMATION

KATHLEEN CORDOVA

SAN DIEGO GAS AND ELECTRIC COMPANY

8330 CENTURY PARK CT CP32D

SAN DIEGO CA 92123

Email: KCordova@SempraUtilities.com

Status: INFORMATION

KURT KAMMERER DIRECTOR OF PROGRAMS SAN DIEGO REGIONAL ENERGY OFFICE

PO BOX 60738

SAN DIEGO CA 92166-8738 Email: kjk@kjkammerer.com Status: INFORMATION

JIM HENDRY

SAN FRANCISCO PUBLIC UTILITIES COMM.

1155 MARKET ST, 4TH FLR SAN FRANCISCO CA 94103 Email: jhendry@sfwater.org

Status: PARTY

PHILLIP J. MULLER SCD ENERGY SOLUTIONS 436 NOVA ALBION WAY SAN RAFAEL CA 94903 Email: philm@scdenergy.com

Status: INFORMATION

DIANE I. FELLMAN DIRECTOR, REGULATORY & MARKET

AFFAIRS
NRG WEST
73 DOWNEY ST
SAN FRANCISCO CA 94117

FOR: SELF

Email: Diane.Fellman@nrgenergy.com

Status: PARTY

JOY C. YAMAGATA

SAN DIEGO GAS & ELECTRIC/SOCALGAS

8330 CENTURY PARK COURT, CP 32 D

SAN DIEGO CA 92123-1533

Email: JYamagata@SempraUtilities.com

Status: INFORMATION

WENDY KEILANI

SAN DIEGO GAS & ELECTRIC

8330 CENTURY PARK COURT, CP32D

SAN DIEGO CA 92123

Email: WKeilani@SempraUtilities.com

Status: PARTY

**DESPINA NIEHAUS** 

SAN DIEGO GAS AND ELECTRIC COMPANY

8330 CENTURY PARK COURT-CP32H

SAN DIEGO CA 92123-1530

Email: DNiehaus@SempraUtilities.com

Status: INFORMATION

SANDRA ROVETTI REGULATORY AFFAIRS MANAGER

SAN FRANCISCO PUC 1155 MARKET ST, 4TH FLR SAN FRANCISCO CA 94103 Email: srovetti@sfwater.org

Status: INFORMATION

**ELENA STOIAN** 

SAN JOSE STATE UNIVERSTIY

ONE WASHINGTON SQUARE SAN JOSE CA 95192

Status: INFORMATION

CENTRAL FILES

SDG&E AND SOCALGAS

8330 CENTURY PARK COURT, CP31-E

SAN DIEGO CA 92123-1550

Email: CentralFiles@SempraUtilities.com

Status: PARTY

CLIFF ROCHLIN

SOUTHERN CALIFORNIA GAS COMPANY

555 W. FIFTH ST, ML 22A1 LOS ANGELES CA 90013 FOR: SEMPRA ENERGY Email: crochlin@socalgas.com

Total number of addressees: 455

CARLOS F. PENA SEMPRA ENERGY 101 ASH ST, HQ12 SAN DIEGO CA 92101

Email: CFPena@SempraUtilities.com

Status: INFORMATION

STEVEN C. NELSON SEMPRA ENERGY 101 ASH ST HQ-12

SAN DIEGO CA 92101-3017

Email: SNelson@SempraUtilities.com

Status: INFORMATION

THEODORE ROBERTS SEMPRA BROADBAND 101 ASH ST, HQ 13 SAN DIEGO CA 92101-3017

FOR: Sempra Global

Email: TRoberts@SempraUtilities.com

Status: PARTY

RICHARD M. ESTEVES

SESCO, INC.

77 YACHT CLUB DRIVE, STE 1000 LAKE HOPATCONG NJ 7849

FOR: SESCO INC. Email: sesco@optonline.net Status: INFORMATION

OSA L. WOLFF ATTORNEY
SHUTE, MIHALY & WEINBERGER, LLC

396 HAYES ST

SAN FRANCISCO CA 94102 Email: wolff@smwlaw.com Status: INFORMATION

TREVOR DILLARD RAE REGULATORY RELATIONS SIERRA PACIFIC POWER COMPANY

6100 NEAL ROAD, MS \$4A50 / PO BOX 10100

RENO NV 89520-0024 Email: tdillard@sppc.com Status: INFORMATION

ELENA MELLO

SIERRA PACIFIC POWER COMPANY

6100 NEIL ROAD RENO NV 89520

FOR: SIERRA PACIFIC POWER COMPANY (SPPC)

Email: emello@sppc.com

Status: PARTY

KEITH W. MELVILLE SEMPRA ENERGY 101 ASH ST, HQ 12 SAN DIEGO CA 92101

Email: KMelville@SempraUtilities.com

Status: INFORMATION

DANIEL A. KING ATTORNEY
SEMPRA ENERGY RESOURCES

101 ASH ST

SAN DIEGO CA 92101 FOR: Sempra GLobal

Email: DAKing@SempraUtilities.com

Status: PARTY

WILLIAM TOBIN SEMPRA GLOBAL 101 ASH ST, HQ08C SAN DIEGO CA 92101

Email: WTobin@SempraGlobal.com

Status: INFORMATION

MARCIE MILNER DIRECTOR - REGULATORY AFFAIRS

SHELL TRADING GAS & POWER COMPANY

4445 EASTGATE MALL, STE 100

SAN DIEGO CA 92121

Email: marcie.milner@shell.com

Status: INFORMATION

CHRISTOPHER HILEN ASSISTANT GENERAL COUNSEL

SIERRA PACIFIC POWER COMPANY 6100 NEIL ROAD, PO BOX 10100

RENO NV 89520 Email: chilen@sppc.com

Status: INFORMATION

CATHY ENGEBRETSON SIERRA PACIFIC INDUSTRIES

PO BOX 496014

REDDING CA 96014-6014

FOR: SIERRA PACIFIC INDUSTRIES

Status: PARTY

**BOB HINES** 

SILICON VALLEY LEADERSHIP GROUP

224 AIRPORT PARKWAY, STE 620

SAN JOSE CA 95110 Email: bhines@svlg.net Status: INFORMATION

Total number of addressees: 455

JOHN R. REDDING

ARCTURUS ENERGY CONSULTING, INC.

44810 ROSEWOOD TERRACE MENDOCINO CA 95460-9525

FOR: Silicon Valley Leadership Group Email: johnrredding@earthlink.net

Status: PARTY

DAN WOODS
WHITE & CASE LLP

633 WEST FIFTH ST, STE 1900 LOS ANGELES CA 90071-2007

FOR: Smurfit Stone Container Corporation,/Delta Power

Co./Willamette Industries, Inc./E.F.Oxnard, Inc.

Email: dwoods@whitecase.com

Status: PARTY

ROBERT BEACH

SONOMA COUNTY WATER AGENCY

1022 SIERRA VISTA CT.

GARDNERVILLE NY 89460-8686

Status: PARTY

SALLE E. YOO ATTORNEY DAVIS WRIGHT TREMAINE

505 MONTGOMERY ST, STE 800 SAN FRANCISCO CA 94111-6533

FOR: South San Joaquin Irrigation District

Email: salleyoo@dwt.com

Status: PARTY

WILLIAM BLATTNER

SOUTHERN CALIF GAS CO & SDG&E CO

EMAIL ONLY EMAIL ONLY CA 0

Email: WBlattner@SempraUtilities.com

Status: INFORMATION

TORY S. WEBER

SOUTHERN CALIFORNIA EDISON COMPANY

6042 N. IRWINDALE AVE, STE A

IRWINDALE CA 91702 Email: tory.weber@sce.com Status: INFORMATION

AMBER WYATT

SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770 Email: amber.wyatt@sce.com Status: INFORMATION JOSEPH M. KARP ATTORNEY
WINSTON & STRAWN, LLP
101 CALLEORNIA ST. 30TH ELP

101 CALIFORNIA ST, 39TH FLR SAN FRANCISCO CA 94111-5894

FOR: Smurfit Stone COntainer Corporation, f.k.a.Jefferson

Smurfit Corporation(U.S.)

Email: jkarp@winston.com

Status: PARTY

DAVID SAUL

PACIFIC SOLAR & POWER CORPORATION

2850 W. HORIZON RIDGE PKWY, STE 200

HENDERSON NV 89052 FOR: SOLEL, INC.

Email: dsaul@pacificsolar.net Status: INFORMATION

JEFFREY SHIELDS

SOUTH SAN JOAQUIN IRRIGATION DISTRICT

PO BOX 747 RIPON CA 95366

Email: jshields@ssjid.com

Status: PARTY

JEANNE ARMSTRONG ATTORNEY

GOODIN, MACBRIDE, SQUERI, RITCHIE & DAY

505 SANSOME ST, STE 900 SAN FRANCISCO CA 94111

FOR: South San Joaquin Irrigation District/RRI Energy, Inc.

Email: jarmstrong@gmssr.com

Status: PARTY

**HUGH YAO** 

SOUTHERN CALIFORNIA GAS COMPANY

EMAIL ONLY EMAIL ONLY CA 0

Email: HYao@SempraUtilities.com

Status: INFORMATION

AMBER DEAN WYATT

SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE., PO BOX 800

ROSEMEAD CA 91770 Email: amber.wyatt@sce.com Status: INFORMATION

CATHY A. KARLSTAD

SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE. ROSEMEAD CA 91770 Email: cathy.karlstad@sce.com

Total number of addressees: 455

GARY L. ALLEN SOUTHERN CALIFORNIA EDISON 2244 WALNUT GROVE AVE

ROSEMEAD CA 91770 Email: gary.allen@sce.com Status: INFORMATION

LEON BASS SENIOR ATTORNEY SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770 Email: leon.bass@sce.com Status: INFORMATION

BRUCE FOSTER

SOUTHERN CALIFORNIA EDISON COMPANY

601 VAN NESS AVE, STE. 2040 SAN FRANCISCO CA 94102 Email: bruce.foster@sce.com Status: INFORMATION

JANET COMBS SR. ATTORNEY
SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770 Email: janet.combs@sce.com

Status: PARTY

ROBERT KEELER SR. ATTORNEY SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770

FOR: Southern California Edison

Status: PARTY

CASE ADMINISTRATION SOUTHERN CALIFORNIA EDISON COMPANY PO BOX 800 / 2244 WALNUT GROVE AVE.

ROSEMEAD CA 91770

FOR: SOUTHERN CALIFORNIA EDISON COMPANY

Email: Case.Admin@sce.com Status: INFORMATION

LAURA GENAO SOUTHERN CALIFORNIA EDISON COMPANY

LAW DEPARTMENT 2244 WALNUT GROVE AVE. ROSEMEAD CA 91770

FOR: Southern California Edison Company

Email: laura.genao@sce.com Status: INFORMATION KAREN I. LEE

SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770 Email: karen.lee@sce.com Status: INFORMATION

**ROBYN NARAMORE** 

**SOUTHERN CALIFORNIA EDISON** 

2244 WALNUT GROVE AVE ROSEMEAD CA 91770

Email: robyn.naramore@sce.com

Status: INFORMATION

J. ERIC ISKEN

SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770 Email: j.eric.isken@sce.com

Status: PARTY

MICHAEL A. BACKSTROM ATTORNEY SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91770

FOR: Southern California Edison Email: michael.backstrom@sce.com

Status: PARTY

BETH A. FOX ATTORNEY

SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVE, GO1, RM 351C

ROSEMEAD CA 91770

FOR: Southern California Edison Company

Email: beth.fox@sce.com Status: INFORMATION

JANET S. COMBS

SOUTHERN CALIFORNIA EDISON COMPANY

PO BOX 800 2244 WALNUT GROVE AVE

ROSEMEAD CA 91770

FOR: Southern California Edison company

Email: janet.combs@sce.com Status: INFORMATION

MICHAEL D. MONTOYA

SOUTHERN CALIFORNIA EDISON COMPANY 2244 WALNUT GROVE AVE, PO BOX 800

ROSEMEAD CA 91770

FOR: Southern California Edison Company

Email: mike.montoya@sce.com

Total number of addressees: 455

CAROL SCHMID-FRAZEE ATTORNEY
SOUTHERN CALIFORNIA EDISON COMPANY

2244 WALNUT GROVE AVE ROSEMEAD CA 91765

FOR: Southern California Edison Company

Email: carol.schmid-fraee@sce.com

Status: PARTY

DEANA MICHELLE NG SOUTHERN CALIFORNIA EDISON CO.

2244 WALNUT GROVE AVE ROSEMEAD CA 91770

FOR: Southern California Edison Company

Email: deana.ng@sce.com

Status: PARTY

JANICE LIN MANAGING PARTNER STRATEGEN CONSULTING LLC

EMAIL ONLY EMAIL ONLY CA 0

Email: janice@strategenconsulting.com

Status: INFORMATION

JOSEPH GRECO
TERRA-GEN POWER LLC

9590 PROTOTYPE COURT, STE 200

**RENO NV 89521** 

Email: jgreco@terra-genpower.com

Status: PARTY

CAROL A. SMOOTS PERKINS COIE LLP

607 FOURTEENTH ST, NW, STE 800

WASHINGTON DC 20005

FOR: THELEN REID & PRIEST LLP Email: csmoots@perkinscoie.com

Status: INFORMATION

STEVE BOYD
TURLOCK IRRIGATION DISTRICT

333 EAST CANAL DRIVE TURLOCK CA 95381-0949 Email: seboyd@tid.org

Status: INFORMATION

WILLIAM B. MARCUS JBS ENERGY, INC. 311 D ST, STE A

WEST SACRAMENTO CA 95608

FOR: TURN

Email: bill@jbsenergy.com

Status: PARTY

CAROL A. SCHMID-FRAZEE ATTORNEY SOUTHERN CALIFORNIA EDISON COMPANY

PO BOX 800 2244 WALNUT GROVE AVE

ROSEMEAD CA 91770

FOR: SOUTHERN CALIFORNIA EDISON COMPANY

Email: carol.schmidfrazee@sce.com

Status: PARTY

ATTN.: DIRECTOR FACILITIES SERVICES

SRI INTERNATIONAL 333 RAVENSWOOD AVE MENLO PARK CA 94025 Status: INFORMATION

NEIL BURGESS EXECUTIVE DIRECTOR SYCAMORE COGENERATION COMPANY

PO BOX 80598

BAKERSFIELD CA 93380

Email: neburgess@sycamore.com

Status: INFORMATION

ROBERT FINKELSTEIN
THE UTILITY REFORM NETWORK

115 SANSOME ST, STE 900 SAN FRANCISCO CA 94104

FOR: THE UTILITY REFORM NETOWRK (TURN)

Email: bfinkelstein@turn.org

Status: PARTY

STEVE FELTE GENERAL MANAGER
TRI-DAM PROJECT & POWER AUTHORITY

PO BOX 1158

PINECREST CA 95364-0158

FOR: TRI-DAM POWER AUTHORITY

Status: PARTY

MICHEL PETER FLORIO

THE UTILITY REFORM NETWORK

115 SANSOME ST, STE 900 SAN FRANCISCO CA 94104

FOR: TURN

Email: mflorio@turn.org

Status: PARTY

PATRICK MCDONNELL AGLAND ENERGY 2000 NICASIO VALLEY

NICASIO CA 94946

FOR: TXU ENERGY SERVICES Email: pcmcdonnell@earthlink.net

Total number of addressees: 455

JULIEN DUMOULIN-SMITH ASSOCIATE ANALYST

**UBS INVESTMENT RESEARCH** 1285 AVE OF THE AMERICAS NEW YORK NY 10019

Email: julien.dumoulin-smith@ubs.com

Status: INFORMATION

CARLA PETERMAN

UCEL

2547 CHANNING WAY BERKELEY CA 94720

Email: carla.peterman@grnail.com

Status: INFORMATION

CLYDE MURLEY

CONSULTANT TO NRDC

1031 ORDWAY ST ALBANY CA 94706

FOR: Union of Concerned Scientists Email: clyde.murley@comcast.net

Status: PARTY

**BRIAN HANEY** 

UTILITY SYSTEM EFFICIENCIES, INC.

1000 BOURBON ST., 341 **NEW ORLEANS LA 70116** 

FOR: UTILITY SYSTEM EFFICIENCIES, INC.

Email: brianhaney@useconsulting.com

Status: INFORMATION

ANDREW J. DALTON VALERO SERVICES, INC.

ONE VALERO WAY

SAN ANTONIO TX 78249-1616 Email: andrew.dalton@valero.com

Status: INFORMATION

THOMAS LU EXECUTIVE DIRECTOR WATSON COGENERATION COMPANY

22850 S. WILMINGTON AVE CARSON CA 90745 Email: luta1@bp.com Status: INFORMATION

**DOUGLAS DAVIE** 

WELLHEAD ELECTRIC COMPANY, INC

650 BERCUT DRIVE, STE C SACRAMENTO CA 95811 Email: ddavie@wellhead.com Status: INFORMATION

MICHAEL SHAMES

UTILITY CONSUMERS' ACTION NETWORK (UCAN)

3100 FIFTH AVE., STE. B SAN DIEGO CA 92103

FOR: UCAN

Email: mshames@ucan.org

Status: PARTY

ALAN NOGEE

UNION OF CONCERNED SCIENTISTS

2 BRATTLE SQUARE CAMBRIDGE MA 2238 Email: anogee@ucsusa.org

Status: PARTY

LARRY FINNE MANAGER

UNITED AIRLINES

SAN FRANCISCO INTERNATIONAL AIRPORT

SAN FRANCISCO CA 94128 Status: INFORMATION

MICHAEL S. HINDUS

PILLSBURY WINTHROP SHAW PITTMAN LLP

50 FREMONT ST

SAN FRANCISCO CA 94105

FOR: VALERO REFINING COMPANY-CALIFORNIA

Email: michael.hindus@pillsburylaw.com

Status: PARTY

ANDREW J. VAN HORN VAN HORN CONSULTING

12 LIND COURT ORINDA CA 94563

Email: andy.vanhom@vhcenergy.com

Status: INFORMATION

R. THOMAS BEACH PRINCIPAL CONSULTANT

**CROSSBORDER ENERGY** 2560 NINTH ST, STE 213A BERKELEY CA 94710-2557

FOR: WATSON COGENERATION COMPANY

Email: tomb@crossborderenergy.com

Status: PARTY

DANIEL DOUGLASS ATTORNEY

**DOUGLASS & LIDDELL** 21700 OXNARD ST, STE 1030 WOODLAND HILLS CA 91367

FOR: Western Power Trading Forum Email: douglass@energyattorney.com

Total number of addressees: 455

ROD MORTENSEN
WHEELABRATOR HUDSON ENERGY COMPANY, INC.

20811 INDUSTRY RD. ANDERSON CA 96007 Status: INFORMATION

Status: INFORMATION

JERRY BLOOM
WINSTON & STRAWN, LLP
101 CALIFORNIA ST, 39TH FLR
SAN FRANCISCO CA 94111-5894
Email: jbloom@winston.com

ROBERT FREEHLING LOCAL POWER RESEARCH DIRECTOR LOCAL POWER PO BOX 606 FAIR OAKS CA 94574 FOR: Women's Energy Matters/Local Power

Email: rfreeh123@sbcglobal.net

Status: PARTY

KEVIN WOODRUFF WOODRUFF EXPERT SERVICES, INC. 1100 K ST, STE 204 SACRAMENTO CA 95814

FOR: WOODRUFF EXPERT SERVICES Email: kdw@woodruff-expert-services.com

Status: INFORMATION

SETH F. RICHARDSON
WINSTON & STRAWN
333 SOUTH GRAND AVE. STE 3800

LOS ANGELES CA 90071

Email: sfrichardson@winston.com Status: INFORMATION

BARBARA GEORGE
WOMEN'S ENERGY MATTERS
PO BOX 548

FAIRFAX CA 94978

FOR: WOMEN'S ENERGY MATTERS

Email: wem@igc.org Status: PARTY

DONALD C. LIDDELL, PC DOUGLASS & LIDDELL 2928 2ND AVE SAN DIEGO CA 92103

FOR: Woodland Biomass Power, Ltd. Email: liddell@energyattorney.com